To be published in G. Chiarotti, P. Chiaradia (eds.): Physics of Solid Surfaces, Subvolume B, Landolt-Börnstein: Numerical Data and Functional Relationships in Science and Technology - New Series, Subvolume III/45B Springer-Verlag Berlin Heidelberg

Epigraphene : epitaxial graphene on silicon carbide

Claire Berger^{1, 2}, Edward H. Conrad¹, Walt A. de Heer¹

¹ School of Physics – Georgia Institute of Technology, Atlanta, GA 30332, USA ² Institut Néel, CNRS – Université Grenoble Alpes, 38042 Grenoble, France

1. Introduction and overview

1.1. Epitaxial graphene and transferred graphene

Graphene, a single sheet of carbon atoms, and the basic building block of graphite (see Figure 1) has been studied for more than half a century. Last century, academic graphene research focused primarily on surface science of epitaxial graphene on various metal surfaces, as well as on silicon carbide. Only some of its electronic properties were theoretically considered, but none were experimentally probed. In the past decade, initially fueled by its potential for electronics, graphene research has flourished, following two main distinct paths: graphene grown epitaxially on silicon carbide, or epigraphene¹, and graphene that is produced by a variety of methods and is designed to be transferred onto various substrates, known as transferred graphene. Basically, the two materials only differ due to the substrate that they are on, but this difference is fundamentally important. Much research has been devoted to producing large essentially perfect graphene sheets, which was considered by many to be an essential first step for graphene electronics. However, essentially all applications require graphene structures. Graphene electronics is an extreme case that requires highly reproducible graphene nanostructures in order to be technologically interesting. However, most nanolithographic pattering methods are detrimental to graphene. Consequently nanoelectronic graphene devices are not competitive with conventional nanoelectronics. Epigraphene is an exception, and is currently the only type of graphene that is suitable for graphene nanoselectronics. [1] [2]

Historically, the first examples of patterned epitaxial graphene for graphene-based electronics were published in 2004, in a paper titled: "Ultrathin epitaxial graphite and a route to graphene based electronics" [1]. The present review follows the history and the developments of that seminal work, that, incidentally presents the first (monolayer) graphene transport measurements.

In 2005, Novoselov et al. [3] invented a method now generally known as the "Scotch tape method". In this method, adhesive tape is used to cleave (exfoliate) graphene flakes from bulk graphite and transfer them onto oxidized silicon wafers, to demonstrate their field effect properties. In 2004 the same group reported very similar properties in ultrathin graphite flakes transferred on oxidized silicon (using a different deposition method). The paper, titled "Field effect in atomically thin graphite" [4], carefully demonstrates that the measured properties were those of thin graphite (see also similar work by Zhang et al. in 2005 [5]). Nevertheless this paper is widely cited not only for demonstrating graphene properties, but also for the discovery of graphene. The controversial discovery claim was based on the authors presumption, that freestanding graphene should be expected to be chemically unstable, unaware at the time that in 1962 Boehm et al.[6] had already produced and identified freestanding graphene sheets. Boehm also coined the name "graphene" in 1986 [7, 8].

Transferred graphene was originally characterized as "quasi-freestanding" in Ref [3] to distinguish it from other previously known forms of graphene directly grown on substrates including epigraphene. The name reflected the

¹ The term epitaxial graphene originally coined for graphene on SiC [1] is now widely used for graphene grown epitaxially – or not – on various metals. For this reason in this review we refer to epitaxial graphene on SiC as epigraphene.

belief at that time that interactions with the SiO₂ substrate were negligible. Later research found that substrate induced disorder is significant. In contrast, epitaxial graphene on silicon carbide is widely used to demonstrate the intrinsic band structure of graphene[9-11] (see Figure 2) that is not observable in transferred graphene (see Figure 3 and Section 4).

Epitaxial graphene is a graphene film that is directly grown on various crystalline surfaces. The carbon atoms of the graphene layer are orientationally registered and nearly commensurate with the atomic lattice of the substrate surface. The degree of chemical bonding of the graphene to the substrate varies from extremely weak to relatively strong, and the graphene properties are modified accordingly. In contrast, there is in general no coherent atomic registration for graphene films that are transferred onto surfaces, for example on oxidized degenerately doped silicon wafers (graphene transferred on boron nitride single crystals is a notable exception but the registry is not controlled). Various transfer methods have been developed, either by drying graphene flake solutions [4, 6] or by direct mechanical transfer from graphite onto these surfaces (as first done by Novoselov et al. [12] in 2005). The Scotch tape method, has a great advantage for two dimensional electron gas (2DEG) investigations, [12, 13] because the charge density of the graphene layer can be adjusted by electrostatic "back gating", which cannot be done with epitaxially grown graphene. For back gating a parallel plate capacitor is made of a thin SiO₂ dielectric sandwiched between a conducting silicon substrate and the graphene layer(s) that is transferred on it. For epigraphene, by virtue of its growth process directly from the silicon carbide crystal, such an insulating barrier does not exist., However the transfer process has the disadvantage of considerable disorder inherent in the transfer.

Epitaxial graphene on silicon carbide (epigraphene) was first identified in 1962 by Badami [14, 15], followed by van Bommel et al. [16] in 1973, in the investigations of the graphite layers that spontaneously grow on silicon carbide when silicon carbide crystals are heated to extremely high temperatures (>1000 C) in vacuum. The growth proceeds by sublimation of Si from the SiC surfaces, resulting in a carbon-rich surface that reconstructs to produce graphene. It is interesting, that graphitic layers on heated SiC crystals were already noted in investigations by G.E. Acheson, who also first synthesized silicon carbide in 1891.[17] It is worth noting also that in 1907 H.J Round produced the first SiC light emitting diode and SiC diodes were used in the first radio receivers [18]. These are harbingers of SiC based electronics.

The invention of graphene-based electronics [19] (patented in 2003) was based on the earlier graphene research in combination with carbon nanotube electronics research. The choice of SiC as the substrate material was motivated by the strict requirements for high-end electronic materials (i.e. Si, Ge, GaAs, SiN, SiC). This requires a platform that is reliably nanopatternable, which demands that the substrate is a single crystal. Single crystal, electronics grade SiC is commercially available and it is currently extensively used in electronics. In the last decade, SiC based electronics has developed significantly including high temperature complementary metal oxide semiconductor (CMOS) technology [20]

As we will see in this review, epigraphene has been shown to satisfy requirements for electronics grade graphene. Moreover, various schemes to modify the properties of epigraphene using nanopatterning methods and by tuning the interaction with the SiC substrate have added significant flexibility as discussed below.

In contrast, most transferred graphene and thin graphitic flake research currently focuses on chemical properties, para-electronics and physical properties like super-capacitors, transparent and flexible conductors, in optoelectronic and photonic demonstrators and as ultrathin membranes [21, 22]. This shift in emphasis is primarily due to the inherent difficulties of reliably producing high-end electronics grade graphene-based materials by transferring graphene (grown epitaxially on metal surfaces or from graphite) onto various substrates as mentioned above. While there is still considerable research on fundamental graphene properties using small mechanically transferred graphene flakes, expectations that transferred graphene will significantly impact high-end electronics have recently faded. The current confusion of the definition of graphene, is not merely one of semantics, but actually confounds graphene research with nanographite research (a decades old, mature field).

1.2. Definition of graphene

Graphene was defined [7] by H-P Boehm in 1986, and the nomenclature was officially adopted[23] by the International Union of Pure and Applied Chemistry (IUPAC) in 1994. It reads as follows:

"Graphene is a single carbon layer of the graphite structure, describing its nature by analogy to a polycyclic aromatic hydrocarbon of quasi infinite size. Previously, descriptions such as graphite layers, carbon layers or carbon sheets, graphite monolayers have been used for the term graphene. Because graphite designates that modification of the chemical element carbon, in which planar sheets of carbon atoms, each atom bound to three neighbors in a honeycomb-like structure, are stacked in a three-dimensional regular order, it is not correct to use for a single layer a term which includes the term graphite, which would imply a three-dimensional structure. The term graphene should be used only when the reactions, structural relations or other properties of individual layers are discussed."

Consistent with this definition, "graphite" should be used when graphene layers are stacked in the graphitic (Bernal) structure (see Figure 1), so that "few layer graphene" is incorrect and should be referred to as "thin graphite".

In practice, graphene (including transferred graphene) is usually supported on a substrate. The term "freestanding graphene" is often used to suggest the absence of substrate-induced perturbations. Some even have suggested [24] to redefine graphene accordingly:

"Graphene is a single atomic plane of graphite, which—and this is essential—is sufficiently isolated from its environment to be considered freestanding".

However in practice graphene on a substrate is never freestanding. Substrates always affect the properties and quite significantly so in graphene transferred on SiO₂ using the "Scotch tape" method, making this alternative definition ineffective. It makes more sense to adhere to the IUPAC definition of graphene and to apply "quasi-freestanding" relative to a property. For electronic properties, quasi-freestanding implies that the electronic properties are essentially identical to those of an ideal graphene sheet. A recent publication in the authoritative journal Carbon suggests the usage of more precise definitions [25].

The adhesion of graphene to many substrates involving van der Waals forces and/or electrostatic forces usually minimally affects its chemical and electronic properties (see below). But the adhesion to the substrate can be strong, involving significant chemical bonding of the carbon atoms in the graphene layer to the substrate. In these cases the electronic structure (as well as planarity) will be significantly modified. Finally, graphene can also be functionalized, in which cases atoms or molecules are chemically bound to graphene carbon atoms, whereby the electronic structure is typically significantly modified. Consequently chemical functionalization and chemical bonding to a substrate are closely related. Below, examples of all three forms of graphene are presented in the context of epitaxial graphene on silicon carbide. Properties of "structured graphene" ribbons and islands are also discussed.

1.3. Graphite, freely suspended graphene, and graphene isolated on substrates

Until 1987, graphene was known as *monolayer graphite*, emphasizing its nature as a structural unit of graphite. Graphene was already known to be one of the most chemically and mechanically stable materials in nature, with a cohesive energy of \approx 7eV. This extreme stability of the individual graphene sheets results from the sp² bond, as first described theoretically by Pauling [26]. Pauling demonstrated that the sp² hybridization produces three extremely strong symmetrically arranged coplanar bonds (the sigma bonds), which explains why polycyclic aromatic hydrocarbons (including graphene) are flat and rigid: it requires energy to bend these structures.

In addition to the in-plane sigma bonds, each carbon atom has an atomic p_z orbital that extends above and below the plane. Overlap of these orbitals produces π -bands that are responsible for the electronic properties of graphene.

Graphene is so stable that it requires temperatures exceeding 4000° C (as occur, for example, in electric arcs) to convert it into fullerenes, nanotubes, soot and other curved graphitic structures, as was experimentally demonstrated in the late 1989 in the famous experiments by Krätschmer et al.[27] and Ebbesen et al.[28].

In contrast, the interlayer bonding of the graphene layers in graphite is extremely weak, on the order of 30 meV per carbon atom [29]. This small value (less that 1% of C-C bond energy in graphene) represents the difference in energy of a carbon atom in graphite compared to a carbon atom in a free graphene sheet. The weak adhesion explains why graphene layers are so easily peeled from a graphite crystal in the "Scotch tape method" for example. This anciently known property underlies graphite's name (graphite, from *graphein*, the Greek word for to write). Consequently, for all practical purposes, graphite's exceptional chemical and mechanical stability is entirely due to graphene's stability, and the added stability due to the crystal substrate is negligible. This explains the well-known fact, that almost all chemical and physical properties of graphene and graphite are practically identical.

Nowadays the most common forms are: aqueous graphene suspensions produced from exfoliated graphite[6, 30-32]; chemical vapor deposition on various metal substrates (for a review, see [33]); epitaxial growth on SiC by vacuum sublimation[15, 34]; deposition on various substrates by mechanical exfoliation of graphite [3].

The suspensions of graphene flakes, produced and isolated by H-P Boehm in 1962 were, identified with, for those times, novel electron microscopy methods, revealing that they indeed were graphene monolayers. These tour-de-force measurements were published in the most prominent German Science journal of the time [6]. As a carbon chemist Boehm knew that his work did not amount to the discovery of graphene, nor proof of its stability. This work was performed mainly as an academic exercise to demonstrate that freestanding graphene flakes could be made and measured.

Later, graphene was epitaxially grown on many single crystal metal surfaces, by heating metal samples in the presence of carbon containing gasses (chemical vapor deposition (CVD)). In these experiments performed in ultrahigh vacuum, the atomic and electronic structure can be probed using a variety of surface science probes. In 1997 Gall et al. [33] recognized the quasi-freestanding nature of epitaxial graphene on various metal surfaces, as well as the fact that they were natural two dimensional crystals. In their article titled "Two Dimensional Graphite Films on Metals and Their Intercalation", the first sentence of the abstract reads:

"Two-dimensional graphite films (2DGF) on solids are wonderful objects, real nature-made two-dimensional crystals.... A special attention is paid to intercalation of 2DGF — a process when foreign atoms and even molecules (fullerenes C_{60} molecules) spontaneously penetrate between graphite film and metal substrate."

In this paper, and many other last century papers, the quasi-freestanding nature of the graphene layer is explicitly demonstrated, as it was by Forbeaux et al., in early investigations of epigraphene, who realized that the epitaxial graphene layers appeared to be "floating above the substrate".[35]

Several investigators pursued controlled mechanical exfoliation of graphene with explicit attempts to produce graphene by peeling layers from graphite. In 1999, Ruoff and coworkers published a paper titled: "*Tailoring graphite with the goal of achieving single sheets.*" [36] The process involved producing micron sized graphite islands on a substrate and mechanically peeling layers off with an atomic force microscope. The process produced ultrathin graphite flakes but not monolayers. In 2004 Novoselov and co-workers ultrasonicated islands produced by Ruoff's method resulting in suspension of micron-size flakes.[4] The suspensions were dried on oxidized silicon wafers. The electrical properties of micron sized ultrathin graphite flakes were measured. After Novoselov and coworkers measured transport properties of graphene flakes produced by the "Scotch tape method" in 2005 [3], Novoselov et al. [12] and Zhang et al. [13] simultaneously published quantum Hall effect measurements in graphene.

Graphene was first exfoliated chemically by separation of the layers through strong oxidation (graphite oxide, also called graphite paper), dispersion in aqueous solution followed by reduction [6]. Graphene can also be directly exfoliated in solution by ultrasonication in various solvents [31, 32] or using surfactants, similarly to carbon nanotubes and for stabilizing polymers [37], resulting submicron square (<10 µm² in area) graphene flakes.

2. The electronic band structure of graphene

Graphene is a semimetal. This means that there is no energy gap between the valence band and the conduction band. Graphene is therefore often referred to as a gapless semiconductor. It is quite curious that such a material

could nevertheless be interesting for electronics applications that typically involve semiconductors with significant energy gaps (i.e. for Si the band gap is 1.1 eV). In fact, even gapless graphene can be used in certain electronic applications, while limited, like RF electronics. However, developing methods to "open a gap" in graphene is considered to be essential for digital electronics.

In order to understand the band structure of graphene, it is important to realize that the hexagonal graphene lattice is actually a triangular lattice with two carbon atoms in each unit cell, one called A and the other called B (see **Figure 1**). While both A and B are chemically equivalent, they are not equivalent from a crystallographic point of view, which leads to an interesting degeneracy resulting in two-component wave functions as explained below.

2.1. Tight-binding: graphene

The electronic structure of graphene is remarkably well described in the nearest neighbor tight-binding model involving only the p_z orbitals[38, 39] A general review can be found in Castro-Neto et al. [40] where this model is used extensively. In this model the sp² bonding is represented by a quantum mechanical overlap integral $\gamma_0 \approx 3$ eV, which represents the overlap of p_z orbitals centered on neighboring carbon atoms (see Figure 1b); when the p_z orbitals of two neighboring atoms are in phase, the bond energy is $+\gamma_0$ when they are out of phase it is $-\gamma_0$, with values in between when the phase is between these extremes. In this way, waves develop with a wave vector **k**. These quantum mechanical "phase waves" that oscillate at frequency ω have energy $E=\hbar\omega$ and momentum $p=\hbar k$, are the "quasi particles" that transport energy, momentum, and charge in graphene at a speed $v_F = d\omega/dk$ (much like a water wave transports energy and momentum, while the water itself only moves up and down). Quasiparticles with positive energy are called "electrons" those with negative energy are called "holes".

A plot of E(k) is the band structure as shown in Figure 2a

$$E(k) = \pm \gamma_0 \sqrt{\left(1 + 4\cos(\frac{\sqrt{3}}{2}ak_x)\cos(\frac{1}{2}ak_y) + 4\cos^2(\frac{1}{2}ak_y)\right)}$$

Where a= 0.246 nm is the graphene lattice constant (the C-C distance is 0.142 nm). For charge neutral graphene, i.e. at E=0, the Fermi surface consists of a hexagon of 3x2 points (K and K' points; $|\mathbf{K}| = 2\pi a/3$).

It is instructive to consider the case for which $k_x=0$ (i.e. a cut from along the Γ K direction, Γ corresponds to the k=0, see Figure 2a) for which

$$E(k_y) = \pm \gamma_0 (1 + 2\cos(a k_y / 2))$$

This shows two simple cosines that intercept at E=0 for ky= $2\pi a/3$. Negative energies corresponds to bonding states (the π bands), for which all of the p_z orbitals at **k**=0 (the Γ point) are in phase, while the positive energies corresponds to the anti-bonding states (the π^* bands) where p_z orbitals at **k**=0 of the A atoms are out of phase with those of the B atoms. At the K point (E=0) the bonding states and anti-bonding states are degenerate. The group velocity of the quasiparticles at the K-point is $v_F = a\gamma_0 \hbar^{-1} \sqrt{3}/2 \approx 10^6$ m/s.

Near the K and K' points the band structure is reflection symmetric with E(k) cones that touch at the charge neutrality point where the energy E=0. These so-called Dirac cones are also reproduced in the simplified Hamiltonian and eigen energies, near the charge neutrality point:

$$H_k = v_F \sigma \cdot k^*; \qquad E(k) = \pm \hbar v_F k^* \qquad (Eq. 1)$$

Where $k^*=|\mathbf{k}-\mathbf{K}_i|$ the Fermi velocity $v_F = \sigma$ are the 2x2 Pauli spin matrices σ_x and σ_y and \mathbf{k} is the two component momentum vector, k_x and k_y and \mathbf{K}_i represent the K and K' points. In this approximation near E=0, the quasiparticles move at a constant velocity independent of their energy, as for photons (or more appropriately, neutrinos, see below). This property is of fundamental importance for graphene electronics, since it sets a scale for quantum confinement effects. In particular, since the Fermi wavelength writes

$$\lambda_F = 2\pi/k^* = 2\pi\hbar v_F/E_F$$

for E_F =25 meV (corresponding to room temperature) one gets λ_F = 160 nm. Hence, generally, room temperature quantum size effects are expected to be relevant for graphene structures up to this size (which is much larger than for typical metal structures).

2.2. Tight-binding: graphite

Wallace's paper [38] concerned calculating the band structure of graphite. In the graphite Bernal structure, graphene layers are stacked on top of each other, where the A atoms on one layer are directly above B atoms in the two neighboring layers, and the B atoms of that layer are above the centers of the carbon hexagons of neighboring layers (see Figure 1b-c). Consequently, the p_z orbitals of the A atoms (but not the B atoms) in that layer weakly overlap with the B atoms in the neighboring layers. Due to this alignment, the A and B sublattice degeneracy is lifted, causing the electronic structure of graphite to be significantly modified near E=0, even though the interactions are weak. The Dirac cones near E=0 evolve into four hyperbolic bands, two that touch and two that are separated by an energy on the order of the interlayer overlap integral $\gamma_1 \approx 0.4$ eV. Specifically, the eigen energies, near K points and for $k_z=0$, are then

$$E = \pm \gamma_1 \pm \sqrt{\gamma_1^2 + \left(\hbar v_F \left| k - K_1 \right|^2\right)}$$

In this way the graphite quasi particles attain a small mass in the graphene plane (as well as a large mass perpendicular to the plane). The complete model (see for example Ref [41] involves 13 parameters that have been fitted to experiment (the McClure Slonczewski-Weiss parameters [39, 42]).

2.3. Ab initio methods

Both the graphene and graphite π -band structure are well described in the tight-binding approximation (of which the nearest-neighbor version discussed above is the simplest). These models provide analytical solutions that are appealing from a physics point of view, but they involve several empirical parameters. Modern ab initio computer-based simulations are used [43] that do not use free parameters. Density functional methods, of which the local density approximation (LDA) is the most commonly used variety, and the *GW* approximation, which stands for Greens function *G* of the Coulomb interaction *W*. This method is more accurate because it includes the long-range electron correlation effects that are absent in LDA. While numerically more accurate these computations typically do not provide analytical solutions resulting in some loss in transparency of the underlying physics. Ab initio calculations are particularly important in more complex problems, for example, when substrate effects, functionalization, edges and correlations are considered.

2.4. Relativistic interpretation

At the other extreme is the relativistic interpretation of the graphene band structure. Divincenzo [44] first recognized that in Wallace's nearest neighbor, tight-binding Hamiltonian, (Eq. 1) very close to the K points is formally identical to the Weyl Hamiltonian (also known as the massless Dirac Hamiltonian). The Weyl Hamiltonian describes neutrinos as massless spin ½ Fermions (see also Ando et al. [45] and Khveshchenko [46]). For neutrinos, σ in Eq. 1 represents the neutrino spin (its intrinsic angular momentum). For graphene σ is a two component vector (called pseudo-spin) that quantifies the π and π^* components of the wave functions. Furthermore, in the neutrino case, $c = 3 \times 10^8 \text{ m/s}$ (ultra-relativistic) replaces $v_F = c/300$ (non-relativistic speeds) in graphene. Note that relativistic interpretation of the nearest-neighbor tight-binding approximation at $|\mathbf{k} - \mathbf{K}_i| \approx 0$ introduces no new physics beyond the nearest neighbor tight-binding formulation. Moreover, it fails to describe graphene well away from $E \approx 0$ [40]. The analogy with relativistic quantum mechanics to be applied to graphene. Unfortunately, the "relativistic nature" of quasi particles in graphene has led to considerable confusion, especially in the popular science literature.

2.5. Physics near the K point

The electronics properties of graphene near the K points are particularly interesting. There, the Fermi wavelength $\lambda_F=2\pi v_F/k^*$ diverges as $k^* \rightarrow 0$. Moreover, since graphene is neutral at the K point, the electric fields of the quasiparticles are not screened. This implies that the quasiparticles strongly interact with each other resulting in long-range interactions. Therefore they cannot be treated as approximately independent particles (as is usually the case in metals and semiconductors). Furthermore, due to the strength of the interaction, the usual theoretical many body techniques, based on perturbation theory, fail near the K point. This can be understood as follows: in quantum electrodynamics calculations, the perturbation expansions involve powers of the fine structure constant $\alpha=e^2/\hbar c\approx 0.0073 <<1$, typically resulting in converging expansions. However, for graphene $\alpha=e^2/\hbar v_F \approx 2.2>1$, so that the expansions tend to diverge. It is generally expected that these interactions result in a broken symmetry ground state. Furthermore, lack of screening near the K point is also generally expected to greatly increase the Fermi velocity. There has been some recent evidence for this effect [47]. In any case, it is clear, that the graphene must be exceptionally pure for effects close to the K point (with diverging Fermi wavelengths) to be observed. Impurities like charge puddles, random strain, and other defects, even at low concentrations n, will distort the pristine graphene properties at wavelengths $\lambda \approx 1/\sqrt{n}$.

The many body effects near the K point have for a long time been one of the most interesting unanswered questions in graphene physics and they are starting to be addressed experimentally.

2.6. Influence of the substrate : epitaxial graphene

In graphene grown epitaxially on a substrate, the graphene lattice is (almost) commensurate with the substrate lattice, thereby producing a higher order (super cell) modulation of the lattice (examples of higher order patterns for graphene on the buffer layer – see Section 4.1)– are given in Figure 4a-b). The unit cell of the (super cell) is much larger than that of graphene. In general, within a super cell, neither A or B graphene atoms are singled out. Consequently, the graphene A-B sublattice degeneracy is not lifted. Furthermore, if the interaction strength of the substrate to the graphene layer is weak (i.e. van der Waals like and no significant sp³ bonding), then the electronic structure near the K points is not greatly perturbed. Consequently, in those cases, the graphene may be considered to be quasi-freestanding regarding the electronic properties at the K-point, as discussed by Gall et al. [33]. However, the super cell periodicity will cause energy gaps on the order of the interaction strength at periodic intervals in the Brillouin zone [48]. In other cases the bonding to the substrate is so strong that the p_z orbitals of the graphene layer hybridize with the substrate atoms, producing sp³ bonds with the substrate. [33] These chemical bonds significantly affect the graphene electronic properties.

Graphite, or "Graphene on graphite", discussed above can be seen as a special case of epitaxial graphene. But it is exceptional because in this case A-B symmetry is lifted. In Figure 2b-c, the effect A-B symmetry lifting is demonstrated by band structure measurements of AB and ABC stacked epigraphene on SiC (stacking defined in Figure 1a-d).

2.7. Role of stacking: multilayer epitaxial graphene

Graphene multilayers grown on SiC in some cases are not Bernal stacked as they are in graphite, but rather, adjacent layers are rotated by about 30 degrees (see Section 4.2.3). In this case, even though the interlayer interactions are very similar to those in graphite, the A-B sublattice degeneracy is not lifted (as it is in graphite) and consequently, the Dirac cone structure is essentially unperturbed [9] In this way, these multilayers behave like a stack of decoupled graphene layers [49-54]. Multilayer epitaxial graphene (MEG) has been studied in great detail, and has provided a wealth of information on the properties of graphene close to the Dirac point (see Section 4.2).

2.8. Influence of substrate: transferred graphene

Mechanically exfoliated graphene is graphene transferred from graphite to a secondary substrate. The method requires that adhesion of graphene to these substrates is greater that the adhesion of graphene to graphite, otherwise graphene would not transfer. Note that when these transfers are performed in ambient conditions (not

high vacuum), contamination trapped between the graphene layer and substrate is unavoidable, giving rise to charge inhomogeneity (charge puddles) [55-57]. Furthermore since graphene is flexible, significant stress results from the transfer process on the non-flat substrates. This random stress causes disorder in the electronic structure of the graphene [57-59]. Both sources of disorder affect the electronic properties in particular near E=0 in an energy range that is comparable to the energy of the interaction with the substrate and characteristic energies related to the disorder (typically hundreds of meV) [56, 59]. This disorder may obscure the physics near the Dirac point. [59]

In order to mitigate some of these problems, graphene has more recently been deposited on single crystals of boron nitride [48, 60] (a semiconductor with an atomic structure that is similar to graphene), producing a quasi-epitaxial form of graphene that is considerably better ordered than graphene deposited on amorphous substrates.

2.9. Graphene nanostructures

While most research has focused on the electronic properties of extended (quasi two-dimensional) graphene sheets, graphene nanostructures are technologically more relevant and theoretically quite interesting in their own right. As mentioned above, quantum confinement effects are manifested in relatively large graphene nanostructures even at room temperature. In fact, graphene electronics was originally motivated [1, 19] by the predicted properties of graphene nanoribbons that have much in common with carbon nanotubes.

However, the advantage of graphene ribbons compared with carbon nanotubes is that ribbons can be seamlessly interconnected using graphene leads [1], so that entire graphene integrated circuits could in principle be cut out from a single graphene sheet. In this way, major hurdles hindering the development of carbon electronics based on carbon nanotubes (i.e. placement and interconnects) could be overcome.

The first tight-binding calculations performed on graphene nanoribbons were performed by Fujita et al. [61] and Nakada et al. [62] in 1996, who showed that graphene ribbons could be semiconducting or metallic, depending on the width W and the structure of the edges, as shown Figure 5. These simple calculations show that relatively large bandgaps due to width confinement, of the order of Eg=1eV /W (the width W is in nm), can be achieved by patterning narrow ribbons, using, for example, nanolithography methods. While realization of these nanostructures is certainly quite difficult, because of the small sizes and high accuracy required, the overall paradigm of a monolithic structured graphene sheet as a foundation for carbon circuitry is a new direction in electronics (see Section 5.5).

Because the Fermi velocity in graphene is nearly constant, the energy levels $E_{n,m}$ of a rectangular graphene structure, of length L and width W are closely related to the normal modes of a optical wave guide. Consequently,

$$E_{n,m} = hv_F \sqrt{(n/L)^2 + (m/W)^2}$$
 Eq. 2

This is a reasonable approximation but it ignores the particular properties of the edge states mentioned above (Figure 5). It does show, however that when L and W are relatively small, at low temperatures and in charge neutral graphene, only the lowest modes are occupied. For example, at T=4K (corresponding to liquid He) a charge neutral sample of $10\mu m x 10\mu m$ has $E_{1,0}(= 4 \text{ meV}) >> k_B T$, so that only the n=m=0 mode will be occupied. A 100 nm wide ribbon will only have the m=0 modes occupied and therefore it behaves like a one dimensional wire, even at room temperature (see Section 5.5).

3. Silicon carbide and epitaxial graphene on silicon carbide

3.1. G.E Acheson: Silicon carbide, Graphite and Graphene

Silicon carbide is a synthetic compound, first mass-produced by E.G. Acheson in 1891 (who called it Carborundum) by heating aluminum silicate with carbon. This process was patented in 1893 and Acheson founded the Carborundum Company. Silicon carbide was immediately industrially interesting because of its extreme hardness,

so that it was used as an abrasive. Interestingly, Acheson also discovered that when carborundum was heated to a high temperature it produced extremely pure graphite. He patented his graphite-making process in 1896 and founded the Acheson Graphite Company in 1899. The company initially produced graphite for incandescent light filaments and electrolysis anodes and electrodes for arc furnaces. Interestingly Acheson also discovered a way to produce ultrafine graphite colloidal suspensions in water, that he produced in his Acheson Industries Company under the trade-name AquaDAG (DAG: Deflocculated Acheson Graphite). These "graphene-like" conducting inks continue to be used extensively in the electronics industry to produce conductive coatings inside electronic vacuum tubes and cathode ray tubes ever since the early days of radio and television. It is now used for the electronic, automotive, aerospace, appliance, medical, metalworking, die casting and energy storage industries [32].

3.2. Electronics grade silicon carbide

Crystalline silicon carbide is a wide band gap semiconductor that occurs in more than 150 crystalline forms (polytypes) (see for instance Ref. [63]). The most relevant polytypes for electronics and epigraphene are hexagonal 6H (band gap: 3.05 eV) and 4H (3.23 eV) and cubic β -3C (band gap: 2.36 eV) (see Figure 6). The structures consist of Si-C bilayers which stacking determines the polytype. The 4H-(aSiC=3.0805Å and cSiC=10.0848Å[64]) and 6H-hexagonal crystals consist of 4 (6, respectively) Si-C bilayers (see Figure 6). The 4H commercial crystals are generally considered of higher quality. The hexagonal planes are polar, that is the (0001) face is made only of Si atoms (the bulk terminated surface Si atoms have one dangling bond per atom), and the opposite (000-1) has carbon atoms only. Single crystal wafers of these two polytypes with diameters up to 150 mm are produced by several companies. These crystals are produced by the physical vapor transport method (a modified version of the original Lely process), at elevated temperatures above 2000° C in a closed graphite crucible. SiC powder, placed in the hot zone of the growth cell (e.g. 2200° C), sublimes and re-crystallizes in the colder zone (e.g. 2150° C) at a SiC seed. Control of various defects ("micropipes" and screw dislocations) has plagued SiC electronics industry from the outset, however improvements in the quality as well as reduction in cost have been dramatic in the past decade, and this trend is expected to continue.

Electronics grade SiC is commercially available, produced as semi-insulating and degenerately n-doped SiC. For semi insulating SiC, the Fermi level is pinned in the band gap for example by doping with vanadium. SiC is widely used in electronics, especially for discrete components like light emitting diodes and high voltage Schottky diodes. Recently, high voltage (1200 V) MOSFETs have become commercially available, and high temperature CMOS integrated circuits have been demonstrated.[20]

3.3. Epitaxial graphene growth by silicon sublimation

Epitaxial growth of graphene on SiC is non-conventional. Epitaxial growth of graphene on metals typically involves the decomposition of a carbon-containing gas on single crystal metal surface, after which the carbon atoms arrange into the graphene structure that registers with the crystalline metal surface. However, epitaxial graphene growth on SiC is different and typically does not involve an external source of carbon. Rather, silicon sublimes from the hot SiC surfaces, resulting in a carbon rich surface that then forms a graphene layer on the SiC surface.

Van Bommel et al. [16] was among the first to study this graphitization of SiC (as used by Acheson) as a surface science problem (see also [14]). The motivation for these early studies was to understand the surface phase transitions of clean SiC as a function of temperature. For this, 6H-SiC crystals were heated in ultra-high vacuum and their surfaces were measured with low energy electron diffraction (LEED) and Auger electron spectroscopy as a function of annealing temperature. It was discovered that as temperature increases a monolayer graphite layer grows and stable graphite film ultimately formed on the SiC surface [16]. This early work also showed that graphene grows differently on the two polar surface of hexagonal (4H/6H) SiC: the (000-1) carbon terminated face (C-face) and the (0001) silicon terminated face (Si-face) (see Figure 6 and schematics Figure 7d). On the Si face the graphitization proceeded with an initial phase at 1000° C involving a carbon rich reconstructed SiC surface with a 6 $\sqrt{3}$ R30 crystal structure (buffer layer, see Section 4.1.1). This is followed by the formation of an epitaxial graphene layer at higher temperatures and ultimately thin (Bernal) graphite at 1500° C. The C-face, on the contrary, showed signs of different graphene-SiC orientations. While in the years from 1975-2000 there were many studies of Si-face graphene, very few were conducted on C-face graphene because of its perceived disorder (for a review of the earlier

work see Hass et al. [65]). As discussed in Section 4.2.3, the C-face is in fact ordered but the multiple layers are non Bernal stacked, which has important consequences for its electronic properties.

It should be noted that graphene layers growth by silicon sublimation in ultra-high vacuum (the growth method used in earlier studies) typically results in lower quality films that have a so-called "Swiss cheese" nanostructure as shown in Figure 4(d). For this reason, alternative graphitization methods have been developed [34, 66].

3.4. Confinement controlled sublimation (CCS)

In free sublimation of silicon in vacuum, the sublimation rate is controlled only by the temperature. If one considers a silicon carbide crystal in a background of silicon vapor, then equilibrium will be established when the rate of absorption of silicon atoms on the surface equals the sublimation rate. Consequently, this equilibrium temperature depends on the silicon vapor pressure at the SiC surface [67]; in equilibrium the formation of graphene is arrested.

Free silicon sublimation (in UHV) produces defective graphene layers because the dynamics controlling the formation of graphene at a given temperature cannot keep up with the rate at which free carbon is produced at the surface, resulting in poorly developed graphene layers. Consequently, either slowing down the rate of carbon liberation, or increasing the temperature so that the graphene formation process is speeded up should produced better graphene. At the same time, growth over the entire sample surfaces will be uniform if the growth processes occur close to equilibrium.

The confinement controlled sublimation method [34] is designed to provide growth conditions close to equilibrium. In this method, a small silicon carbide crystal is placed in a graphite ampule that is supplied with a small calibrated hole (Figure 7b). The assembly is placed in a vacuum chamber that it is uniformly heated (typically using an radio frequency induction heater, as shown in Figure 7c) to a given temperature (in the range of 1500-1700°C). The silicon that evaporates from the SiC surface will dwell inside the ampoule for relatively long times, before it can escape out of the hole. Consequently, a silicon background pressure builds up and growth occurs in quasiequilibrium conditions. The rate of graphene formation is directly related to the rate at which the silicon escapes. By confining silicon, the growth temperatures can be increased. Ultimately, the quasi-equilibrium growth at elevated temperatures assures a uniform graphene layer extending over the entire crystal surface. Optimum growth conditions (temperatures, growth times, and hole size) are empirically determined (see sections 4.1.2 and 4.2.2).

3.5. Growth in Ar atmosphere (Edison light bulb method)

The rate at which silicon is depleted from the surface, and therefore the graphene growth rate, can also be controlled by heating the silicon carbide crystals in an inert gas background. While the silicon sublimation is unchanged, these atoms must diffuse through the Ar gas in order to escape. This diffusion process results in a Si density gradient in the gas, where the Si density is greatest at the surface. Hence, close to the surface the number of atoms leaving the surface is approximately equal to those returning to it. As in the confinement controlled sublimation method, the return flux of silicon atoms (that depends on the Ar pressure and temperature) and therefore the graphene growth rate can be adjusted. This method pioneered by Emtsev et al. [66] has been successfully applied to produce relatively uniform epigraphene/SiC at the full wafer scale [68-70].

While this method (which is similar to the way tungsten evaporation in incandescent light bulbs is controlled) appears to be similar to the CCS method (see for instance surface studies in Figure 8a-b), there is a fundamental difference. Whereas in the CCS method growth occurs under essentially uniform conditions over the whole surface, in Ar growth, it does not, because it is determined by diffusion of silicon atoms in the Ar gas. Growth rates depend on the boundary conditions and therefore sensitively on the chamber geometry, location on the sample, and critically on Ar flow and convection. Typically the growth temperature of a monolayer epigraphene on the Si-face ranges from 1100°C in UHV [35] to 1550°C in the CCS method (depending on the crucible hole size [34]), up to 1650°C [66] and 2000°C in one atmosphere of argon [71].

Variations of the high pressure growth have been implemented, for instance by supplying Si with (di)silane gas [72], or by providing external carbon atoms from gas (chemical vapor deposition, see for instance [73]) or solid carbon sources (molecular beam epitaxy [74]). In a recent development, higher mobility samples on the Si-face for

quantum Hall effect metrology (see Section 5.2.3) were obtained by balancing both carbon etchant (H₂) and carbon supply gases in a process combining chemical vapor deposition and SiC thermal decomposition. [75, 76]

4. Structure and band structure of epitaxial graphene on hexagonal silicon carbide

Since the early studies started in the 60's [14, 16, 35], epigraphene production methods have improved significantly so that higher quality material is now produced [34, 66]. The improved material, coupled with the ability to use a variety of modern surface science techniques have allowed more detailed electronic and structural information to be uncovered. No other form of graphene has the crystallinity, epitaxial registry and large-scale patternability that allow a broad range of surface analytical tools necessary to study and modify the properties of graphene. As an example Figure 3 shows a comparison of the surface roughness and K-point Dirac cone energy and momentum spread of various forms of graphene. AFM and STM images in Figure 3a-b show that the surface roughness is an order of magnitude larger in transferred graphene. Angle Resolved Photoemission Spectroscopy (ARPES) measurements show the dramatic difference between epigraphene and transferred graphene on SiO₂ (see Figure 3c-d). For large scale graphene grown by CVD on Cu, graphene studies are hindered by rotational disorder and the band gap induced by substrate interactions [77]. At the time of writing only epigraphene is sufficiently ordered to allow graphene's characteristic band structure to be revealed.

Graphene growth on the two 4H-(6H-)SiC polar surfaces are very different. Under identical conditions, graphene growth on the Si (0001)-face, is both slow and orientationally well defined with SiC. On the carbon face, epigraphene grows fast and a graphene layer readily forms. Multilayers subsequently grow with a variety of orientations that are commensurate with the SiC substrate, as can be seen in the LEED patterns of Figure 9 a-b. The status quo of the electronic and crystalline structure epigraphene is reviewed next.

4.1. Si-face

On the Si-face, graphene is rotated 30° with respect to the SiC <10-10> direction (see **Figure 9**b). [35, 65] Film thicknesses (up to about 5 layers) can be accurately controlled by adjusting growth conditions. The first graphene layer (the so-called buffer layer) is bound to the SiC and the π -bands are modified; instead of a Dirac point, a band gap is observed at the K-point.[78] The second layer (i.e. the graphene layer on top of the buffer layer) shows the text-book graphene band structure. Subsequent layers show characteristic thin graphite behavior.

The first ARPES graphene band structure measurements of the Si-face were performed by Rollings et al. [79] and by Ohta et al. [10]. They studied UHV grown graphene films (1 to 4 layer not counting the buffer layer). Figure 2b-e show the ARPES spectrum at the graphene K-point as a function of film thickness. The linear π -bands (Dirac cone) are clearly observed for the monolayer film (Figure 2b; note that the monolayer film is n-doped by ~0.45eV due to the SiC substrate. The AB stacking (Bernal stacking) of the second layer, that causes the lifting of the degeneracy of the two graphene sublattices, is evident from the splitting of the π -bands (Figure 2c). The bands are shifted to lower binding energy (~-0.5eV). Also note that a small band gap has formed. This gap that lies well below the Fermi level can be varied from 0 to 0.25 eV by changing the doping level. [80] The gap originates from the perpendicular electric field due to the difference in doping of the two layers [81]. ARPES shows that the graphene stacking in thicker films is a mixture of Bernal and rhombohedral stacking (see Figure 2d-e). [10, 82]

Electron- phonon, electron-electron and electron-plasmon interactions, are observed in detailed ARPES studies of epigraphene Si face that has the required structural order. Disorder prevents these effects to be observed in other forms of graphene [83-85](for a review see Basov et al. [86] and refs therein). For instance Figure 10 presents non-linearities close to the Dirac point and a kink in E(k) in epigraphene Si-face and potassium-doped quasi-free standing epigraphene. These have been explained by a band renormalization due to many-body interactions (electron–plasmon coupling, plasmarons - see Section 6.4)[84]. Note that in another set of experiments the kink feature was ascribed to a substrate-induced gap.[87]

4.1.1. Buffer layer

The existence of the buffer layer (also called the zeroth layer or nanomesh) was not known in the earlier literature. LDA calculations predicted its existence in 2007 [88, 89] and experimental verification soon followed [90, 91]. The buffer layer is a graphene layer that is bonded to the SiC. Little is however understood about its structure, stability, electronic properties and the nature of the bonding. While, LEED shows a $6\sqrt{3}$ periodicity (Figure 9b), early STM measurements showed only a 6x6 reconstruction (see Figure 11a). [92-96] [97] Later STM measurements, using a lower tunneling voltage of ~0.2eV, revealed the $6\sqrt{3}$ periodicity (Figure 11b) [98]. This higher order reconstruction is poorly ordered (with coherent domain sizes of only a few $6\sqrt{3}$ unit cells i.e.~5-6nm). This explains why surface X-ray diffraction (SXRD) finds no evidence of a $6\sqrt{3}$ structure [99]. STM experiments also show groups of adatoms (possibly trimers) at the interface of UHV grown buffer.[90] These adatoms may be responsible for the surface reconstruction, although other studies rule out any significant presence of Si within the interfacial layer [100].

Riedl et al. [91, 101] demonstrated that the buffer layer is structurally a graphene layer. They showed that by intercalating hydrogen between the SiC and the buffer layer, the buffer layer transforms reversibly to a normal graphene layer (Figure 8c). This newly formed graphene layer is simply vertically displaced by 2.1Å along with the other graphene layers on top, if there are any, as shown by high-resolution X-ray reflectivity [102]. Moreover, when the hydrogen is intercalated in a monolayer graphene film, the monolayer graphene electronic structure transforms into that of an AB-stacked bilayer (Figure 8d).

The decoupling of the buffer from the substrate by hydrogen intercalation is an example of the ease that foreign atoms intercalate at the SiC-buffer layer interface. Intercalation and buffer-layer decoupling have been demonstrated with Si [103], Ge [101, 104], O₂ [105], H₂O [106], Au[107], Li [108], Pt [109], F [110] and others.

The crystallographic structure of the buffer layer is essentially that of graphene, but interactions with the substrate transforms it into a semiconductor. Early band structure work[35] [78] on UHV grown samples indicated a large band gap, and two non-dispersing surface states g_1 and g_2 are observed [78] located 0.5 eV and 1.6 eV below the Fermi level (see Figure 11d): the distance from the valence band maximum to E_F is ~2eV indicating that the band gap is at least 2eV wide. However, STS measurements of the band gap shown in Figure 11c gives a much smaller gap value of ~0.4eV. [90] This discrepancy remains to be resolved.

The large unit cell size of the buffer layer impedes an exact determination of its structure. Based on photoemission spectroscopy measurements of the C1s core level, it was argued [78] that one third of the carbon in the buffer layer is bonded to Si in the SiC below. A more detailed x-ray standing wave enhanced XPS (XSW-XPS) measurement [100] indicates that of the two C1s peaks corresponding to the buffer, one (S1 at 284.75eV binding energy) is attributed to C-C bonds in graphene, and the other (S2 at 285.55 eV binding energy) is consistent with C-Si bonds. The S1 carbon is 2.39Å above the last Si layer in the bulk SiC while the S2 carbon is closer to the bulk at 2.07Å. The distance of the buffer to the SiC surface is in agreement with transmission electron microscopy [111] and earlier x-ray reflectivity experiments that put a non-flat graphitic layer (0.16 Å rms thickness) 2.3 Å above the last SiC layer [112].

The in-plane lattice constant of the buffer layer is found [113] to be tensile strained relative to graphite. What is surprising is that when the buffer layer is covered with graphene, not only is the buffer layer's strain reduced by more than half, but the buffer and monolayer lattice parameters are incommensurate with each other (epigraphene is compressed compared to graphite). Clearly more experimental work will be required to clearly understand the buffer-SiC structure.

Theoretical predictions of the structure of the buffer or monolayer on the Si-face have been similarly limited by the long computation time when dealing with a very large $6\sqrt{3}$ unit cell [88, 89] Calculations of a fully relaxed $6\sqrt{3}$ graphene-SiC cell starting from a bulk terminated (0001) surface [114, 115] have shown that a superhexagonal mesh develops that is at least consistent with the 6x6 periodicity observed in STM. [93, 98, 115] While the calculated bands [114] show some features that are similar to the measured bands of Emtsev et al. [78] in Figure 11d there are a set of distorted π -bands that appear near the K-point. In the calculations, these bands are due to the π -bands of carbon atoms in the buffer layer that are not bounded to Si atoms in the SiC surface. The non-bonded

carbon atoms form a superhexagonal network seen in STM (Figure 11b). Whether or not the predicted superhexagonal distorted π -bands can be observed experimentally remains to be determined.

4.1.2. Growth mechanism

The growth of epigraphene is unusual because it proceeds by decomposition of a crystal rather than by adding atoms to a surface. Very few theoretical studies have addressed this question [116, 117]. STM studies [90, 92, 118, 119] agree that the top layer is continuous and that subsequent graphene layers growth below it. STM, AFM, LEEM and cross-sectional Transmission Electron Microscopt (TEM) observations show fingerlike structures on the Si-face (Figure 12a). Based on these observations, growth models have discussed graphene nucleation at step edges. The terraces recede to remove enough Si to form graphene. The key assumption is that the decomposition of a SiC step edge depends on the local curvature of the front, so that the growth is understood from a competition between capillary smoothing and a decomposition driven step edge roughening (see Figure 12c-d). This is described by a linear stability analysis of a step equation of motion as a function of growth process variables: temperature, background Si and inert gas pressure. [116] It has been argued that epigraphene growth is diffusion-controlled and therefore naturally leads to instabilities (such as the finger-like features ofFigure 12a) instead of the advancement of straight steps. [120] However, simple step flow growth can occur for a step height of three SiC bilayers, which correspond almost exactly to the number of carbon atoms required to form a graphene layer [120]. As seen in Section 4.5, patterned steps have been introduced to direct graphene growth at step edges.

Models were further developed [117] for graphene growth on non-planar nano-faceted 6H–SiC substrates; the model parameters are the effective energy barriers for the nucleation and propagation of graphene at the SiC steps. The main result, in agreement with kinetic Monte Carlo simulations, is that the original nano-facet is fractured into several nano-facets during graphene growth, which is related to terrace width distribution. [117].

Because the growth of epigraphene is driven by Si evaporation, atomic transport of silicon and carbon on graphene and in-between layers is critical to the understanding of multilayer growth processes. Results from density functional theory calculations show that Si atoms can move almost freely on graphene and between graphene layers, while C atoms have much larger diffusion barriers. The results provide an explanation of the high Si sublimation rates during the growth of epigraphene even after graphene layers are formed on the surface.[121]

4.2. C-face

The main difference between graphene on the Si- and C-face is as follows. Graphene grows very fast on the C-face compared to the Si-face. With the CCS method [34] 5-10 layer films are typical and films thicker than 20 layers are easily grown while single layer films are much harder to achieve [34, 122-125]. There is no evidence of a buffer layer on the SiC(000-1) surface. However SiC interface reconstruction may occur, as the 2x2 and 3x3 reconstruction observed in UHV-grown samples [126]. Subsequent layers show a mostly ordered stacking with a distribution of relative rotations alternating around 0° and 30° rotations. These various rotations correspond to graphene/ SiC commensurate structures. This is clearly shown in the C-face graphene LEED pattern in Figure 9c, where diffuse intensity arcs are seen in stead of the sharp hexagonal pattern of diffraction spots seen in Si-face graphene.

4.2.1. Microstructure

In well-formed multilayer samples [34] the top graphene layer is continuous and drapes over the steps on the SiC substrates and no grain boundaries are observed in STM [127]. AFM and STM images show that the graphene film has pleats (also called folds, puckers, ridges, creases, rumples or wrinkles in the literature) [49, 128]. The pleats are typically 1–10 nm high, are typically spaced 3–10 μ m apart and are thought to result from the differential expansion of the silicon carbide and graphene and the very weak coupling of the graphene to the substrate. The monolayer pleats can be easily displaced by an AFM tip, in a manner of nanoscale ironing, showing the weak graphene – substrate interaction.[124]

Estimates of the coherent domain size from Raman, STM, and transport measurements indicate that graphene domains are large, 400–1000nm [49, 53], with a lower limit of 300 nm as observed in surface x-ray diffraction (SXRD) [129] (resolution set by the SiC steps that destroy the x-ray coherence). The film roughness is also

extremely small, with atomically flat terraces [128] (in contrast to the nm roughness typically observed in transferred CVD films) (see Figure 13a, b, e).

4.2.2. Growth

There is indication that graphene can also nucleate on C-face terraces [123, 124, 130], in contrast with Si-face graphene growth that is mostly promoted at step edges. Graphene is more readily formed where Si can escape easily from the SiC substrate, that is at defects[130], screw dislocations in SiC [71] and holes in the graphene film acting as "volcanoes" [131]. Because of this, although large flat and uniform areas are possible (Figure 13c), thickness uniformity is more difficult to achieve on the C-face and thicker films tend to show more thickness variations as demonstrated by LEEM IV analysis[132] (Figure 13d).

Various scheme have been developed to realize a more uniform growth on the C-face. For large-area graphene, the key is to monitor the Si out-diffusion to grow graphene in near equilibrium condition, either by confining the Si vapor in an enclosure (CCS growth [34, 49], see Section 3.4), by applying a noble gas back pressure (similar to the Si-face in [66]), limiting the Si escape with a graphite cap [124] or in between 2 SiC chips [133], or maintaining a Si partial pressure with a external flux of Si (similar to Si-face work in [72]). Excess graphene was also etched during growth with H₂[125]. Another approach is to functionalize the SiC surface prior to growth, and it was shown that nitrogen –seeded 4H-SiC surface induces more thickness uniformity on top of C-face narrow mesas.[134]

From a device perspective, it is obviously advantageous to grow nanostructures at predefined locations. To this end structured growth methods have be developed, the most promising of which is to etch patterns into the SiC substrate prior to growth, that serve as template for graphene nanostructures [135, 136]. Preferential growth was also induced in pits and mesas [137] (see also improved uniformity on the Si face of mesas [138]). Capping AlN [139] or removable SiN masks [140] have been successfully used to prevent or enhance graphene growth in the masked area.

4.2.3. Rotational stacking

Unlike on the Si-Face, graphene on the C-face is not Bernal stacked. It must be emphasized that this C-face graphene stacking is not "turbostratic graphite". In the literature, turbostratic graphite refers to disordered graphitic materials composed of non-Bernal stacked platelets that are typically on the order of 10 nm in size [141]. These platelets are randomly distributed in an otherwise AB-stacked film.[142, 143] In contrast, in CCS grown C-face graphene multilayers are ordered and very large (order of at least 10's of microns). The first graphene layer that forms is rotationally aligned 30° relative to the SiC. A \pm 7° rotation is occasionally observed [127]. Thicker layers exhibit well-ordered rotational stacking. SXRD experiments shows that this stacking consists of alternating 30° and $0^{\circ} \pm \sim$ 7° rotated graphene sheets approximately every other layer. Rotational angles around 0° correspond [144, 145] to commensurate structures between graphene and the SiC(000-1) surface. LEED pattern of the rotated layers produces arcs centered at 0°, which were mistakenly attributed to small disordered grains of HOPG graphite [16, 146]. These observations indicate that properly produced and annealed C-face multilayers are ordered, while otherwise produced multilayers exhibit disorder, that can be extreme. For example, it has been reported that C-face graphene grows as small graphite crystallites similar to HOPG graphene.[147] However, the samples used in those studies were grown in argon at much higher temperatures (1800°C-2000°C). This turbostratic structure is most likely due to an amorphous SiC interface layer that forms at these high growth temperatures.[148]

In C-face multilayered graphene it was estimated that AB stacked layers represent less than 19% of the layers in the film.[144] STM,[149] [141] ARPES,[144], μ -LEED[150], Raman spectroscopy[53] and electronic transport[49] all confirm these results. In STM, these commensurately stacked layers show up as moiré patterns with large supercell sizes. Figure 14c-e show an STM image from C-face films showing moiré patterns from three graphene layers commensurately stacked above each other. μ -LEED presents these commensurate rotations as diffraction patterns with a large supercell (see Figure 14a-b). The supercell size varies over several micron length scales but the top layer orientation is always 30° relative to SiC. This indicates that while the commensuration angle varies in the layers below, the top layer is continuous over very large areas.

4.2.4. Electronic structure

The electronic structure from C-face graphene is very different from Si-face graphene. Whereas ARPES of Si-face few layer graphene exhibits overlapping, interacting Dirac cones (see Figure 2), ARPES of C-face multilayer graphene exhibits multiple undistorted Dirac cones that are mutually displaced, corresponding to individual graphene layers of the rotational stacked film. Figure 15a shows the band structure from an 11-layer C-face film measured in ARPES. [9] Two rotated Dirac cones are clearly visible. Note also that the graphene Dirac point is very near E_F . The limited mean free paths of the photoelectrons in the ARPES experiment only allows the top three layers to be probed. The data shows that that these three layers are undoped (since the Dirac points coincide with the Fermi level) and that the band structure of each layer corresponds to that of a free graphene sheet. (The band structure of bilayer graphene is only rarely observed.)

Standard area-averaged ARPES cannot distinguish whether the two cones in Figure 15a are due to a stack of two graphene sheets or from two uncorrelated graphene sheets that are laterally separated by a distance within the 50 μ m ARPES beam diameter. k-PEEM (a version of micro-ARPES) only measures the band structure within the illuminated area. Figure 15b shows the k-PEEM from a 2-layer C-face film using a 7 μ m illumination aperture.[151] Besides the two mutually displaced (rotated) primary Dirac cones, a third set of Dirac cones is also visible. The third set of cones results from an umklapp process, that translate the Dirac cone of one layer by the reciprocal lattice vector of the second layer and visa versa (see Figure 15b) as shown by Mathieu et al.[151] . The observation of this interaction proves that the rotated graphene sheets must be indeed stacked on top of each other. The signal specifically cannot be due to separate rotated grains in the beam.

The observation that C-face graphene multilayers are electronically decoupled has motivated a number of theoretical studies. [129] Interesting effects are anticipated because the stacking that gives the moiré pattern lifts the degeneracy between sublattices in a non-trivial way. In particular a strong reduction of the Fermi velocity is predicted for small rotation angles, both with ab initio and tight-binding methods [152-156]. Interlayer couplings between the rotated layers may introduce a singularity in the energy spectrum due to geometric considerations. [155], and observed peaks in the STS local density of states that scale with the rotation angle were interpreted as such (Van Hove singularities) [157, 158] The fact that thick C-face graphene films are electronically similar to a stack of independent, mutually rotated graphene sheets, warrants the accepted nomenclature multi-layer epigraphene, or MEG.

4.2.5. Raman spectroscopy- thickness determination

Raman spectroscopy, that is sensitive to electron-phonon coupling, is a widely used technique to characterize graphene. The spectrum typically presents three peaks, labeled D, G and 2D (see Figure 16). Since the D peak is only Raman active when the lattice symmetry is broken, its presence is indicative of structural defects (e.g. point defects and edges). Figure 16 a shows the Raman spectrum for thick MEG. Figure 16 b shows the Raman spectrum of a C-face graphene monolayer before and after subtraction of the SiC Raman background signal (that is invisible in Figure 16 a due to the film thickness) [159]. The absence of the D peak attests to the quality of C-face epigraphene monolayers and MEG.

The 2D peak profile evolves from a single Lorentzian in graphene to a characteristic shouldered wide peak in graphite. Therefore, the shape and width of the 2D peak are often used to determine the number of layers in thin (Bernal stacked) graphite films [160] as well as in Si-face graphene films, where graphene layers are Bernal stacked. However, on the C-face, the layers are electronically decoupled and the 2D lineshape does not evolve with increasing thickness but remains a simple Lorentzian peak [53](see Figure 16a) which profile cannot be used to determine the film thickness.

The relative intensity of the graphene Raman peaks compared with those related to SiC increase with the number of layers, but these ratios are difficult to calibrate accurately [161, 162]. In LEEM, the interference of the incoming electrons with the electrons that are reflected from the SiC–graphene interface produce oscillations in the reflected electron intensity. The number of layers equals the number of oscillations. [163]. While accurate, this method is complex and therefore cannot be used as a routine diagnostic tool. Auger spectroscopy provides an accurate

measurement of very thin epigraphene layers by comparing the ratio of the Si peak intensity with that of C. For thicker layers, ellipsometry gives a very good estimate of the average number of layers [164].

For graphene monolayers, the position of the Lorentzian 2D peak depends on the electronic doping density and strain in graphene [165-168]. In quasi-free standing epigraphene on the Si-face, the 2D peak position is constant within about 1cm⁻¹ for electron doping n<5x10¹²cm⁻² and shifts by 4 cm⁻¹ for p-doped graphene [168]. Shifts in the 2D peak position for C-face monolayers have been attributed mainly to strain and to a much lesser extend to doping density [137]. Note that an unexpected Raman intensity enhancement is observed by collecting light through the SiC substrate, in a reverse configuration. The effect is explained it in terms of dipole radiation at the dielectric surface. [168].

However, besides sensitive to strain and doping, shifts in Raman peaks in graphene are determined by a subtle interplay between the phonon and electron energy dispersions [169]. The 2D peak in Si-face epigraphene is blueshifted compared to exfoliated graphene [170] which indicates compressive strain in epigraphene. But, actually the lattice parameter a=2.456Å of epigraphene Si-face [113] is larger than the theoretical value for graphene, indicating expansion. It would be instructive to know the lattice parameter of transferred graphene and compared to the free graphene value. Comparison with HOPG is more subtle because the 2D peaks is composed of 4 peaks reflecting the more complex graphite electronic band structure.

4.3. Epigraphene on other faces

There are very few epigraphene studies on other than those on the (0001) and (000-1) faces, due to the lack of available substrates. Contrary to the Si- and C- faces, theses surfaces are non-polar, i.e. they have an equal amount of silicon and carbon atoms, which is quite interesting regarding the difference in graphene growth and morphology between the 4H/6H-C- and Si-faces. As expected graphene was found to grow on the two perpendicular surfaces, so called a-(11-20) and m-(1-100) faces. From XPS, LEED, LEEM, and in agreement with density functional theory calculations, it is concluded that there is no buffer on either face. [171] Differences were observed between the two faces. On the SiC(1-100), micro-LEED reveals multiple rotations similar to C-face graphene (although the stacking has no been studied), while on the SiC (11-20) face, graphene grows with a single orientation. ARPES shows the typical conical bands of graphene, with a negative charge density of 4×10^{12} cm⁻². This is unexpected since the graphene doping density in the absence of buffer (quasi-free standing graphene) was argued [172] to be due to spontaneous polarization in the polar Si and C-faces. The charge density therefore could be attributed to the work-function difference between graphene and silicon carbide. Transport properties confirm that lightly doped single layers can be grown on the (11-20) face [173], however the heavily doped substrate confine measurements to very low temperature.

4.4. Epigraphene on 3C-SiC

Graphene layers have also successfully been grown on cubic β -3C-SiC (Figure 6), both on the hexagonal (111) and the cubic (001) face of commercial polycrystalline substrates or overlayers grown on silicon or 6H-SiC substrates. [174-179]. Note that the commercial production of 3C-SiC substrates is limited. In contrast to the 4H and 6H-SiC, large single crystal 3C-SiC is not available.

Studies on 3C-SiC are nevertheless interesting in several respects. In contrast to hexagonal SiC, 3C-SiC does not spontaneously electrically polarize (see Section 5.1.2), so that graphene on 3C-SiC is found to be quasi-neutral (see Figure 17b)[176, 180]. Step free growth and large area monolayer coverage have also been reported for 3C [179] (Figure 17a). Since thin layers of 3C-SiC can be epitaxially grown on silicon, this has been proposed as a strategy for integration of SiC based epigraphene with Si-based electronics (see Figure 17c and Section 5.3.3).

For 3C-SiC epilayers on silicon substrates, the polar Si-terminated 3C-SiC(111)/Si(111) surface grown in UHV shows graphene Bernal stacking with an interfacial buffer layer, similarly to the 4H- or 6H-SiC(0001) surface. Conversely, the C-terminated 3C-SiC(111)/Si(110) shows a non-Bernal stacking, with the absence of an interfacial buffer layer, consistent with a C-face termination. The quality of these graphene films is poor as shown by large Raman D peaks. The disorder results from Si diffusion through SiC grain boundary (due to a large ~ 20% lattice

mismatch between Si and 3C-SiC) and the lower graphene growth temperature (limited by the Si melting point at 1414°C). Growing an epitaxial AlN layer on Si prior to SiC growth significantly reduces Si out-diffusion and helps grow higher quality epigraphene. [175]

Graphene was also grown on the 3C-SiC(001) surface, which demonstrates that a hexagonal template is not required for the growth of graphene. LEED patterns show no evidence for a buffer layer (growth in Ar at 1800°C). [181]

4.5. Nanostructured graphene

From the outset of epigraphene electronics research it was clear that graphene nanostructures would be required [1, 2]. Moreover, these structures would necessarily need to be as defect free and reproducible as possible, in order to serve as building blocks for electronic devices. Early calculations [61, 62] had already predicted that quantum confinement would play an important role in graphene ribbons. Depending on the edge structure a graphene ribbon would be semiconducting or metallic. Those calculations predicted that all zigzag ribbons would be metallic while armchair ribbons would alternate between being metallic or semiconducting, depending on their widths (see Section 2.9).

Early work on exfoliated and on epitaxially grown graphene focused primarily on graphene nanoribbons produced by electron beam lithography and plasma etching. Epigraphene nanoribbons were found to be metallic exhibiting quantum confinement effects at cryogenic temperatures [49], which implied that edge scattering was reasonably coherent. Disorder in the ribbon was not large enough to cause localization. On the other hand, charge neutral exfoliated e-beam patterned nanoribbons were invariably found to become insulating at cryogenic temperatures, implying a band gap. In fact, the magnitude of the band gap was determined to be consistent with Eg=1eV nm/W as expected for the quantum confinement gap in specific armchair ribbons. However it was soon found that the low temperature insulating behavior was due to a transport gap that opened due to charge puddles in the ribbons combined with edge disorder [182-187].

To improve edge order, "bottom-up growth" methods were developed to grow graphene nanoribbons. [135, 188, 189] Very narrow graphene ribbons have been grown at steps on Au(788) facets by catalyzing molecular precursors into linear polyphenylenes. [188, 189] While the edge order is essentially perfect [189] so that the armchair edge ribbons are consistent with the predicted [62, 190] finite size band gap, scalability as required for electronics is still lacking. Moreover, the growth is limited to metal surfaces and it is expected that the transfer process to semiconducting substrates will face the same problems as other transferred CVD graphene films. Hence, these efforts have largely been abandoned for large-scale electronics.

Epigraphene, however, provides another solution by growing ribbons on lithographically patterned steps etched into the SiC substrate [135, 137, 191], and exploiting the fact that graphene growth proceeds first on the facet walls of natural step edges on the Si-face of SiC [192]. In order to produce nanoribbons, trenches are first etched in SiC that serve as a template for graphene growth. This allows ribbons and other nanostructures to be accurately defined on a substrate as required for nanoscale integrated circuits. The method allows thousands of parallel ribbons as narrow as 20nm and with well annealed edges to be grown at once over mm2 area (see Figure 18), No further (potentially damaging) post growth processing of graphene is required. In this process, the ribbon width is defined by the trench depth, that is very well controlled using standard plasma SiC etching processes. [135]

Epigraphene also has a distinct advantage because Si-face graphene is epitaxial with the SiC(0001) surface, therefore the ribbon orientation is also predetermined. The graphene's zigzag (ZZ) or armchair (AC) edges, naturally align with the SiC step edge simply by etching steps in the SiC in a given SiC crystallographic orientation. When a trench in SiC is oriented perpendicular to the $\langle -1100 \rangle$ direction, the graphene that grows has its AC edge parallel with the step edge. Trenches perpendicular to the $\langle 11-20 \rangle$ direction produce ZZ ribbons. For convenience, we will refer to these SiC step edges as AC edge steps (or AC facet walls) and ZZ step edges (or ZZ facet walls) (see Figure 19). Structured sidewall growth has been used to produce a wide variety of shapes (see Figure 18(c-f) [191]. Of particular interest are pillars (or circular pits) around which graphene rings are grown for quantum interference pattern experiments [191].

The structured SiC chip is then heated to 1100 C to allow the vertical etched sidewalls to crystallize into the equilibrium facets onto which the graphene ribbons grow around 1500-1600 C. The exact temperature depends on the specifics of system, such as the Si escape leak in the furnace CCS method [34], or the face to face geometry in the current annealing growth [193, 194].

4.5.1. Armchair Edge Sidewall Ribbons

The structure of AC sidewall ribbons has been studied extensively with a number of techniques. AFM shows that the trench walls for AC steps facet into planes with an average facet angle of 28-30°. Early TEM studies of graphene growth on natural AC step edges used exceptionally thick Si-face films (>4 layers).[192, 195]. The graphene was found to tend to nucleate at the bottom of the step edges so that graphene grows thicker on the step facet than on the (0001) surface (Figure 20a). Several edge terminations are observed (see Figure 20a-b).

Under certain CCS growth conditions cross sectional transmission electron microscopy measurements show mini facets near the top edge and bottom of the primary facet, where the sidewall merges into the (0001) surface on top and bottom of the sidewall.[111] The composite scanning transmission electron microscopy (STEM) image of Figure 20c shows that the graphene ribbon continuously drapes over the entire sidewall. The ribbon appears to be anchored to the edges (or mini terraces) at the mini facets. At those places, the SiC-graphene distance is slightly smaller than that of the SiC-buffer layer spacing. Note that nm-wide ribbons have been predicted at SiC step edges on 6H-SiC.[117]. Ultimately, the graphene ribbon merges into the buffer layer on top (and often at the bottom) of the sidewall

ARPES from the sidewall ribbons gives additional information on both the electronic and topographical structure of these ribbons. [196] Since the facet wall normal is tilted relative to the (0001) plane, the graphene on the facet is tilted by the same angle (see Figure 21a). The step edge the K-point of the facet graphene can be measured (see Figure 21a) by appropriately rotating the sample. ARPES shows that the graphene that grows on the sidewalls has a different doping than the graphene on the (0001) surface. While the monolayer graphene on the (0001) is n-doped by ~0.45eV, the graphene on the facet wall is only slightly p-doped (<70meV). The facet angle can be determined from the relative rotation angle between the (0001) Dirac cone and the facet Dirac cones. The facet contains both (-1106) and (-1107) planes, similar to those seen in the STEM studies in Figure 20c.

In addition to the (0001) surface and sidewall Dirac cones, there is a range of angles, consistent with the position of the mini ribbons, where the set of cones appear to be gapped (see Figure 21d). [196] LDA calculations of bent wellordered nm-wide ribbons predict an energy gap in good agreement with the observed ARPES and TEM data. Whether the origin of the gap observed in ARPES is due to quantum confinement in nm-wide ribbons or related to other effects (like buffer graphene near the step edge or the mini-ribbon (0001) terraces that are tightly bound to the SiC substrate), it is nonetheless remarkable that a band gap opens in a structure entirely made of graphene (see Section 5.4). In any case, the measured gap confirms that these nanoribbons have smooth edges, comparable to the atomically precise ribbons fabricated by bottom-up chemistry [189].

4.5.2. Zig-Zag Edge Sidewall Ribbons

Several structural studies have been focused on graphene growth on commercial off-axis SiC. After annealing, the miscut combined with step bunching produces arrays of ZZ substrate steps [197, 198] to produce ZZ sidewall ribbons with typically >3 layers. Electron energy loss spectra in STEM indicates that the graphene on the sidewall is decoupled from the SiC (see Figure 20d).[198]

Single graphene layers can be grown on ZZ sidewalls. Transport measurements made on ZZ ribbons produced by current annealing have been found to be metallic and even demonstrate ballistic transport properties [136]. However, ARPES measurements of arrays of ZZ ribbons do not show a Dirac cone, while a Dirac cone is observed for AC ribbons that are produced using the same CCS method. The absence of a Dirac cone in this case was originally attributed to bonding of the graphene layer to the ZZ sidewall facet to produce a buffer layer [199]. It is however also possible that, depending on the growth condition, the ZZ annealed sidewall is not straight because natural SiC ZZ steps have a known instability towards nano-facetting into local AC steps [200] (see Figure 22a-b). A similar

instability has been seen in patterned ZZ-edge steps in SiC [199], meaning that instead of a ZZ edge, a series of small AC nanofacets, $\pm 60^{\circ}$ relative to the average ZZ step edge direction, would form as shown in Figure 22c. If ARPES requires that the sidewalls are well ordered, which could explain why the graphene signal was not observed using those methods, the short order length scale necessary for the observation in PEEM indicates that the two different growth methods may lead to two different structures.

5. Electronic transport properties of epigraphene

As shown below epigraphene presents the characteristics expected for graphene. The large surface area and high structural quality allow for experiments that wouldn't be possible with small flakes or rough substrates and for large-scale integration. In some instances specificities are observed, in particular related to the original rotational stacking on the C-face. Here we only review electronic properties as they pertain to epigraphene, and the reader is referred to review articles for general graphene electronic properties that are beyond the scope of this article.

As explained in the introduction, what made graphene attractive in the first place is its potential for electronics. Extremely high electronic mobility were anticipated from the onset [1], based on the room temperature ballistic conduction of the closely related carbon nanotubes [201]. By choosing the appropriate substrate, mobilites in the range of 100,000 cm²/Vs are achieved in 2D graphene on boron nitride [60], up to more than 10⁶ cm²/Vs at room temperature in the neutral rotated epigraphene C-face layers [202] or for graphene on graphite at low temperature, and unrivaled room temperature ballistic conduction up to 15 μ m in epigraphene nanoribbons. Graphene offers also the possibility to vary the charge carrier and thereby the conduction by electrostatic gating [1, 4, 5] and patterning by standard lithographic techniques is available.

5.1. Transport properties

5.1.1. Charge density

The first graphene layer above to the SiC interface is negatively charged, as shown in the ARPES measurements presented in Section 4.1. For the Si-face, the n-doping of about 10^{13} cm⁻² as measured by photoemission spectroscopy [79, 84] is believed to arise, at least partially, from interface states associated with the SiC interface and the buffer layer.[172] Similar doping of a few is 10^{12} cm⁻² is found on the C-face [131] by photoemission in UHV. Because of screening on a length scale of about one layer, the charge density decreases from one layer to the next as observed in ARPES [131], in optical spectroscopy [203] and electron energy loss [204] spectroscopy experiments. More precisely, a charge profile consistent with $n_l=n_{le} e^{-1.1(l-1)/\lambda}$ was determined in C-face multi-layered epigraphene by mid-infrared pump-probe spectroscopy, where l is the graphene layer index from the SiC interface, $n_{le}=9.6 \times 10^{12}$ cm⁻² and $\lambda=1.2$ [203]. A full calculation can be found in Ref [205]. As a result, the layers away for the SiC interface in multilayer C-face are quasi neutral and present the properties of ideal graphene close to the Dirac point, as close as 8 meV, that is $n \sim 5x10^9 \text{cm}^{-2}$ (see Section 5.1.3 and 5.2.2).

For graphene exposed to air, the top layer shows sign of contamination by the environment, that results in a counter positive doping. The measured charge density changes sign from negative to positive for epigraphene monolayer C-face left in air typically within an hour, after that time it stabilizes. [123] It has been consistently observed that the effect is reversible by heating above $80-100^{\circ}$ C. After air exposure, ambient-environment Kelvin probe microscopy of Si-face single-layer epigraphene indicates a highly uniform carrier concentration of the order of 10^{12} cm⁻² with very small carrier fluctuations $\sim 10^{10}$ cm⁻² (corresponding to measured rms surface potential variation of 12 meV) [206].

On epigraphene Si-face, when the SiC interface is saturated with hydrogen and the buffer layer becomes the first graphene layer (see Section 4.1.1) a consistent p-doping is observed. The positive sign and the charge carrier doping level are explained by a polarization doping arising from the bulk spontaneous polarization of the pyroelectric SiC substrate [172]. This is because at the interface with graphene, this polarization is equivalent to a negative charge that is balanced by a positive charge in the graphene layer, and to a lesser extend by a positive space charge in the SiC substrate depletion layer. When the epigraphene layer resides on the (non H2-lifted) buffer layer , the presence

of donor-like states associated with the SiC/buffer interface is believed to overcompensate the positive polarization doping yielding the measured n-doping of epigraphene Si-face. The polarization doping model was further confirmed by ARPES measurements of the charge densities of quasi-free standing graphene on 6H-SiC(0001), 4H-SiC(0001) and 3C-SiC(111). [180] The SiC spontaneous polarization, and accordingly the epigraphene charge density, decrease from 4H to 6H and to 3C-SiC so that quasi free standing graphene on 3C-SiC is quasi- neutral.

5.1.2. Square resistance, mobility and charge density in single layers

In two-dimensional systems, transport is characterized by the square resistance defined by $R_{sq} = (V/I) W/L = (\rho/t)$ where V is the voltage difference between the voltage probes, I the current through the sample, W, L, t are the sample width, length and thickness respectively and ρ , the material resistivity. An important property of graphene that sets it apart for ordinary metals is that the charge density can be varied by orders of magnitude, inducing a large change in R_{sq} . The square resistance of clean graphene, hereafter simply called resistivity, goes through a sharp maximum at charge neutrality (transport at charge neutrality goes beyond the simple one electron picture). The maximum resistivity value and peak sharpness as a function of charge density depend on the graphene structural quality and source of scattering. In the simplest picture of short range scatterers, a conductivity independent of carrier density was expected because of the linear in energy density of states $\mathcal{N}(E)[207]$ [208]. The resistance maximum calls for other scattering mechanisms, including charge impurity scattering, resonant scattering (for a review of transport mechanisms, see for instance Ref. [208]).

For "dirty" graphene samples, at energies away from the charge neutrality point, the conductivity rises linearly with the charge density n [208, 209] $\sigma = \sigma_{res} + Ce(|n/n_{imp}|)$. Here n_{imp} describes the charge impurity concentration, and

 $\sigma_{\rm res}$ reflects the fact that graphene conducts even at zero carrier density. The sharpness of the conductance minimum can phenomenologically also be captured with $\sigma = e\mu \sqrt{n_{imp}^2 + n^2}$ [210].

Note that with the usual definition σ = neµ, the mobility µ would reach unphysically large values at charge neutrality. Indiscriminate and improper use of mobility equations implies that the mobility is infinite at zero charge density, while in fact, the actually mobility (taking into account the charge disorder) is actually much more modest. In disordered samples zero charge density arises from compensated puddles of electrons and holes and transport proceeds by hoping from puddle to puddle.

When the graphene is not too close to charge neutrality, the mobility μ can nevertheless be determined from the measurement of $\sigma = 1/\text{Rsq}$ and $n = 1/(\text{Be}\rho_{xy})$ where ρ_{xy} is the transverse (Hall) resistivity in the linear regime of magnetic field B ($\mu_H = \rho_{xy}/\text{ B.Rsq}$).

Note that field effect mobilities are often reported, which are determined by modulating the charge density on either the Si or C faces (see also Section 5.3.1). For this, techniques include counter-doping in air [123], with molecules (such as Tetracyanoquinodimethane [101]), with a charge induced silicon nitride top dielectric coating [211], by metal intercalation at the interface (see Section 4.1.1), lifting the buffer layer by hydrogen intercalation to saturate the dangling bonds on the SiC substrate [91]. In a more controllable way, the charge density can be varied by applying a voltage V_g to a top electrostatic gate (standard planar capacitor) as widely used for epigraphene high frequency transistors (see Section 5.3.1). For this, epigraphene is coated with a thin dielectric (Al₂O₃, HfO, SiN, ...) plated with a metal (typical Au or Al) [212] [137, 213]. The (surface) charge variation Δ n induced on graphene is Δ n = (ϵ O ϵ /td) Vg/e, where ϵ and td are the dielectric constant and thickness of the dielectric respectively. Examples of modulation of the resistance of epigraphene are given in Figure 23. Top gating was also achieved, with a UV light photochemically sensitive polymer [214] and electrolytes [215]. The mobility is related to the slope of the transconductance dI_{sd}/dV_g for a graphene strip where a current I_{sd} flows between source and drain contacts; this is the configuration of a field effect transistor (FET).

Because deposition of a dielectric on graphene tends to reduce the mobility, attempts were made to back-gate epigraphene [216]. High energy nitrogen implantation into SiC produces a thin conducting layer buried into the insulating SiC substrate, to which a voltage can be applied. However interface states limit the gate efficiency and efficient doping was observed only for quasi-free standing epigraphene (H2 passivated). Besides, the gate is inefficient at low temperature (the carriers in SiC are frozen in) and at room temperature it relies on the Schottky-like contact between conducting SiC and graphene ('Schottky capacitor' regime) [216, 217].

For epigraphene on the Si-face mobility values are somewhat limited and on the order of 1500 cm²/Vs at charge density of about 10^{13} /cm⁻², even for Hall bars patterned on single terraces. Reported mobilities increase dramatically by decreasing the charge density n_s, as expected, with record low temperature values of $\mu \approx 30,000$ cm²/Vs for n_s $\approx 5 \times 10^{10}$ cm⁻² [218, 219]. When the buffer layer is lifted by hydrogen intercalation and graphene becomes p-doped, higher mobilities are observed (roughly by a factor two). It is not clear what causes the increase in scattering when the buffer is present. As for the conductivity, high values can be reached with large charge density obtained with top gating or molecular doping: record low conductivity of 16Ω /sq was reported by intercalating the ionic conductor FeCl₃ [220]

Epigraphene on the carbon face shows higher mobilities ², despite contaminations and monolayers draping over multiple SiC steps [123, 137]. For charge densities of the order of a few 10^{12} cm⁻², μ =20,000 cm²/Vs was reported early on [123] and up to 39,800 cm²/Vs by reducing the charge density to n= 1.9x1011 cm- (Figure 27a). A top gate induces scattering. Nevertheless top-gated FET mobilities of 7,000-8,000 cm²/Vs are routinely found at room temperature [221] for n~ 1.5x1012 cm-2(see Figure 23b for instance).

5.1.3. Charge density and mobility in multilayers

For multilayers on the C-face, the graphene layers retain the electronic structure of single layer graphene, as discussed in Section 4.2.4. Mobility values of the quasi neutral graphene layers in the middle of the stack are the highest measured in any 2D graphene system to date, reaching μ =10⁶ cm²/Vs [202] for n=10¹⁰cm⁻² at room temperature and independent of temperature. [52] The mobility was estimated by infra-red spectroscopy measurements in magnetic field (see Section 5.2.2).

Square resistance of about 200 Ω are commonly measured for about 5 layers MEG on the C-face, with mobility of 20,000 cm²/Vs or more [49]. Because of interlayer screening, a top gate addresses primarily the top layer [212] (see Figure 23c).

For graphene on the Si-face, Bernal stacked layers have a quadratic band structure at the K and K' points (see Section 4.14.1). The electric field perpendicular to the layers that builds up due to of the intrinsic charge density difference causes a small gap to open [80]. This may explain the observation of massive carriers in thin epitaxial graphite in magneto-conductance measurements [222].

5.1.4. Scattering at SiC steps : the Si-face

For single epigraphene layer on the Si-face, high charge density is associated with moderate mobility of the order of 1,500 cm²/Vs. Several sources of scattering have been implicated including: the substrate, substrate phonons, SiC steps and single layer-bilayer junctions. Hydrogenated quasi-freestanding graphene shows high mobility, roughly by a factor two.

A significantly higher channel conductance is reported for devices patterned parallel to the intrinsic SiC steps direction compared to across the SiC steps [223] [224]. Similar anisotropic transport has been measured by local four-point probe and scanning probe techniques [225-227]. The anisotropy was attributed to various mechanisms. This may be related to a reduction of the carrier concentration for graphene on the sidewall of steps [136, 223], to a n/p junction at the step edge [196], to the opening of a band gap at the step edge [196] (see Section 4.5) or to the presence of bilayers multilayers on the steps (see Figure 8b). Conversely a careful surface preparation carried before graphitization to reduce the SiC step height notably increases the mobility [193] and very homogeneous device-to-device electronic properties (carrier concentration and mobility) are achieved at the wafer scale when the devices are fabricated entirely on single terrace monolayer domains.[69]

5.1.5. High current carrying capability

One of the most remarkable properties of graphitic materials is their high current carrying capability. For instance

electromigration breakdown was reported for exfoliated graphene at current densities of 1.6 mA/ μ m (corresponding to 5x10⁸A/ cm². For epigraphene in vacuum, conductance increases for current densities up to ~1.3x 10⁹A/ cm² due to local cleansing of the graphene channel. At higher current the sample glow reflects local heating above 1200^oC and SiC decomposes. The graphene film breaks down at a critical current density on the order of 1.3-2 x 10⁹A/cm² [228], to be compared to 3 x 10⁹A/cm² for carbon nanotubes [229].

5.2. Transport in magnetic field

5.2.1. Low field (anti)- weak localization

As in any metallic diffusive system, quantum interference effects are observed at low temperature. These are corrections to the classical conductivity arising from interference phenomena, when the electronic wave function keeps its phase for multiple scattering events (phase coherence length L\$\$\\$>\$mean free path l\$). These corrections have characteristic functional forms, which magnetotransport data can be fitted to in order to extract relevant transport parameters. In particular weak localization involves an enhanced probability for an electron to be coherently scattered back to its starting point in a loop trajectory. The enhancement originates from interferences between the clockwise and anti-clockwise trajectories. As a result, the resistance is up to twice as much as with no interference, and any mechanism producing incoherent scattering (temperature) or dephasing (magnetic field, spin scattering, spin-orbit) will destroy the interference and restore the classically lower resistance. Therefore a decrease of resistance is expected in a magnetic field (i.e. negative magnetoresistance that peaks at zero field).

Graphene adds an interesting twist to it. Because the character of the wave function in graphene cannot be changed by long-range scattering (only short-range scattering can locally destroy the equivalence of the A and B sublattices), 180° backscattering of electrons is forbidden and the resistance is low. Dephasing the wave function in a magnetic field can restore backscattering, yielding a positive magnetoresistance (so called weak anti-localization)[230].

Weak anti-localization in graphene was first observed in epigraphene grown on the C-face [50] for a high mobility sample (μ =11,600 cm2/Vs). As seen in Figure 24, the magnetoresistance can be fitted very accurately to weak anti-localization theory with only one temperature dependent parameter, ascribed to electron-electron scattering. Besides providing yet more evidence that multilayer epigraphene behaves like a stack of mono layers, these measurements are consistent with a dominant scattering at low temperature that preserves the pseudo-spin, most probably from long-range potentials arising from charges in the substrate.

On the Si-face, detailed analysis of low-field magnetoresistance and temperature dependence shows evidence for strong electron-electron interaction effects, giving logarithmic dependence. This is difficult however to detangle from the logarithmic dependence arising from the Kondo effect. Even by adding magnetic impurities studies so far are not conclusive. [231]

5.2.2. Landau levels spectroscopy on the C-face

As the magnetic field increases the energy levels bundle up in discrete (quantized) Landau levels. A simple picture is that the electron energy is minimized for orbits that are a multiple of the Fermi wavelength λ . An exact calculation for a standard 2D electron gas (2DEGs) gives $E_N = \hbar\omega_c(N+1/2)$ with the angular frequency $\omega_c=eB/m^*$ and N is the band index. Because graphene dispersion relation E(k) is not quadratic but linear, the energy of the Landau levels varies as the square root of magnetic field and N: $E_N = \pm c^* \sqrt{2\hbar eB|N|}$, where c* is the effective Fermi velocity. For a review of graphene in magnetic field see Ref. [232]. The square root dependence is characteristic of graphene, and there is no $\frac{1}{2}$ offset. The Landau index level N can take positive and negative values (the electrons and hole levels being identical). The N=0 level is special. It is field independent, pinned at $E_0=0$ and it is equally composed of holes and electrons states. The filling factor reflects the number of filled Landau levels.

The first direct observation of the characteristic $\sqrt{B|N|}$ energy dispersion in graphene was made in multilayered epigraphene by infrared magneto-spectroscopy [51]. Infrared light is absorbed (see Figure 25a) when its frequency matches inter-level separation (according to specific optical absorption rules, namely only transition LL -(N+1) \rightarrow

LN and LL-N \rightarrow LN+1 are allowed). Only quasi-neutral graphene can be probed in the infrared light and magnetic field range accessible. In C-face epigraphene, the energy versus field plot in Figure 25b displays text-book graphene behavior and the slope give c*=1.1x10⁶m/s, very close to the expected value for graphene (we refer the reader to [233] and [86] for a discussion of the renormalization of the Fermi velocity due to electron/hole excitation effects and substrate-screening effects). Electron-phonon coupling also manifests itself in the magneto-Raman in epigraphene C-face. The position and width of the Raman G band at 196 meV undergo oscillations (avoided crossings) at magnetic field values for which the G band energy crosses the optically active inter-Landau level transitions.[234]

It is significant that the $E_{\pm 1} \rightarrow E_0$ transition is observed at magnetic field as low as $B_0=40 \text{ mT}$ [52]. To resolve Landau levels requires that the level are minimally broaden by disorder. This simply means that an electron needs to complete a cyclotron orbit before being scattered. Because the cyclotron radius is inversely proportional to B, this translates into the criterion, $\mu B_0 > 1$. The low B_0 value gives extremely high mobility for the neutral MEG layers: $\mu > 250,000 \text{ cm}^2/\text{Vs}$, and from a direct measurement of the level broadening a value of $\mu = 10^6 \text{cm}^2/\text{Vs}$, was estimated at room temperature [52, 202]. Moreover, the level broadening varies linearly with energy [202], which provides some insight about carrier scattering; in particular it rules out extrinsic scattering mechanisms, such as resonant scatterers or charge impurities, in the epigraphene layers which are protected from the environment.

A great advantage of graphene compared to conventional 2DEGs is that the electron gas is exposed and not buried at the interface between two semiconductors. It is therefore readily accessible to direct imaging and spectroscopy. Local spectroscopic measurements of the Landau levels were performed using high resolution scanning tunneling spectroscopy (STS) on the top C-face layer at very low temperature (10 mK) and in magnetic field (up to 15 Tesla) [54] (See Figure 26). The Landau levels correspond to peaks in the STS local density of states (dI/dV spectra), and the graphene behavior is confirmed. The splitting of the four quantum states that make up a degenerate graphene Landau level (two levels per valley, two per spin) is studied in great detail as the magnetic field is increased (Figure 26b). Splitting due to broken valley degeneracy occurs first, followed by spin degeneracy as the field increases. Most unexpectedly, states with non-integer Landau level filling factors of 7/2, 9/2 and 11/2, are also observed suggestive of new many-body states in graphene. The graphene layers below the probed top surface may also influence STS results, by serving as a charge reservoir but also by screening interactions between top-layer electrons. They may also have a role in the observed fractional filling factor. [54] (see also [205])

Real space mapping of the two-dimensional spatial distribution of the electronic states of a Landau level was achieved for the first time on a 2DEGs by using scanning tunneling spectroscopy on the top C-face epigraphene layer. The local density of states mapping of the zeroth Landau level shows extraordinary fine details giving much insight into the nature of the Landau levels in graphene. [128] In particular unlike disordered patterns found in conventional quantum Hall systems, an organized pattern of localized states and extended states is observed in epigraphene C-face. STS measurements reveal local splitting of the zeroth Landau level corresponding to the local top bilayers AA, AB or BB stacking arising for the C-face rotational stacking [128]. This observation has important implications for transport phenomena. Spatial modulation of the sublattice symmetry and redistribution of charges between layers [205] may account for observations of fractal-like structure in magnetoresistance measurements of this material [49].

5.2.3. High field Shubnikov - de Haas oscillation and quantum Hall effect

When the magnetic field is increased, the Fermi energy E_F crosses the successive Landau levels E_N ; the density of states maximizes for $E_F=E_N$ resulting in resistance oscillations that are periodic in 1/B. (Schubnikov- de Haas oscillations). The oscillations are very well developed in epigraphene on both the Si-face [1] (Figure 27d) and on the C-face [49](Figure 27a). The charge density ns ($EF=\hbar vF\sqrt{\pi}$ ns) can be determined from the period of the oscillations (slope of the index N versus 1/B, see Figure 27a). In the plot of Figure 27a (top inset) the zero intercept with the N axis is a signature of graphene (no ½ off-set) even for MEG. The temperature dependence (Lifshitz-Kosevich equation) gives access to the separation between levels that is also characteristic of graphene in multilayers C-face. The Landau levels smoothly merge into equidistant levels at low field, when the cyclotron orbit size is larger than the ribbon width and confinement effects dominate the energy quantization (particle in a box effect) [49].

At high field or very low charge density, similarly to other 2D electron gases, the resistance oscillations develop into the quantum Hall effect (QHE). The resistance becomes vanishingly small at the oscillation minima, concomitant with plateaus in the transverse conductance that are quantized in units of e^2/h : $\sigma_{xy} = ve^2/h$ ($e^2/h = 1/25.812.807 \Omega^-$ ¹:). The zero resistance comes from the physical separation of forward and backward moving charges at each edge of the sample. Simply put, since the opposite travel direction cannot be reached, the electrons cannot be scattered and the resistance vanishes. The particularity of graphene is in the unique sequence of plateaus at filling factors $v=4(N+\frac{1}{2})=2, 6, 10$ for N=0, 1, 2, ... instead of v=2N, N=1, 2, 3, ... (without spin splitting) in normal 2DEGs. The factor 4 in graphene reflects spin and valley degeneracy. The second particularity of graphene is the large separation between Landau levels, for instance $\Delta E = E_{N=1} - E_0 = 36\sqrt{B} (meV) \sim 400$ K at 1 Tesla, to be compared to ΔE =1.7B(meV)~20 K in GaAs/AlGaAs 2DEGs heterostructures. This allows to observe this quantum effect at much lower magnetic field [76, 137] (see Figure 27b) and much higher temperature than the sub-kelvin temperature range of QHE in 2DEGs. In graphene the QHE was observed even at room temperature [235], see also inset of Figure 27b for single layer epigraphene C-face. This particularly sought for in metrology applications for resistance standards based on the quantized e2/h plateaus working at a few Tesla (see Figure 27d) and potentially at liquid nitrogen temperature, instead of the current sub-Kelvin range. Large area high mobility graphene are however required, which epigraphene can provide [76, 236, 237].

The first observation of the QHE in epigraphene was on high mobility graphene on the C-face [123, 137]. Environmental positive counter doping [123] or top gate controlled [137] was used to observe the QHE in low magnetic field (v=2 plateau below 3 Tesla, see Figure 27b). Progress was rapidly made also on the Si-face with proper charge density compensation (see Section 5.1.1 [236] [237] with a measured quantum Hall resistance quantization accuracy of $3x10^{-9}$ at 300 mK [76, 236], rivaling with the best 2DEG standards. The relative discrepancy between the quantized Hall resistances in the graphene sample and in a reference GaAs was found as low as $(-2\pm4)x10^{-10}$, which demonstrates that epigraphene can substitute GaAs QHE resistance standard [76, 238]. This result demonstrates the structural integrity and uniformity of epigraphene over hundreds of micrometers. A particular type of disorder is however necessary to observe the QHE to insure that states in the bulk of graphene are localized, so that the states at both edges are well decoupled. It turns out that the QHE for epigraphene Si-face shows bi-layer inclusions [239], by coupling (or channeling) of states from both edges.

5.3. Towards electronic devices

5.3.1. High frequency transistors

Digital electronics requires that the gate-modulated current through the transistor channel be switched between on and off values, with a large I_{on}/I_{off} ratio and small I_{off} for minimum loss. Because graphene doesn't have a band gap, the conductance is non zero at all gate voltage and the channel current cannot be turned off velocity (for a review, see for instance [240]). This means that I_{on}/I_{off} is limited to values of a few tens (Figure 23), to be compared to at least 104 in CMOS technology, and that poor current saturation is observed at high bias voltage [241]. Nevertheless, graphene can be used for amplification at high frequency, owing to its intrinsic low dimensionality, high carrier mobility and large carrier. Epigraphene field-effect transistors have shown fast progress. For radio frequency transistors, the two most important small-signal figures of merit are the cutoff frequency f_T (the maximum frequency for current amplification, i.e. at which the current gain is unity) and the maximum oscillation frequency f_{max} (that is the maximum frequency for power gain).

The intrinsic properties of epigraphene are quite promising in view of the transconductance $g_m=dI_{sd}/dV_g$ and high carrier velocities[242]. The transconductance describes the efficiency of the gate voltage V_g to modulate the channel current I_{sd} and directly relates to the intrinsic cut off frequency $f_T = g_m/(2\pi C_G)$, excluding parasite resistance or capacitance, with C_G the gate capacitance. Large transconductance up to 2 mS/µm and output current above 5 mA/µm have been reported in epigraphene Si-face [243]. Higher carrier velocities can be achieved with high saturation velocity and high mobility. As a substrate SiC compares favorably with SiO₂ because optical phonon, argued to be the main source of hot carrier scattering, have a higher energy [241]. Following this line of argument, THz operation is predicted on SiC substrates.

The highest cutoff frequency (after removal of parasitic effects) in epigraphene made by conventional scalable process is f_T =280 GHz on the Si-face (L_g =40nm) [243], to be compared with record f_T = 427GHz (L_g =67 nm) for a transferred gate on peeled graphene [244], and with world record 0.5-THz f_T in SiGe heterojunction bipolar transistors [245] (Note that the cutoff frequency depends on the gated length L_g). However, use in any practical circuit requires high values of the power gain f_{max} , which still remains much lower than fT in most graphene devices (see Figure 28c). High fmax values (fmax =70 GHz for Lg=100nm) were only very recently demonstrated (see Figure 28b). This was realized with high mobility epigraphene C-face (room temperature FET mobility μ = 6,000 cm2/Vs; fT=110GHz) and by optimization of the device dual gate design (shown in Figure 28a), with low contact and gate resistances. [221]

Note that multilayer graphene is not as efficiently gated because of electronic interlayer screening that results in a constant conductance in parallel (as shown in Figure 23c). On the Si-face, better performances are obtained when the entire channel lies within a single graphene terrace, avoiding step edge scattering (Sec 5.1.4) [246]. Another challenge for any graphene- based transistor is to develop gate dielectrics that preserve high mobility and charge homogeneity in graphene. With no dangling bonds to adhere to, dielectric coating on chemically passive graphene requires a nucleation layer or a roughened top graphene layer [247]. Atomic layer deposition (ALD) or evaporation of high K dielectrics (Al₂O₃, HfO₂, SiO₂)[248] or Si₃N₄ [243] have best RF performance. Progress is also being made with less perturbing coating, like boron nitride (a substrate with no dangling bonds where graphene retains a high mobility), or polymers [214]. The latter however, may induce unwanted hysteretic behavior. Good results (in terms of low leakage, high-capacitance gate-dielectrics and no damage to graphene) have been reported for oxides grown by ALD on graphene seeded by self-assembled molecular layers of perylene-3,4,9,10-tetracarboxylic-3,4,9,10-dianhydride (PTCDA). [213] The PTCDA molecules show long-range order that is not perturbed by defects in the epigraphene or SiC atomic steps. [249] A chemical route using solution-based self-assembling nanodielectrics have been processed on epigraphene Si-face and seem promising.[250]

Contact resistance to graphene is an interesting fundamental problem. The question of how to inject from a 3D to a 2D material while matching energy and momentum is difficult considering that at charge neutrality the graphene Fermi surface is reduced to points in k-space. Similarly to dielectric coatings, metal-to-graphene non-wettability issues are solved by using an adhesion layer (Ti, Cr). The lowest reported contact resistance on epigraphene is less than 100 $\Omega\mu$ m [221] for gold-plated high work function metal Pd or Pt, similarly to carbon nanotubes.

Frequency multiplication and mixing have also been developed for epigraphene. This is based on the nonlinearity of graphene FETs near the Dirac point : the $I(V_g)$ has a voltage square ambipolar behavior at zero bias, unlike in other semiconductor transistors, that greatly suppresses odd-order harmonics in epigraphene-FET devices [251]. Epigraphene -based frequency mixers show excellent linearity [251]. Moreover the observed low phase noise and low 1/f noise in epigraphene-FETs [252] is an important consideration for nonlinear circuits. Ultra-wideband detection (2–110 GHz) was also demonstrated in epigraphene-FETs. Their performance is comparable to or better than state-of-the-art FET-based direct millimeter-wave detection without dc biases applied. [253]

5.3.2. Spintronics

Spintronics devices use electronic spin instead of charge to process information. Fundamental to this vision is the possibility of efficient spin propagation over long distances, but despite several decades of intense research, spin transport efficiencies have remained low in metals and semiconductors. Carbon–based materials however are good candidates owing to small spin-orbit coupling and results on high-mobility multilayer C-face epigraphene (μ =17,000 cm²/Vs) indeed presents a breakthrough in the field [254]. Spin transport efficiencies reach 75%, spin signals are in the M Ω range and spin diffusion lengths exceed 300 µm (spin lifetimes >100 ns), which is significantly larger than any other material. For these measurements, the devices were provided with two-terminal high-impedance tunnel contacts (evaporated cobalt on alumina) acting as spin polarizer and analyzer (see Figure 29b). The spin signal is measured by the difference ΔR in the resistance when the two ferromagnetic electrodes have parallel or antiparallel magnetization. The M Ω range signal, as shown in Figure 29c, is to be compared to the few tens of Ohms generally reported in other materials [255]. Similarly, much larger relative spin signals are also measured in C-face epigraphene with $\Delta R/R\approx 10\%$. [254] Micrometer long spin diffusion lengths on C-face epigraphene were also inferred from the resonance width in spin resonance experiments. In that case, the transition between Zeeman-split levels probed by microwave radiation in magnetic field was detected electrically. [256]

On the Si-face, sub-µm spin relaxation length and 1-2 ns spin relaxation time were determined from measurements in a non-local configuration and in Hanle spin precession measurements. In the former experimental configuration, a spin-polarized current is sent between two contacts. The spin accumulation generated at the injection contact diffuses and is detected at two remote spin sensitive voltage contacts. [257] Hanle precession experiments yield an order of magnitude lower diffusion coefficient but higher spin relaxation time on monolayer epigraphene compared to quasi-free standing epigraphene (hydrogenated buffer). This was tentatively explained by localized states in the buffer layer.[258]

5.3.3. Large-scale integration – integration with Si

Epigraphene is the only graphene platform where large-scale transistor fabrication was demonstrated. Hundreds of devices [212] were produced on mm size SiC chips as early as 2008 (an example is given in Figure 23c), and more than 10,000 two years later [135] (see Figure 30a). Wafer size patterning of high frequency transistors was demonstrated on a 100 mm wafer (see Figure 30b) in 2011 [68]. An integrated circuit was fabricated on a single SiC chip with a graphene FET that operates as a broadband radio-frequency mixer at frequencies up to 10 gigahertz with thermal stability up to 400 K [259].

As SiC is a large band gap semiconductor widely used in the high power electronics industry, it is interesting to develop schemes to integrate the electronic properties of epigraphene and SiC. One such scheme is to utilize the semiconducting SiC as the transistor channel and graphene as electrodes [215]. A variation of the technique consists in using a quasi-freestanding hydrogenated epigraphene bilayer as the gate, so that the entire device can be carved in one step from epigraphene on SiC (the monolithic wafer-scale electronics scheme is presented in Figure 31). It is expected that this will allow for high currents, high operation speed and high operation temperatures. Current on-off ratios exceeding 10⁴ were indeed observed. [260]

In another promising design [261], the channel between co-planar multilayer graphene source and drain consists of the accumulation layer at the interface of semi-insulating SiC and a surface silicate that forms after high temperature annealing (**Figure 31**). The current flow over the one-dimensional C-face graphene/SiC barrier exhibits on/off ratio over 10⁶, with current densities up to 35 A/m. Significantly, transport is dominated by tunneling at low temperatures, while it is dominated by thermal activation above the Schottky barrier at high temperatures. Tunneling FETs are in fact much sought after to overcome the transconductance thermal limitation.

Another approach is to integrate epigraphene to Si-based electronics. The first Graphene-On-Silicon (GOS) strategy takes advantage of the heteroepitaxy growth of 3C-SiC on Si to produce epigraphene on SiC covered Si substrate (3C-SiC is the only SiC polytype known to grow on Si) [175, 177, 178]. Epigraphene growth on SiC/Si subtrates was discussed in Section 4.4. FETs were produced on GOS with SiC layers on both Si(111) and Si(110) substrates. Besides the usual epigraphene top gates [262] and back gates have also been realized by using of the thin (less than 100nm) heteroepitaxial SiC film on Si as the dielectric [263]. Modulation of the current in the graphene channel is observed. Despite non-negligible leakage currents to the Si substrate through the granular SiC, a room temperature logic inverter was demonstrated.[264]

In another strategy [265], thin monocrystalline silicon layers are transferred onto previously graphitized SiC substrates using well-established industrial Silicon–On-Insulator (SOI) wafer bonding and smart-cut techniques. The transferred crystalline silicon layer bonded on top of SiC/epigraphene is ready for silicon-CMOS devices implementation. It is connected to the sealed epigraphene layer beneath it by metallic vias managed in the sub-micron thin Si layer. The key advantage of the technique is that epigraphene is grown and structured prior to bonding. The process is therefore compatible with epigraphene high temperature growth and preserves epigraphene high structural quality integrity with no degradation. The process produces monolithic integration of graphene on SiC/silicon 3D stacked layers and is fully compatible with Very-Large Scale Integration Technology (See Figure 32a-b).

5.4. Band gap

The two main schemes to create a gap in gapless graphene include narrowing the channel width W to open a confinement gap and locally functionalize graphene (sp^3 –like covalent bonding). The order of magnitude of the confinement gap Eg=1eV nm/W (see Section 2.9) means patterning at the tens nm scale or below for any practical application. This is usually done by lithography and oxygen plasma etching of 2D graphene, which tends to create disordered edges that severely reduce mobility (see Section 5.5).

Band gaps are reported in all-graphene structures. Epigraphene grown on minifacets a few nm wide that border the sidewall ribbons, as described in Section 4.5, [111] have been associated with the ~ 0.5 eV wide band gap observed by ARPES [196](see Figure 21). Buckled epigraphene also presents a band gap larger than 0.7 eV from ARPES measurements[266]. Buckling arises in this case from a submonolayer concentration of nitrogen seeded on SiC that pins epigraphene to the SiC interface as graphene grows by thermal decomposition. The buffer layer is also a well-known all-graphene semiconductor, with a band gap of at least 1 eV according to early band structure measurements [35] [78]. Recent ARPES measurements put the distance between the Fermi level in the gap and the top of the conduction band closer to 0.5 eV [267] (see Section 4.1.1). For semiconducting graphene, seamless connection to graphene provides optimum carrier injection, that can be provided in this case by the ballistic ribbon grown on the sidewall facets, as described in Sections 4.5 and 5.5. Transport measurements confirm the semiconducting nature of the buffer layer. [256]

Other seamless planar junctions have been proposed between semiconducting functionalized graphene and graphene structures. These include for instance locally converting an epigraphene channel through oxidation between graphene contacts [268], fluorination [269] or covalent chemistry with nitrophenyl groups [270], where a band gap was demonstrated [271]. The formation of covalent carbon-oxygen/fluorine or molecule bonds locally changes the electronic structure and the transport properties of the epigraphene from metallic to semiconducting.[270] Large on-off ratios up to 10⁵ and current saturation at high bias voltage are measured [269].

Remarkably ferromagnetic behavior was observed at room temperature by functionalization epigraphene with nitrophenyl groups. The interpretation is that magnetic products may be produced when two groups are grafted on the A sublattice, leaving 2 unpaired spins in the B sublattice. [272] Although quite promising, progress may be hindered by inhomogeneous graphene functionalization that consequently reduces the mobility.

Graphite oxide, which is graphene functionalized with epoxy and hydroxyl groups, was fabricated by direct oxidation of epigraphene, either by the Hummers' method [268] or through milder oxidation [273]. Nanopatterning under a biased conducting AFM tip also leads to local graphene oxidation. [274] Interestingly epigraphene (both Si and C-face) resists the very harsh chemical treatment and doesn't exfoliate so that multilayer graphene oxide is produced directly on chips. In that case the layers are very well ordered, exhibit excellent inter-layer registry and little amount (<10%) of intercalated water. [275] This good epigraphene oxide quality makes it possible to reduce locally graphene oxide into graphene with a heated AFM tip. Conducting nanoribbons are drawn down to 12nm wide in an insulating graphene oxide matrix in a single step that is clean, rapid, and reliable [276].

5.5. Sidewall Ribbons

As presented in Section 2.9 early theoretical work on graphene ribbons [62, 190, 277] [278] predicted different electronic properties based on their edge orientation: zig-zag ribbons have quasi-non dispersive states close to charge neutrality while two out of three armchair ribbons are gapped, see Figure 5. Following these papers and early proposal for graphene nanoelectronics [1, 2], ribbons a few tens to hundreds of nm wide have been patterned by oxygen etching 2D graphene. It was found that transport in these ribbons is quenched at charge neutrality and at low temperature, resulting in gate-induced current on/off ratio up to 10⁶ with a large temperature dependence [182, 184, 186, 187]. All were attributed to electron localization effects due to edge disorder. Following Eq.2

 $E_{n,m} = hv_F \sqrt{(n/L)^2 + (m/W)^2}$, described in Section 2.9, it is clear that only the m=0 modes are occupied in narrow ribbons at low temperature. It is therefore expected that these 1D edge states wouldn't be conducting if too much disorder is present.

On the contrary, as shown in Figure 33, 40nm charge-neutral epigraphene sidewall ribbons (Section 4.5 are found to be metallic, with conductance of the order of h/e^2 . This demonstrates the presence of conducting states in the confinement gap (as discussed in Section 2.9, Figure 5) and indicates single-channel ballistic transport. [136]

Direct proof of ballistic transport is provided by the measurements of a sidewall zigzag ribbon presented in Figure 33a in a four point configuration, where the current is passed through the outer tips and voltage is measured at the two inner probes (see inset of Figure 33a). The distance d between the nanoscopically sharp tungsten voltage probes in ohmic contact with a single ribbon can be controlled in situ in the built-in electron microscope. It is remarkable that (i) The resistance R_{4p} extrapolates to $R_{4p}(d=0)=h/e^2$ at zero length indicating single channel transport (ii) The resistance per unit length $R'=\Delta R_{4p}/\Delta d$ decreases after in situ heating (a process known to clean graphene), and becomes length independent (bottom trace in Figure 33a), indicative of extremely long mean free path (the transmission of a 1D single conducting channel can be related to a mean free path λ according to $R=(h/e^2)$ (1+d/ λ), as indicated in Figure 33a. (iii) The probes are invasive and interrupt the current flow. In a diffusive conductor a probe merely affects the resistance that is dominated by scattering on multiple impurities and defects. Here it is shown that introducing a passive tip in between two voltage probes doubles the resistance; two passive probes triple it. Similarly four point and two point (where the same tips are used as current leads and voltage probes) measurements give essentially the same result. This is because right-moving charges that enter the passive probe from the left, say, will leave the probe going either left or right with equal probability; hence the transmission probability is half. [136]

It is not understood why only one conducting channel is measured. For short ribbons at room temperature a second channel contributes and vanishes exponentially for length longer than 160nm; similarly the single channel conductance of long ribbons drops to zero as the distance between probes increases beyond 16µm (see Figure 33b). At low temperature, for a fixed contact geometry applied to a curved ribbon (see Figure 33c - inset), the conductance increases with increasing the gate voltage. This can be understood by the Fermi level moving to the upper bands so that many more channels contribute to the conduction. The conduction in the upper bands is found to be small, with mean free path of the order of 50 nm, which is comparable to the ribbon width. From the temperature dependence of Figure 33c it can be seen that each of the curves can be displaced vertically to overlap the others (Figure 33d), indicating that only the resistance at Vg=0 is temperature dependent. This was explained in terms of a two-mode transport model.[136] These exceptional results underscore the importance of well-prepared ribbons, with well-defined and annealed edges.

6. Optical and Plasmonic Properties

Recent years have seen a surge in publications dealing with photonic, optoelectronic and plasmonic properties of graphene and graphene-based devices. Concepts and figures of merit of prototype devices have been described in recent reviews [279-281]. Although the challenge remains to scale up from single device to large scale production and integration with existing photonic and electronic platforms [280], potential applications cover a large range from passive components such as transparent electrodes in displays or touch screens, to active elements as proposed in photovoltaic module, photodetectors, optical modulators, fast saturable absorbers for high power lasers, infra-red filters, frequency convertors, for THz emission or detection, plasmonic devices, etc. These exploit various properties of graphene, notably high mobility transport, optical transparency, wave-length independent absorption, and non-linear optical response properties. In fact graphene interacts strongly with light (graphite is black) with a remarkably high absorption coefficient in the near-infrared and visible range: $\gamma = \pi \alpha = 2.3\%$ per layer (α = $(1/4\pi\epsilon_0) e^2/\hbar c \approx 1/137$ is the fine structure constant). This is one to three orders of magnitude larger than that of technologically relevant photonic materials such as $In_{0.53}Ga_{0.47}As$, GaAs, or germanium at a wave length of 1.55 μ m [281], the wavelength used in most fiber optic telecommunication systems. However since graphene is just one layer, it is usually considered as optically transparent. One optical property that is specific for epigraphene on the Cface, is that the optical conductivity is expected to depend on the layer stacking and the relative orientation of bilayers.[282]

Quite uniquely, graphene's gapless band structure allows charge carrier generation by absorption of light in a broad wavelength range. In neutral graphene an electron-hole pair can be excited by any radiation from THz to ultraviolet, since there is no minimum energy difference between conduction and valence band, as shown Figure 34b (see for

instance Ref. [280] and refs therein). Since a photon can only be absorbed when its energy matches the energy difference between an occupied and an empty state, for energies below 2EF in graphene (see Figure 34b) the absorption is therefore blocked (so-called Pauli blocking) and the graphene layer is transparent. This provides in principle a controllable way to tune transparency with doping. Carrier multiplication, promoted by strong electron-electron interaction and weak electron-phonon coupling, and high carrier mobility are favorable for ultrafast conversion of light to electrical signals [280]. Integration of graphene with metal contacts, gates, in cavities, wave guides, imbedded in polymers, in composite stacks, coated with molecules or colloidal quantum dots have been devised to increase optical absorption, and photoresponse [280]. As for epigraphene, studies of the optical properties and realization of photonic devices are however quite limited and were mostly directed towards ultrafast spectroscopy and plasmonic effects that we briefly discuss below.

6.1. Ultrafast optical spectroscopy

Ultrafast time-resolved optical pump-probe spectroscopy has been used to study the dynamic response and the hotcarrier relaxation and cooling dynamics in epigraphene with femtosecond resolution. In this technique, an ultrafast optical pump pulse excites electrons and holes to high energies, while a probe pulse, whose frequency can be tuned from the visible [283] through the infrared [203] to the THz frequencies [284, 285], probes the dynamics response (see Figure 34). The electron dynamic response is an important property for all photodetectors and photovoltaic devices that rely on the conversion of light into free electron-hole pairs. Best performances can be are achieved if the photoexcited carriers transfer their excess energy into the production of additional electron-hole pairs (carrier multiplication) through carrier-carrier scattering processes rather than production of heat (emission of phonons). Similarly the ultimate performance of high-speed electronic devices will depend on the energy exchange between large electric fields induced hot carriers and the lattice.

Epigraphene is particularly well suited for ultrafast optical experiments due to its large area and homogeneity. Up to cm-scale surface areas provide sample sizes that are large enough even for the typically large probe spot sizes at frequencies below 300 GHz. The signal is enhanced by the multiple layers in C-face epigraphene without compromising the graphene characteristics since each layer behaves electronically like a monolayer graphene. Finally, transmission experiments benefit from the transparency of the SiC substrate over broad frequency window from the visible to the THz range.

In ultrafast time-resolved THz spectroscopy experiments an ultrafast optical pump excitation generates electronhole pairs in a non-equilibrium state, while a broadband single-cycle THz probe pulse probes the dynamic THz response of the hot-carrier relaxation and cooling dynamics. The high-energy non-equilibrium electrons and holes thermalize with the background of cold carriers due to very efficient carrier-carrier scattering which forms a single hot-carrier Fermi-Dirac distribution within ~100-200 fs, as observed in epigraphene [203, 283-286]. The hotcarrier distribution is characterized by an elevated transient carrier temperature and a transient Fermi level. As the hot carriers relax and cool, both the transient carrier temperature and Fermi level shift return to equilibrium. The THz carrier dynamics are the result of an interplay between efficient carrier-carrier scattering, which maintains a thermalized hot-carrier distribution, and carrier-optical-phonon scattering, which removes the energy from the hot carriers to the lattice [287, 288]. This dynamics is shown to depend critically on the doping density, ranging from a few picoseconds at doping density n >10¹³cm⁻² to hundreds of picoseconds at doping densities n <10¹⁰cm⁻², due to the vanishing density of states at the Dirac point. In MEG, the low-energy dynamics is governed by a unique cooling pathway enabled by interlayer energy transfer via screened Coulomb interactions between Dirac electrons, with the highly doped layers near the SiC substrate acting as a heat sink for the quasi-neutral top layers [284].

In a special semiconducting form of epitaxial graphene, in which an energy gap of up to 0.7 eV was engineered by buckling [266], the relaxation of the dynamics is enhanced by up to two orders of magnitude, which is attributed to direct electron-hole recombination via optical phonon emission over a broad range of carrier temperatures[285].

These experiments, consistent with results on other forms of graphene [279-281] [289, 290], show that carriercarrier scattering is highly efficient in a wide range of photon wavelengths and produces secondary hot electrons that can drive currents. This multiple hot-carrier generation is therefore interesting for highly efficient broadband conversion of light into electronic energy for optoelectronic applications. Applying an external magnetic field leads to a significantly longer relaxation [291], which is attributed to a reduction of electron-electron scattering. This is because in a magnetic field the energy levels bunch into unequally spaced Landau levels (see Section 5.2.2), therefore inter-level carrier–carrier scattering processes (Auger processes) are not allowed because they cannot conserve energy. This result is in sharp contrast with normal 2D electron gases, where the formation of equidistant Landau levels in magnetic field increases scattering . Graphene's zeroth Landau level is special because the three LL₀, LL₁ and LL₁ Landau levels are equally spaced so that Auger processes are allowed [292]278]. Neutral C-face layers provide an ideal system to study carrier-carrier scattering processes involving the LL₀ level, and strong Auger processes have been observed by addressing the levels individually with circularly polarized light [292].

Ultrafast optical spectroscopy also gives insight into the electronic structure. Consistent with ARPES measurements [9], the upper limit for a potential band gap is found to be below 1 meV in C-face graphene [203]. The doping density in the successive layers in MEG was also evaluated, corresponding to the energies of Pauli blocking ($E_F \sim 350$ meV, 210meV, 135meV, 90meV) [203], see Figure 34b. Also the polarization of the THz emitted light by coherently controlled photocurrents (see below) indicates that there is some coupling between the graphene layers in MEG although the electronic structure of each layer is that of graphene.[293]

Coherently controlled photocurrents have been produced in MEG. [294] The method uses two phase-locked beams at frequency ω and 2ω . Quantum interference between single-photon and two-photon absorption breaks the lattice symmetry, so that the photoinjected carriers have an anisotropic distribution in k resulting in a net current flow. The current flow direction is controlled by the polarization of the pump beam. The transient current is detected by the electromagnetic pulse at the THz frequencies that it generates. Note that this all-optical injection of current could provide a noncontact way of injecting directional current in graphene. Quantitative studies of the magnitude of the effect show that current decay time in graphene are longer compared to common semiconductor like GaAs.

6.2. THz generation

Graphene exhibits a large nonlinear optical response due to its linear energy dispersion that leads to harmonic generation at THz frequencies. However, second-order nonlinear effects important for applications like frequency difference or rectification processes, are forbidden by the centro-symmetric nature of graphene. Generation of coherent THz radiation (0.1 to 4 THz and projected up to 60 THz) via a second-order nonlinear effect was nonetheless achieved in multilayer epigraphene on the C-face where it is induced by the anisotropy of the in-plane photon momentum [295]. This dynamical photon drag effect relies on the transfer of photon momentum to the carriers by the ponderomotive electric and magnetic forces. Interestingly, the effect is related to the usually neglected next-nearest-neighbor coupling of C atoms in graphene and to the asymmetry between the electrons and holes dynamics. These properties are particularly interesting for the generation of ultra-broadband terahertz pulses in compact room temperature THz sources.

6.3. Photo-current

The electrical response of epigraphene under illumination was studied for both Si and C-faces. Similarly to the classical Hall effect, the crossed AC electric and magnetic fields of circularly polarized light can create a voltage, what is called the circular AC Hall effect. Graphene centrosymmetric structure prevents other helicity-driven photocurrent effects, and the circular AC Hall effect was observed for the first time in epigraphene Si-face, owing to the large area available for the terahertz (THz) laser radiation. [296]

In one report [297], a significant increased photocurrent on the Si-face compared to the C-face (and CVD graphene) is attributed to the presence of the buffer layer, although the mechanism is unclear. In an attempt to increase photocurrents, asymmetric contact configurations were applied on top of C-face and Si-face epigraphene (Au/epigraphene/Al [298] and Ti/epigraphene/Pd [299]). The enhanced photoresponse is attributed to the difference in built-in electric fields at the two epigraphene /metal interfaces. Thicker layers show stronger response [298, 299]. In another implementation, an epigraphene strip was locally modified by laser illumination (LEG) creating epigraphene/LEG/epigraphene Schottky junctions. In the LEG strip, epigraphene is converted into a poor conductor (possibly graphene oxide) and the device show nanosecond photoresponse, promising for

photodetectors [300]. In the same line enhanced photosensitivity of epigraphene was also reported after electrochemical oxidation in nitric acid [301]. This is attributed to the formation of deep traps at the electro-oxidized epigraphene interface, which release charge carriers under illumination and prolong the lifetime of the photocarriers. The SiC substrate can also induce photocurrent resonance. This was observed in the spectral region where SiC has a negative dielectric constant.

As SiC is a wide-band gap semiconductor in its own right, epigraphene was also used as the (transparent) contact to probe the photo-generated charge carriers in the semi-insulating 4H-SiC substrate [302], with best response in the ultra-violet.

6.4. Plasmonics

An important motivation for plasmonics is to merge the fields of photonics and electronics using nanoscale confined electromagnetic fields. Graphene presents the advantage that its plasmonic properties (frequency, propagation direction) are highly tunable by changing its carrier density [303, 304], structural characteristics, such as the graphene nanostructure dimension, packing density, or number of stacked layers or by applying a magnetic field. Also owing to the low ohmic loss of the material, long lifetimes and high degree of optical field confinement are observed.

Like for other metal surface, collective charge density oscillations of the electron gas (plasmons) can propagate at the surface of graphene. These electromagnetic waves have their own, strictly two-dimensional, band-dispersion relation that is different in graphene than in ordinary metals [86, 305]. Plasmons can couple to light (creating surface plasmon-polaritons, that can convert light into electronic signals), to elementary charge (to create plasmarons), to interband electron–hole (e-h) pair excitation (to create plexciton) and to phonons. All the plasmon coupling mechanisms mentioned above have been observed in epigraphene.

The first observation of plasmons in graphene was realized in epigraphene Si-face by high-resolution electronenergy-loss spectroscopy (EELS). The predicted dispersion relation $\omega_p \propto \sqrt{q}$ at small wave vector **q** was confirmed as well as a kink in the dispersion in the vicinity of the Fermi wavevector [204, 305]. This dip in the plasmon dispersion was shown to be an intrinsic property of pristine epigraphene and does not depend on defect concentration, on the number of graphene layers or temperature. It is attributed to the decay of plasmons due to resonant coupling between plasmons and e-h pair excitations. [306] At higher **q** an unusual damping of the plasmon mode is observed, because of the combined effect of phase space limited backscattering for electrons in graphene and enhanced electron-electron elastic scattering.[305] In an unexpected application, epigraphene was used to demonstrate that coating materials with graphene can enhance radiative heat transfer between materials in the near field [293. This exploits the tunability of graphene plasmons that can match the frequency of phononpolariton or plasmon-polariton resonances in a given material. Best heat transfer is indeed expected when resonances in the opposing bodies match each other. This effect may be interesting for energy conversion devices.[307]

Plasmons in highly charged epigraphene on the Si-face strongly interact with the surface phonons of the SiC polar substrate [204]. In this system, the plasmon and phonon energy bands show avoided crossings. Despite electronic screening from one layer to the next, the plasmon-phonon coupling extends to multilayers.

Bound states of plasmons and electrons/holes (plasmarons) can play a strong role in renormalizing the bands around the Dirac energy [83, 84]. In particular for potassium doped quasi-freestanding epigraphene (hydrogenlifted buffer layer), ARPES measurements show that the energy band-dispersion close to the Dirac point is reconstructed in a nontrivial way with four distinct bands instead of the usual pair [83] (see Figure 10b). This was interpreted as evidence for many-body effects (electron–electron, electron–plasmon and electron–phonon couplings) in the dynamics of quasiparticles.

Surface plasmons manifest themselves as resonance absorptions in the optical transmission spectra. Excitation of plasmons was shown to dramatically modify the magneto-optical response [308], in particular the giant Faraday rotation observed for the first time in epigraphene [309].

Despite the inefficiency of plasmon excitation and detection by light due to the large wavevector mismatch of light with graphene plasmons, light-matter interactions was demonstrated in graphene nanostructures via confined plasmons. In this case, electromagnetic radiation can excite plasmon resonances in graphene, even in nanostructures having dimensions much smaller than the incident light wavelength. These structures can be either intentionally patterned or arise naturally, such as the step-terrace microstructure of epigraphene on the Si face or the tapered monolayers structures grown on the C-face [290]. In magnetic fields, plasmons can couple with the cyclotron resonance (inter Landau level transition) to form a hybrid mode. Infrared magneto-spectroscopy measurements have taken advantage of large arrays of sub 100 nm patterned quasineutral ribbons (E_F <17meV) of high mobility (μ >50,000cm²/Vs) on epigraphene C-face to study this hybrid mode in detail. The observed energy shift exhibits a peculiar energy – magnetic field scaling that distinguishes it from conventional 2DEGs and in highly doped graphene. [310]

Complementary to the spectroscopic results above, coupling of plasmons and light have enabled plasmon fields imaging in real space on epigraphene monolayers [304] (see Figure 35) or multilayer [311] C-face nanoribbons [304]. Strong optical field confinement (plasmon to incident light wavelength reduction by a factor more than a hundred)[310] and long propagation distances are observed [304], and the time evolution of the electric field distribution over the surface [312] could be detected in the THz regime.

Important to the field of plasmonics is the possibility of plasmon tuning, also demonstrated in epigraphene. The plasmon wavelength can be varied in epigraphene over a wide spectral range by slightly changing the incident light wavelength [304], owing to the strong dependence the dielectric constant of SiC on the wavelength and the sensitivity of plasmon to the dielectric environment. The dependence of the plasmon dispersion on the carrier density (proportional to $n^{1/4}$) was measured on a top gated epigraphene Si-face, [313] where plasmon transport was studied by time-resolved electrical measurements of a charge pulse travelling in a plasmon mode. Change in the plasmon velocity was also realized by varying the magnetic field, charge density and top gate screening effect in epigraphene on the Si-face. [313]

7. Conclusion

In this review we have presented the status quo of the science and technology of epigraphene. Historically epigraphene is the first graphene platform to be scientifically and technologically developed and it is currently the frontrunner for graphene electronics. Epigraphene surface science research predated transferred graphene research by several decades, the concept of graphene electronics was first proposed and patented based on epigraphene electronics research, and electronic transport properties of monolayer graphene were first demonstrated on epigraphene. Because of its remarkable high quality, it is the only material to reveal the characteristic graphene band structure in ARPES, and epigraphene nanostructures reveal one dimensional unique electronic transport properties that have not been seen elsewhere. In contrast to all other graphene flavors, epigraphene survived on the route to become a viable electronics platform, because only it satisfies the many necessary requirements. After more than a decade of research into epigraphene nanoelectronics, the field is still strong, in contrast to nanoelectronics that is based on transferred graphene that has rapidly declined due to difficulties in fabricating the necessary low-defect nanostructures. We can therefore comfortably predict that it is only a matter of time before graphene electronics is realized. However considerably more work needs to be done before this dream is realized.

Acknowledgements

Zhigang Jiang, Ted Norris and Momchil Mihnev are thanked for a critical reading of the manuscript. C.B. and W.A.d.H. thank the AFOSR and NSF under grants No FA9550-13-1-0217 and 1506006, respectively. Additional support is provided by the Partner University Fund from the French Embassy. CB acknowledges partial funding from the EU Graphene Flagship program. E.H.C acknowledges support from NSF under grants No. DMR-1401193 and No. DMR-1005880.

8. Figure Captions

Figure 1: (a) Graphene is a one atom thick 2 dimensional honeycomb structure of carbon atoms. (b) Two Bernal stacked graphene layers are the constitutive building block of hexagonal graphite (c) Three Bernal stacked graphene layers. (d) ABC stacking of graphene layers makes the unit cell of rhombohedral graphite. A and B atoms correspond to the two sublattices forming the honey comb structure. The tight-binding γ_i parameters, corresponding to first, second, etc neighbors, for electronic structure calculations are indicated. (From Ref. [314]).

Figure 2:(a) Band structure E(k) of graphene, and zoom in close to the K point, showing the two inverted cones. (b)– (e) The π -bands near E_F for 1– 4 graphene layers, respectively, measured for epigraphene (Si-face) by Angle Resolved Photo-Emission Spectroscopy. The dashed lines are from a calculated tight-binding band structure, with band parameters adjusted to reproduce the measured bands. Red and orange lines are for ABAB and ABAC stacking, while blue lines are for rhombohedral ABC stacking. (From Ref.[10]).

Figure 3: Comparison of epigraphene with different exfoliated graphene materials. (a) AFM images of exfoliated graphene [315]. (b) An STM image of epigraphene grown on SiC-(0001).(courtesy Philip First, Georgia Tech and Joseph Stroscio, NIST-CNST). Note the ten- fold increase in flatness of the epigraphene film. (c) A comparison of the ARPES measured Dirac cone for 2D epigraphene grown on SiC-(000-1)[9] (left), the band structure of a 2D exfoliated film (center), [316] and a 24nm wide epigraphene sidewall ribbon.[196] Note that the energy and momentum scales are the same in all images in (c) and (d). Top panels in (c) are measured E(k) and bottom panels are constant energy cuts through E(k) showing the momentum distribution curves (MDC) of the bands. The Δ k broadening in exfoliated graphene gives a coherence length, Lc, = $2\pi/\Delta$ k of 1-3nm. The momentum distribution curve width for epigraphene, including the 24 nm wide ribbons, is entirely due to instrument resolution giving L_c >10 times longer than exfoliated graphene. (d) Energy and momentum distribution curves and MDCs of the highest quality exfoliated graphene.[317]

Figure 4: Epigraphene Si-face grown in UHV (a)-(c). Atomically resolved Scanning Tunneling Microscopy image of epigraphene (sample bias = 0.2eV) (From Ref.[126]). (a) Monolayer on SiC showing the moiré superstructure. Atomic resolution STM image of (b) monolayer and (c) bilayer (4x4 nm² images ; Sample bias: -40 mV). (c) Bilayer epigraphene (from Ref.[118]). (d) Large scale AFM image showing a Swiss cheese morphology (scale bar: 1μm)[66].

Figure 5: Tight –binding band structure calculation for narrow graphene ribbons with (top) armchair and (bottom) zigzag orientation for various ribbon widths as labeled by the number N of row of carbon atoms. Zigzag ribbons have flat band at E=0, and armchair present a gap for specific widths. (e) and (j) N=30.(From [62])

Figure 6: SiC-hexagonal (H) and cubic (C) polytypes. Red: carbon atoms; blue: silicon atoms. The unit cell comprises respectively of 2, 3, 4 and 6 Si-C bilayers. A, B, C refers to the stacking sequence (From Ref. [318]).

Figure 7: Schematics of the confinement controlled sublimation growth method [34]. Contrary to Si sublimation in vacuum (a), in the CCS method (b), the Si vapor is confined in a graphite enclosure provided with a calibrated leak. (c) Photograph of the inductive furnace used for the CCS. (d) Schematics of graphene growth on the two polar faces of 4H and 6H-SiC.

Figure 8: (a-b) Si-face epigraphene grown under 1 atm Ar pressure. (From Ref. [66]) (a) AFM image with a nominal thickness of 1.2 monolayer covering the SiC steps (b) LEEM image revealing macro-terraces covered with graphene. Area covered with 1, 2 and 3 graphene layers have been identified by the presence of 1, 2 or 3 reflectivity minima, respectively. Two and three layers are located at the step edges. (c-e) Hydrogen intercalation after epigraphene growth on the Si-face (c) (left to right) E(k) dispersion measured in ARPES for: pristine buffer layer, after hydrogen intercalation, after annealing at 900C. (d) Same as (c) for a monographene layer. (e) Schematics of the hydrogen buffer lifting from the SiC substrate.

Figure 9: (a) The unit cell structure of 4H-SiC. Filled circles are carbon atoms and open circles are silicon atoms. (b) LEED pattern of monolayer graphene grown on the Si-face. The subscript "G" refers to coordinates in the graphene lattice constant. (c) LEED pattern of 10-layer graphene film grown on the C-face. The diffuse graphene rings are marked.

Figure 10: Renormalization of the band structure due to electron-plasmon interactions (a)-(b) Epigraphene Si-face (n- doped 1.1x10¹³ cm⁻²). ARPES measured E(k) along a line through the K point and along the vertical double arrow in the inset of (b). The kink shape is outlined by the dashed line (b) same band but along the horizontal double arrow (in that direction the other band cannot be observed in ARPES). (c) Schematic picture of the renormalized band. (d-e) Same experiment but for potassium doped (n=1.7x10¹³ cm⁻²) quasifree standing epigraphene Si-face along the (c) vertical and (e) horizontal directions, showing the strongly modified band due to plasmaron (electron-plasmon coupling) (f) Schematic picture of the renormalized band and (g) comparison with non-renormalized graphene.

Figure 11: (a) Atomically resolved STM images of the 6x6 structure at U_{tip} =1.7 V (green dashed line marks the 6x6 cell)(from Ref. [35]). (b) STM at U_{tip} =0.2 V showing the 6 $\sqrt{3}$ cell [from Ref. [35]]. (c) Differential conductance measurements obtained on the buffer and monolayer showing a band gap feature of ~0.8eV on the buffer [from Ref. [90]]. (d) ARPES intensity map of the SiC(0001) (6 $\sqrt{3}x$ 6 $\sqrt{3}$) R30 surface (hv =50 eV). Vertical dashed line marks the K-point of graphene. Two states, g_1 and g_2 above the SiC valence band are also marked. (From Ref. [78])

Figure 12: Epigraphene growth model (Si-face). (a) $10\mu m \times 10\mu m$ atomic force microscopy image of Si-face epigraphene grown by the CCS method, showing step edge growth. (b) Graphene growth kinetic processes on a vicinal surface by step decomposition (c)-(d) Onset of step instability: once the thermal decomposition of SiC has started a positive feedback mechanism promotes further decomposition and graphitization where it has already begun. (From Refs [116, 117])

Figure 13: C-face epigraphene grown by the CCS method. (a) 2.6nmx2.6nm atomic resolution STM image corresponding to the blue square in (b). (b) 400nm x 400nm STM image. Bottom full height scale: 0.1 nm. The RMS height variation is less than 0.02nm of this large area. (c) Atomic Force Microscopy image showing extended flat regions bordered by continuous graphene pleats. Scale bar : $10 \mu m$. (a) A 50 μm field of view LEEM image of a C-face graphene film with an average thickness of 3-layers. Contrast is due to local layer thickness marked in the figure. (e) 400nm long STM image across two regions of different orientations, showing that the top graphene layer is continuous and flat (rms along the profile line in red : 50pm) (STM images: courtesy Philip First, Georgia Tech and Joseph Stroscio, NIST-CNST).

Figure 14: (a) μ -LEED image of a ($\sqrt{57x}\sqrt{57}$)_GR6.59° superlattice from a commensurately rotated two layer C-face graphene film (E= 96 eV). The image is from a 1 μ m area [From Ref. [150]]. The principle diffraction rods from two rotated graphene sheets are marked. The inset shows a blow-up of the superstructure unit cell in graphene units. (b) A calculated LEED pattern from the same $\sqrt{57}$ structure commensurate (log scale). (c) STM topography of moiré patterns on multilayer epigraphene (sample bias V_S=0.5 V, tunnel current I=100 pA). (c) Two similarly sized moiré patterns due to three graphene layers result in a large superlattice unit cell. Layers 1 (surface layer) and layers 2 and 3 have comparable rotation angles but in opposite directions. (d) High resolution image of a region in (c). (e) STM image of a different moiré superlattice. (f) High resolution image from within the same area. [141]

Figure 15: (a) ARPES measured band structure of an 11-layer C-face graphene film grown on the 6H SiC(000-1). The ARPES resolution was set at 7 meV at hv=30eV. The scan in k_y is perpendicular to the SiC <10-10>_{SiC} direction (i.e. graphene rotated 0° relative to SiC). Two linear Dirac cones are visible. (b) k-PEEM constant energy surface (BE=1.3 eV) from a 2ML C-face graphene film using a 7µm field aperture. Two sets of rotated graphene Dirac cones are visible (relative rotation of 21.9°). The reciprocal lattice constants of the two BZ are g_1 and g_2 . A third set of cones, on a different radius, are obtained by coherent diffraction of either the g_2 set of cones by the principal lattice

vector g1 (dotted line), or the g_1 set of cones diffracted by the secondary lattice vector g_2 (dotted line) (from Ref. [319]).

Figure 16 Raman spectroscopy of C-face epigraphene. (laser excitation: 532 nm)(a) A 26nm thick multilayer epigraphene (average thickness determined by ellipsometry), showing the two main graphene Raman peaks (G and 2D) and a very low D peak, indicative of very low disorder and extended graphene sheet. Inset: 2D peak fitted by a single Lorenzian peak (FWHM=25cm⁻¹, position=2702 cm⁻¹). (b) for thinner films the raw spectrum (blue) contains both the graphene and SiC contribution. The latter (black trace) is subtracted by a Non-negative Matrix Factorization method [159] to reveal a pure graphene Raman spectrum (red).

Figure 17: Epigraphene on 3C-SiC. (a) LEEM image of epigraphene on 3C-SiC (111), showing 98% monolayer coverage. The 3C-SiC substrates were grown by sublimation epitaxy on 6H-SiC, and epigraphene was grown at 2000°C under 1atm Ar (From Ref. [179]). (b) ARPES of epigraphene on n-type 3C-SiC(111) showing quasi-neutrality after hydrogen intercalation (From Ref. [180]). (c) Epitaxial graphene formation on 3C-SiC/Si thin films, LEED pattern shows rotational order on 3C-SiC(100) compatible with C-termination, and Bernal stacking on 3C-SiC(111) compatible with Si-termination. (From Ref. [175])

Figure 18: (a) 6µm field of view (FOV) of a sidewall graphene trench array. (b) An expanded 25µm FOV of (a) shows a parallel array used for area averaged ARPES measurements. (c-f) Examples of epigraphene sidewall structures produced by the structured growth method [135], demonstrating its versatility: (c) an array of parallel ribbons grown on the step edges of a vicinal surface, (d) graphene grown in many orientations on the sidewalls of complex convoluted trenches, (e) a Hall bar with eight transverse voltage probes, and (d) rings grown on the sidewalls of pillars. The top images in (a-d) and the left image in (f) represent the AFM topography; the corresponding bottom and right images are electrostatic force microscopy images showing bright areas where graphene grows at the edges of the etched patterns (From. Ref. [191]).

Figure 19: (a) Schematic definition of AC and ZZ graphene ribbons. (b) shows the orientation of the graphene Brillouin zone with respect to the ZZ and AC directions. (c) and (e) show a schematic of an AC edge step on SiC and the corresponding AFM image of an AC step edge array. (d) and (f) show a schematic of an ZZ edge step on SiC and the corresponding AFM image of an ZZ step edge array.

Figure 20: High Resolution cross-sectional TEM images of graphene layers grown on SiC steps. (a) - (b) Views of different edge terminations of natural sidewall graphene. Graphene layers are observed as dark contrast lines [From Ref. [192]]. (c) False-color image of a sub-facet region near the top of a 30nm deep etched facet sidewall (armchair). The image is an overlay of a high-angle annular dark-field image (HAADF) (red) that enhances the SiC and a low- angle annular dark-field image (green) that enhances the graphene. (d) HAADF atomic resolution scanning TEM image of a ZZ-edge step facet on 4H-SiC(0001) with 5 graphene layers, cross sectioned along the [1-120] direction. Note the difference of the distance between the first graphene layer and the SiC (From Ref. [198]).

Figure 21: (a) Schematic of how sidewall graphene is measured in ARPES. The Dirac cone from the (0001) surface is at an angel ϕ relative to the (0001) surface normal n₍₀₀₀₁₎. The Dirac cone from the sidewall graphene is at an angle θ relative to the Dirac cone of the (0001) surface. (b) A constant energy cut of reciprocal space as a function of the detector angle, θ , from an AC step array. Facet cones from facets on both sides of the trenches are marked.

Figure 22: Schematic of SiC step flow during H₂ etching: (a) AC step edges and (b) ZZ-step edges. (From Ref. [200])

Figure 23: Top-gated graphene on Si and C-faces. (a) Top gated Si-face graphene (40nm SiOHx,(HSQ e-beam resist)), gate effect on the resistivity ρ_{xx} (300K) and Hall effect ρ_{xy} (4K – 5T). Inset: image of the patterned Hall bar, with all graphene contacts, before and after top-gate deposition. On-off ratio=31 (From [320]). (b) Top gated C-face graphene (monolayer with 40nm alumina gate dielectric); 2 point conductivities. The two branches give FET

mobility μ_{FET} = 8700 cm²/(V.s) on the n-side, to be compared to Hall mobility μ_{Hall} = 7500cm²/(V.s) at n = 1.6 x 10¹²/cm² (Vg = 0 V), and $\mu_{FET} \sim 5000$ cm²/(V.s) on the p-side. Inset: AFM image of a C-face monolayer graphene (scale bar 5 µm) over SiC steps. The white lines are graphene pleats characteristic of C-face epigraphene. (From ref. [221]). (c) Top gated multilayer C-face graphene (40nm HfO₂) of mobility up to 5000 cm²/V.s. Inset: optical image of the multiple devices patterned on a single SiC chip (From Ref. [212]).

Figure 24: Weak anti-localization in multilayer epigraphene (C-face) (a) Low-field magnetoresistance for a sample of 100 μ mx1000 μ m at various temperatures. Inset: magnetoresistance peak near B=0 at 1.4 K. The electron mobility is 11600 cm²/V.s, and the transport time τ =0:26 ps. (b) Low-field resistance (open circles) at 4.2 K after subtracting the high temperature background resistance. Solid line is a fit to the model of weak anti-localization, showing excellent agreement with the data. The only temperature dependent parameter is the phase coherence

time τ_{ϕ} potted in the inset, that is consistent with electron-electron scattering time. (From ref. [50])

Figure 25: Landau level magneto-spectroscopy. (a) Temperature dependence of the transmission spectra taken at B=0.8T, showing the $L_{-1}(0)$ to $L_0(1)$ transition up to room temperature. Successive spectra are shifted vertically by 0.15 for clarity. The Lorentzian fits to the data (blue curve) indicate no peak shift or widening up to room temperature (from Ref. [52]). (b) Position of the absorption lines as a function of the square root of the magnetic field. Dashed lines are calculated energy of the transitions between Landau levels. The transitions $L_{-m(-n)}$ to $L_{n(m)}$ are labeled corresponding to the diagram (color coded) in the inset.

Figure 26: Landau level tunneling spectroscopy. (a) Scanning Tunneling Spectroscopy at low temperature in 5T of multilayer epigraphene (C-face) showing graphene Landau level peaks in the local density of states. Insert 5x5nm STM image. The scale is in pm. (From Ref. [321]). (b) High-resolution Landau level spectroscopy of the fourfold states that make up the N=1 Landau level above 11 Tesla. The large energy separation between peaks corresponds to valley splitting (ΔE_v); The spin splitting is indicated by up (down) arrows in each valley. (From Ref. [54])

Figure 27: Magneto-transport in epigraphene C-face (a-b) and Si-face (c-d). (a) Multilayer C-face Hall bar (1 μ m x 5 μ m, mobility= 12 500 cm² /V.s, charge density= 4.5.x 10¹² cm⁻²). Shubnikov de Haas oscillations are observed in the magnetoresistance (main panel), as outlined once a background is subtracted (bottom panel). The level index is plotted as a function of 1/B, where B is the position of the magnetoresistance maxima (From Ref. [322]). (b) Monolayer C-face at low charge density (n=1.9x10¹¹ cm⁻²) and high mobility (39,800 cm²/V.s). The v=2 ρ_{xy} plateau and vanishingly small ρ_{xx} are observed as low as 3 T at 4K. Inset quantum Hall effect observed at 200K, well above liquid nitrogen temperature (From Ref. [137]). (c) Shubnikov-de Haas oscillation in ρ_{xx} and corresponding ρ_{xy} for monolayer Si-face at high charge density. (From Ref. [1]). (d) QHE for metrology in Si-face epigraphene. At lower charge density, quantization is observed above 3T (main panel). The relative deviation Δ RH/RH of the Hall resistance RH to h/(2e²) and of R_{xx} are plotted in the top and bottom panel respectively. It shows a standard uncertainty around 1x10⁻⁹ (From Ref. [76]).

Figure 28: High frequency epigraphene transistors. (a) Scanning electron microscopy image of a symmetric dual gate transistor, featuring a graphene channel between source (S) and drain (D) modulated by a gate (G) fabricated on C-face graphene. (b) High frequency characteristics of a 100nm gated channel. The cutoff frequency is f_T = 110GHz for the current gain (obtained by extrapolation of theoretical slope of 20 dB/decade to |H21| = 1, where |H21| = $\partial Ids / \partial Igs$, with Igs the gate– source current) and f_{max} = 70GHz for the power gain U=1/2. (From Ref. [221]). (c) State-of-the-art graphene high-frequency electronics (stars) compared with high electron mobility transistors (From Ref. [242]). Red stars: C-face epigraphene FET [221]. Vertical bar indicates record f_T in transferred graphene [244].

Figure 29: Spintronics. (a) Resistance versus magnetic field at 1.4 K presenting a large resistance change $\Delta R=1.7 M\Omega$ when the magnetization of the two Co electrodes are parallel or anti parallel. High tunnel resistances are important for efficient spin injection (from Ref. [254]). The arrows indicate the direction of the magnetic sweep; the peak shift
is due to magnetic hysteresis of the Co contacts. (b) Scanning electron micrograph of the two-terminal lateral spin valve. The two Al_2O_3/Co electrodes (red) are deposited on a $10\mu m$ wide epigraphene channel grown on the C-face.

Figure 30: Wafer scale graphene transistor integration. (a) Integration of 10,000 sidewall epigraphene transistors on a 4x6 mm SiC chip, with Al_2O_3 gate dielectric. (From Ref. [135]). (b) 100 mm wafers scale FETs utilizing hydrogenated Si-face graphene and HfO₂ gate dielectric (From Ref. [68]).

Figure 31: Combined graphene - SiC field effect transistors. (left panels) Semiconductor SiC forms the transistor channel that is provided with graphene ohmic source and drain contacts on the Si-face (made of monolayer graphene + buffer, see (b)). Hydrogenated graphene (bilayer graphene, see (c)) produces the FET Schottky-like gate. The device geometry is shown in the electron micrograph of (d) (From Ref. [260]). (right panel) The SiC/SiOx interface forms the transistor channel on the C-face, as sketched in the inset. Contacts are provided by multilayer epigraphene. Inset (bottom) AFM image of the device, the channel is in between the bright graphene pads. Current on-off ratios of more than 10⁶ are demonstrated at room temperature.(From Ref. [261])

Figure 32: (a) concept of Si-SiC wafer bonding. The top Si wafer supports the CMOS platform and the SiC wafer the epigraphene electronic devices. The devices on top and bottom wafers are connected by metal vias through the thin Si wafer. (b) Realization of the bonding after SiC graphitization with an intermediate Al₂O₃ bonding layer. (From Ref. [265])

Figure 33: Ballistic transport in sidewall ribbons (From Ref. [136]) (a) Resistance R versus probe spacing d. The tungsten probes are positioned on a single ribbon in situ in an electron microscope. The different traces correspond to ribbons after various cleansing by in-situ annealing. Linear fits extrapolate to $R(d=0)=h/e^2=25.8 \text{ k}\Omega$, Slope $\Delta R/\Delta L$ gives mean free path $\lambda=4.2$, 28, 16, 58 µm (top to bottom). Cyan line is consistent with zero slope. (b) Conductance versus probe spacing with the same set up as in (a) for an extended probe spacing range, showing the decay of one conducting channel after d=250nm, and the second at d=17µm. (c) Temperature dependence of the conductance G of a top-gated ribbon with fixed contacts (39nm x 1.6µm). Note the conductance G=0.9 h/e² at charge neutrality, and the weak temperature and gate voltage V_g dependences. Inset: AFM and EFM images of the ribbon grown on a curved SiC natural step. S, D and G refer to the Source, Drain and Gate area respectively. (d) All the G vs V_g curves are parallel and collapse on a single curve indicating that only the resistance at $V_g=0$ is temperature dependent.

Figure 34: Pump Probe Spectroscopy. (a) Band structure and carrier distribution after optical excitation. The pump photons (red) excite carriers that rapidly thermalize establishing a new hot thermal distribution that slowly cools down (From Ref. [203]. (b) After optical excitation an optical probe probes the carrier distribution. (left) For neutral (gapless) graphene, all transitions are allowed. (right) For doped graphene, only excitations larger than twice the Fermi level can create an electron-hole pair (Fermi Blockade).

Figure 35: Plasmons launched and imaged with SNOM. (a) Schematics of the experimental configuration). An infrared laser light illuminates a metallized AFM tip (yellow). (b) The near-field amplitude image (colour scale) is acquired for a tapered epigraphene ribbon ($12 \mu m \log p$). (c) calculated local density of optical states (LDOS) at a distance of 60nm from the graphene surface.[304]

9. References

C. Berger, Z.M. Song, T.B. Li, X.B. Li, A.Y. Ogbazghi, R. Feng, Z.T. Dai, A.N. Marchenkov, E.H. Conrad, P.N. First, W.A. De Heer, Ultrathin epitaxial graphite: 2D electron gas properties and a route toward graphene-based nanoelectronics, J Phys Chem B, 108 (2004) 19912-19916.
 W.A. de Heer, C. Berger, P.N. First, Patterned thin films graphite devices and methods for making the same, US patent 7015142, (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 2006), (provisional Application No.60/477,997 filed June 2007, Issued March 2007, Issued

Application No.60/477,997 filed June 12 2003, Issued March 21 2006), (provisional Application No.60/477,997 filed June 12 2003, Issued March 21 2006).

[3] K.S. Novoselov, D. Jiang, F. Schedin, T.J. Booth, V.V. Khotkevich, S.V. Morozov, A.K. Geim, Two-dimensional atomic crystals, Proceedings of the National Academy of Sciences of the United States of America, 102 (2005) 10451-10453.

[4] K.S. Novoselov, A.K. Geim, S.V. Morozov, D. Jiang, Y. Zhang, S.V. Dubonos, I.V. Grigorieva, A.A. Firsov, Electric field effect in atomically thin carbon films, Science, 306 (2004) 666.

[5] Y.B. Zhang, J.P. Small, M.E.S. Amori, P. Kim, Electric field modulation of galvanomagnetic properties of mesoscopic graphite, Physical Review Letters, 94 (2005) 176803.

[6] H.P. Boehm, A. Clauss, U. Hofmann, G.O. Fischer, Dunnste Kohlenstoff-Folien, Z Naturforsch Pt B, B 17 (1962) 150-&.

[7] H.P. Boehm, R. Setton, E. Stumpp, Nomenclature and Terminology of Graphite-Intercalation Compounds, Carbon, 24 (1986) 241-245.

[8] H.P. Boehm, Graphene-How a Laboratory Curiosity Suddenly Became Extremely Interesting, Angew Chem Int Edit, 49 (2010) 9332-9335.

[9] M. Sprinkle, D. Siegel, Y. Hu, J. Hicks, A. Tejeda, A. Taleb-Ibrahimi, P. Le Fevre, F. Bertran, S. Vizzini, H. Enriquez, S. Chiang, P. Soukiassian, C. Berger, W.A. de Heer, A. Lanzara, E.H. Conrad, First Direct Observation of a Nearly Ideal Graphene Band Structure, Physical Review Letters, 103 (2009) 226803.

[10] T. Ohta, A. Bostwick, J.L. McChesney, T. Seyller, K. Horn, E. Rotenberg, Interlayer interaction and electronic screening in multilayer graphene investigated with angle-resolved photoemission spectroscopy, Physical Review Letters, 98 (2007) 206802.

[11] S.Y. Zhou, G.H. Gweon, J. Graf, A.V. Fedorov, C.D. Spataru, R.D. Diehl, Y. Kopelevich, D.H. Lee, S.G. Louie, A. Lanzara, First direct observation of Dirac fermions in graphite, Nat Phys, 2 (2006) 595-599.

[12] K.S. Novoselov, A.K. Geim, S.V. Morozov, D. Jiang, M.I. Katsnelson, I.V. Grigorieva, S.V. Dubonos, A.A. Firsov, Two-dimensional gas of massless Dirac fermions in graphene, Nature, 438 (2005) 197.

[13] Y.B. Zhang, Y.W. Tan, H.L. Stormer, P. Kim, Experimental observation of the quantum Hall effect and Berry's phase in graphene, Nature, 438 (2005) 201.

[14] D.V. Badami, Graphitization of Alpha-Silicon Carbide, Nature, 193 (1962) 569-&.

[15] D.V. Badami, X-ray studies of graphite formed by decomposing sicilon carbide Carbon, 3 (1965) 53-57.

[16] A.J. Van Bommel, J.E. Crobeen, A. Van Tooren, LEED and Auger electron observations of the SiC(0001) surface, Surf Sci, 48 (1975) 463-472.

[17] E.G. Acheson, Production of Artificial Crystalline Carbonaceous Materials : A method for making Carborundum, an industrial abrasive called also silicon carbide., in, Carborundum Company, Pennsylvania, USA, 1893.

[18] N. Zheludev, The life and times of the LED - a 100-year history, Nat. Photonics, 1 (2007) 189-192.

[19] W.A. de Heer, The invention of graphene electronics and the physics of epitaxial graphene on silicon carbide, Phys Scripta, T146 (2012) 014004.

[20] D.T. Clark, E.P. Ramsay, A.E. Murphy, D.A. Smith, R.F. Thompson, R.A.R. Young, J.D. Cormack, C. Zhu, S. Finney, J. Fletcher, High Temperature Silicon Carbide CMOS Integrated Circuits, Mater Sci Forum, 679-680 (2011) 726-729.

[21] K.S. Novoselov, V.I. Fal'ko, L. Colombo, P.R. Gellert, M.G. Schwab, K. Kim, A roadmap for graphene, Nature, 490 (2012) 192-200.

[22] A.C. Ferrari, F. Bonaccorso, V. Fal'ko, K.S. Novoselov, S. Roche, P. Boggild, S. Borini, F.H.L. Koppens, V. Palermo, N. Pugno, J.A. Garrido, R. Sordan, A. Bianco, L. Ballerini, M. Prato, E. Lidorikis, J. Kivioja, C. Marinelli, T. Ryhanen, A. Morpurgo, J.N. Coleman, V. Nicolosi, L. Colombo, A. Fert, M. Garcia-Hernandez, A. Bachtold, G.F. Schneider, F. Guinea, C. Dekker, M. Barbone, Z. Sun, C. Galiotis, A.N. Grigorenko, G. Konstantatos, A. Kis, M. Katsnelson, L. Vandersypen, A. Loiseau, V. Morandi, D. Neumaier, E. Treossi, V. Pellegrini, M. Polini, A. Tredicucci, G.M. Williams, B. Hee Hong, J.-H.

Ahn, J. Min Kim, H. Zirath, B.J. van Wees, H. van der Zant, L. Occhipinti, A. Di Matteo, I.A. Kinloch, T. Seyller, E. Quesnel, X. Feng, K. Teo, N. Rupesinghe, P. Hakonen, S.R.T. Neil, Q. Tannock, T. Lofwander, J. Kinaret, Science and technology roadmap for graphene, related two-dimensional crystals, and hybrid systems, Nanoscale, 7 (2015) 4598-4810.

[23] H.P. Boehm, R. Setton, E. Stumpp, Nomenclature and Terminology of Graphite-Intercalation Compounds (lupac Recommendations 1994), Pure Appl Chem, 66 (1994) 1893-1901.

[24] A.K. Geim, Graphene: Status and Prospects, Science, 324 (2009) 1530-1534.

[25] A. Bianco, H.M. Cheng, T. Enoki, Y. Gogotsi, R.H. Hurt, N. Koratkar, T. Kyotani, M. Monthioux, C.R. Park, J.M.D. Tascon, J. Zhang, All in the graphene family - A recommended nomenclature for two-dimensional carbon materials, Carbon, 65 (2013) 1-6.

[26] L.C. Pauling, The nature of the chemical bond : and the structure of molecules and crystals. An introduction to modern structural chemistry, (Cornell University Press, New York, 1948.

[27] W. Kratschmer, L.D. Lamb, K. Fostiropoulos, D.R. Huffman, Solid C-60 - a New Form of Carbon, Nature, 347 (1990) 354-358.

[28] T.W. Ebbesen, P.M. Ajayan, Large scale synthesis of carbon nanotubes, Nature, 358 (1992) 220-222.

[29] L.A. Girifalco, R.A. Lad, Energy of Cohesion, Compressibility, and the Potential Energy Functions of the Graphite System, J Chem Phys, 25 (1956) 693-697.

[30] K.R. Paton, E. Varrla, C. Backes, R.J. Smith, U. Khan, A. O'Neill, C. Boland, M. Lotya, O.M. Istrate, P. King, T. Higgins, S. Barwich, P. May, P. Puczkarski, I. Ahmed, M. Moebius, H. Pettersson, E. Long, J. Coelho, S.E. O'Brien, E.K. McGuire, B.M. Sanchez, G.S. Duesberg, N. McEvoy, T.J. Pennycook, C. Downing, A. Crossley, V. Nicolosi, J.N. Coleman, Scalable production of large quantities of defect-free few-layer graphene by shear exfoliation in liquids, Nature Materials, 13 (2014) 624-630.

[31] Y. Hernandez, V. Nicolosi, M. Lotya, F.M. Blighe, Z. Sun, S. De, I.T. McGovern, B. Holland, M. Byrne, Y.K. Gun'Ko, J.J. Boland, P. Niraj, G. Duesberg, S. Krishnamurthy, R. Goodhue, J. Hutchison, V. Scardaci, A.C. Ferrari, J.N. Coleman, High-yield production of graphene by liquid-phase exfoliation of graphite, Nat Nano, 3 (2008) 563-568.

[32] H.N. America, Aquadag® Water Based Graphite Coating / Additive, in, 2015.

[33] N.R. Gall, E.V. RutKov, A.Y. Tontegode, Two dimensional graphite films on metals and their intercalation, INTERNATIONAL JOURNAL OF MODERN PHYSICS B, 11 (1997) 1865-1911.

[34] W.A. de Heer, C. Berger, M. Ruan, M. Sprinkle, X. Li, Y. Hu, B. Zhang, J. Hankinson, E.H. Conrad, Large area and structured epitaxial graphene produced by confinement controlled sublimation of silicon carbide, Proc Nat Acad Sci, 108 (2011) 16900-16905.

[35] I. Forbeaux, J.M. Themlin, J.M. Debever, Heteroepitaxial graphite on 6H-SiC(0001): Interface formation through conduction-band electronic structure, Phys Rev B, 58 (1998) 16396-16406.

[36] L. Xuekun, Y. Minfeng, H. Hui, S.R. Rodney, Tailoring graphite with the goal of achieving single sheets, Nanotechnology, 10 (1999) 269.
 [37] J.W.T. Seo, A.A. Green, A.L. Antaris, M.C. Hersam, High-Concentration Aqueous Dispersions of Graphene Using Nonionic, Biocompatible Block Copolymers, J Phys Chem Lett, 2 (2011) 1004-1008.

[38] P.R. Wallace, The Band Theory of Graphite, Phys. Rev., 71 (1947) 622-634.

[39] J.C. Slonczewski, P.R. Weiss, Band Structure of Graphite, Phys. Rev., 109 (1958) 272-279.

[40] A.H. Castro Neto, F. Guinea, N.M.R. Peres, K.S. Novoselov, A.K. Geim, The electronic properties of graphene, Rev Mod Phys, 81 (2009) 109-162.

[41] A. Gruneis, C. Attaccalite, L. Wirtz, H. Shiozawa, R. Saito, T. Pichler, A. Rubio, Tight-binding description of the quasiparticle dispersion of graphite and few-layer graphene, Phys Rev B, 78 (2008) 205425.

[42] J.W. McClure, Band Structure of Graphite and de Haas-van Alphen Effect, Phys. Rev., 108 (1957) 612-618.

[43] Y.W. Son, M.L. Cohen, S.G. Louie, Half-metallic graphene nanoribbons, Nature, 444 (2006) 347-349.

[44] D.P. Divincenzo, E.J. Mele, Self-Consistent Effective-Mass Theory for Intralayer Screening in Graphite-Intercalation Compounds, Phys Rev B, 29 (1984) 1685-1694.

[45] T. Ando, T. Nakanishi, R. Saito, Berry's phase and absence of back scattering in carbon nanotubes, J Phys Soc Jpn, 67 (1998) 2857-2862. [46] D.V. Khveshchenko, Ghost excitonic insulator transition in layered graphite, Physical Review Letters, 87 (2001).

[47] C. Hwang, D.A. Siegel, S.K. Mo, W. Regan, A. Ismach, Y.G. Zhang, A. Zettl, A. Lanzara, Fermi velocity engineering in graphene by substrate modification, Sci Rep-Uk, 2 (2012) 590 (594pp).

[48] L.A. Ponomarenko, R.V. Gorbachev, G.L. Yu, D.C. Elias, R. Jalil, A.A. Patel, A. Mishchenko, A.S. Mayorov, C.R. Woods, J.R. Wallbank, M. Mucha-Kruczynski, B.A. Piot, M. Potemski, I.V. Grigorieva, K.S. Novoselov, F. Guinea, V.I. Fal'ko, A.K. Geim, Cloning of Dirac fermions in graphene superlattices, Nature, 497 (2013) 594-597.

[49] C. Berger, Z.M. Song, X.B. Li, X.S. Wu, N. Brown, C. Naud, D. Mayou, T.B. Li, J. Hass, A.N. Marchenkov, E.H. Conrad, P.N. First, W.A. de Heer, Electronic confinement and coherence in patterned epitaxial graphene, Science, 312 (2006) 1191-1196.

[50] X.S. Wu, X.B. Li, Z.M. Song, C. Berger, W.A. de Heer, Weak antilocalization in epitaxial graphene: Evidence for chiral electrons, Physical Review Letters, 98 (2007) 266405.

[51] M.L. Sadowski, G. Martinez, M. Potemski, C. Berger, W.A. de Heer, Landau level spectroscopy of ultrathin graphite layers, Physical Review Letters, 97 (2006) 266405.

[52] M. Orlita, C. Faugeras, P. Plochocka, P. Neugebauer, G. Martinez, D.K. Maude, A.L. Barra, M. Sprinkle, C. Berger, W.A. de Heer, M. Potemski, Approaching the Dirac Point in High-Mobility Multilayer Epitaxial Graphene, Physical Review Letters, 101 (2008) 267601.

[53] C. Faugeras, A. Nerriere, M. Potemski, A. Mahmood, E. Dujardin, C. Berger, W.A. de Heer, Few-layer graphene on SiC, pyrolitic graphite, and graphene: A Raman scattering study, Appl Phys Lett, 92 (2008) 011914.

[54] Y.J. Song, A.F. Otte, Y. Kuk, Y.K. Hu, D.B. Torrance, P.N. First, W.A. de Heer, H.K. Min, S. Adam, M.D. Stiles, A.H. MacDonald, J.A. Stroscio, Highresolution tunnelling spectroscopy of a graphene quartet, Nature, 467 (2010) 185-189.

[55] J. Martin, N. Akerman, G. Ulbricht, T. Lohmann, J.H. Smet, K. Von Klitzing, A. Yacoby, Observation of electron-hole puddles in graphene using a scanning single-electron transistor, Nat Phys, 4 (2008) 144-148.

[56] J.H. Chen, C. Jang, S.D. Xiao, M. Ishigami, M.S. Fuhrer, Intrinsic and extrinsic performance limits of graphene devices on SiO2, Nat Nanotechnol, 3 (2008) 206-209.

[57] Y.W. Tan, Y. Zhang, K. Bolotin, Y. Zhao, S. Adam, E.H. Hwang, S. Das Sarma, H.L. Stormer, P. Kim, Measurement of scattering rate and minimum conductivity in graphene, Physical Review Letters, 99 (2007) -.

[58] S.V. Morozov, K.S. Novoselov, M.I. Katsnelson, F. Schedin, L.A. Ponomarenko, D. Jiang, A.K. Geim, Strong suppression of weak localization in graphene, Physical Review Letters, 97 (2006).

[59] S. Das Sarma, S. Adam, E.H. Hwang, E. Rossi, Electronic transport in two-dimensional graphene, Rev Mod Phys, 83 (2011) 407-470.
[60] C.R. Dean, A.F. Young, I. Meric, C. Lee, L. Wang, S. Sorgenfrei, K. Watanabe, T. Taniguchi, P. Kim, K.L. Shepard, J. Hone, Boron nitride substrates for high-quality graphene electronics, Nat Nanotechnol, 5 (2010) 722-726.

[61] M. Fujita, K. Wakabayashi, K. Nakada, K. Kusakabe, Peculiar localized state at zigzag graphite edge, J Phys Soc Jpn, 65 (1996) 1920-1923.
 [62] K. Nakada, M. Fujita, G. Dresselhaus, M.S. Dresselhaus, Edge state in graphene ribbons: Nanometer size effect and edge shape dependence, Phys Rev B, 54 (1996) 17954-17961.

[63] S.E. Saddow, A. Anant Agarwal, Advances in Silicon Carbide - Processing and Applications, Artech House, 2004.

[64] A. Bauer, J. Kräußlich, L. Dressler, P. Kuschnerus, J. Wolf, K. Goetz, P. Käckell, J. Furthmüller, F. Bechstedt, High-precision determination of atomic positions in crystals: The case of 6H- and 4H-SiC, Phys Rev B, 57 (1998) 2647-2650.

[65] J. Hass, W.A. de Heer, E.H. Conrad, The growth and morphology of epitaxial multilayer graphene, J Phys-Condens Mat, 20 (2008) 323202.
[66] K.V. Emtsev, A. Bostwick, K. Horn, J. Jobst, G.L. Kellogg, L. Ley, J.L. McChesney, T. Ohta, S.A. Reshanov, J. Rohrl, E. Rotenberg, A.K. Schmid, D. Waldmann, H.B. Weber, T. Seyller, Towards wafer-size graphene layers by atmospheric pressure graphitization of silicon carbide, Nature Materials, 8 (2009) 203-207.

[67] S.K. Lilov, Study of the equilibrium processes in the gas phase during silicon carbide sublimation, Materials Science and Engineering: B, 21 (1993) 65-69.

[68] J.A. Robinson, M. Hollander, M. LaBella, K.A. Trumbull, R. Cavalero, D.W. Snyder, Epitaxial Graphene Transistors: Enhancing Performance via Hydrogen Intercalation, Nano Lett, 11 (2011) 3875-3880.

[69] T. Yager, A. Lartsev, R. Yakimova, S. Lara-Avila, S. Kubatkin, Wafer-scale homogeneity of transport properties in epitaxial graphene on SiC, Carbon, 87 (2015) 409-414.

[70] C. Dimitrakopoulos, Y.M. Lin, A. Grill, D.B. Farmer, M. Freitag, Y.N. Sun, S.J. Han, Z.H. Chen, K.A. Jenkins, Y. Zhu, Z.H. Liu, T.J. McArdle, J.A. Ott, R. Wisnieff, P. Avouris, Wafer-scale epitaxial graphene growth on the Si-face of hexagonal SiC (0001) for high frequency transistors, J Vac Sci Technol B, 28 (2010) 985-992.

[71] R. Yakimova, T. Iakimov, G.R. Yazdi, C. Bouhafs, J. Eriksson, A. Zakharov, A. Boosalis, M. Schubert, V. Darakchieva, Morphological and electronic properties of epitaxial graphene on SiC, Physica B, 439 (2014) 54-59.

[72] R.M. Tromp, J.B. Hannon, Thermodynamics and Kinetics of Graphene Growth on SiC(0001), Physical Review Letters, 102 (2009) 106104.
 [73] W. Strupinski, K. Grodecki, A. Wysmolek, R. Stepniewski, T. Szkopek, P.E. Gaskell, A. Gruneis, D. Haberer, R. Bozek, J. Krupka, J.M. Baranowski,

Graphene Epitaxy by Chemical Vapor Deposition on SiC, Nano Lett, 11 (2011) 1786-1791.

[74] E. Moreau, F.J. Ferrer, D. Vignaud, S. Godey, X. Wallart, Graphene growth by molecular beam epitaxy using a solid carbon source, Phys Status Solidi A, 207 (2010) 300-303.

[75] A. Michon, S. Vezian, E. Roudon, D. Lefebvre, M. Zielinski, T. Chassagne, M. Portail, Effects of pressure, temperature, and hydrogen during graphene growth on SiC(0001) using propane-hydrogen chemical vapor deposition, J Appl Phys, 113 (2013) 203501.

[76] F. Lafont, R. Ribeiro-Palau, D. Kazazis, A. Michon, O. Couturaud, C. Consejo, T. Chassagne, M. Zielinski, M. Portail, B. Jouault, F. Schopfer, W. Poirier, Quantum Hall resistance standards from graphene grown by chemical vapour deposition on silicon carbide, Nat. Commun., 6 (2015) 6806.

[77] A.L. Walter, S. Nie, A. Bostwick, K.S. Kim, L. Moreschini, Y.J. Chang, D. Innocenti, K. Horn, K.F. McCarty, E. Rotenberg, Electronic structure of graphene on single-crystal copper substrates, Phys Rev B, 84 (2011) 195443.

[78] K.V. Emtsev, F. Speck, T. Seyller, L. Ley, J.D. Riley, Interaction, growth, and ordering of epitaxial graphene on SiC{0001} surfaces: A comparative photoelectron spectroscopy study, Phys Rev B, 77 (2008) 155303

[79] E. Rollings, G.H. Gweon, S.Y. Zhou, B.S. Mun, J.L. McChesney, B.S. Hussain, A. Fedorov, P.N. First, W.A. de Heer, A. Lanzara, Synthesis and characterization of atomically thin graphite films on a silicon carbide substrate, J Phys Chem Sol, 67 (2006) 2172-2177.

[80] T. Ohta, A. Bostwick, T. Seyller, K. Horn, E. Rotenberg, Controlling the electronic structure of bilayer graphene, Science, 313 (2006) 951-954.
[81] E.V. Castro, K.S. Novoselov, S.V. Morozov, N.M.R. Peres, J.M.B.L. Dos Santos, J. Nilsson, F. Guinea, A.K. Geim, A.H. Castro Neto, Biased bilayer graphene: Semiconductor with a gap tunable by the electric field effect, Physical Review Letters, 99 (2007) 216802.

[82] C. Coletti, S. Forti, A. Principi, K.V. Emtsev, A.A. Zakharov, K.M. Daniels, B.K. Daas, M.V.S. Chandrashekhar, T. Ouisse, D. Chaussende, A.H. MacDonald, M. Polini, U. Starke, Revealing the electronic band structure of trilayer graphene on SiC: An angle-resolved photoemission study, Physical Review B, 88 (2013) 155439.

[83] A. Bostwick, F. Speck, T. Seyller, K. Horn, M. Polini, R. Asgari, A.H. MacDonald, E. Rotenberg, Observation of Plasmarons in Quasi-Freestanding Doped Graphene, Science, 328 (2010) 999-1002.

[84] A. Bostwick, T. Ohta, T. Seyller, K. Horn, E. Rotenberg, Quasiparticle dynamics in graphene, Nat Phys, 3 (2007) 36-40.

[85] S.Y. Zhou, D.A. Siegel, A.V. Fedorov, A. Lanzara, Kohn anomaly and interplay of electron-electron and electron-phonon interactions in epitaxial graphene, Phys Rev B, 78 (2008) 193404.

[86] D.N. Basov, M.M. Fogler, A. Lanzara, F. Wang, Y.B. Zhang, Colloquium: Graphene spectroscopy, Rev Mod Phys, 86 (2014) 959-994.

[87] S.Y. Zhou, G.H. Gweon, A.V. Fedorov, P.N. First, W.A. de Heer, D.H. Lee, F. Guinea, A.H. Castro Neto, A. Lanzara, Substrate induced band gap opening in epitaxial graphene, Nature Materials, 6 (2007) 770.

[88] F. Varchon, R. Feng, J. Hass, X. Li, B. Ngoc Nguyen, C. Naud, P. Mallet, J.-Y. Veuillen, C. Berger, E.H. Conrad, L. Magaud, Electronic structure of epitaxial graphene layers on SiC: effect of the substrate, Physical Review Letters, 99 (2007) 126805

[89] A. Mattausch, O. Pankratov, Ab initio study of graphene on SiC, Physical Review Letters, 99 (2007) 076802.

[90] G.M. Rutter, N.P. Guisinger, J.N. Crain, E.A.A. Jarvis, M.D. Stiles, T. Li, P.N. First, J.A. Stroscio, Imaging the interface of epitaxial graphene with silicon carbide via scanning tunneling microscopy, Phys Rev B, 76 (2007) 235416.

[91] C. Riedl, C. Coletti, T. Iwasaki, A.A. Zakharov, U. Starke, Quasi-Free-Standing Epitaxial Graphene on SiC Obtained by Hydrogen Intercalation, Physical Review Letters, 103 (2009) 246804.

[92] P. Mallet, F. Varchon, C. Naud, L. Magaud, C. Berger, J.-Y. Veuillen, Electron states of mono- and bilayer graphene on SiC probed by scanningtunneling microscopy, Physical review B (Rapid Communication), 76 (2007) 041403(R).

[93] F. Owman, P. Martensson, The SiC(0001) 6V3x6V3 reconstruction studied with STM and LEED, Surf Sci, 369 (1996) 126-136.

[94] L. Li, I.S.T. Tsong, Atomic structures of 6H-SiC(0001) and (0001) surfaces, Surf Sci, 351 (1996) 141-148.

[95] T. Tsukamoto, M. Hirai, M. Kusaka, M. Iwami, T. Ozawa, T. Nagamura, T. Nakata, Annealing effect on surfaces of 4H(6H)-SiC(0001)Si face, Applied Surface Science, 113 (1997) 467-471.

[96] C.S. Chang, I.S.T. Tsong, Y.C. Wang, R.F. Davis, Scanning Tunneling Microscopy and Spectroscopy of Cubic Beta-Sic(111) Surfaces, Surf Sci, 256 (1991) 354-360.

[97] W. Chen, H. Xu, L. Liu, X.Y. Gao, D.C. Qi, G.W. Peng, S.C. Tan, Y.P. Feng, K.P. Loh, A.T.S. Wee, Atomic structure of the 6H-SiC(0001) nanomesh, Surf Sci, 596 (2005) 176-186.

[98] C. Riedl, U. Starke, J. Bernhardt, M. Franke, K. Heinz, Structural properties of the graphene-SiC(0001) interface as a key for the preparation of homogeneous large-terrace graphene surfaces, Phys Rev B, 76 (2007) 245406.

[99] A. Charrier, A. Coati, T. Argunova, F. Thibaudau, Y. Garreau, R. Pinchaux, I. Forbeaux, J.M. Debever, M. Sauvage-Simkin, J.M. Themlin, Solidstate decomposition of silicon carbide for growing ultra-thin heteroepitaxial graphite films, J Appl Phys, 92 (2002) 2479-2484.

[100] J.D. Emery, B. Detlefs, H.J. Karmel, L.O. Nyakiti, D.K. Gaskill, M.C. Hersam, J. Zegenhagen, M.J. Bedzyk, Chemically Resolved Interface Structure of Epitaxial Graphene on SiC(0001), Physical Review Letters, 111 (2013) 215501.

[101] S. Forti, U. Starke, Epitaxial graphene on SiC: from carrier density engineering to quasi-free standing graphene by atomic intercalation, J. Phys. D-Appl. Phys., 47 (2014) 094013.

[102] J.D. Emery, V.H. Wheeler, J.E. Johns, M.E. McBriarty, B. Detlefs, M.C. Hersam, D.K. Gaskill, M.J. Bedzyk, Structural consequences of hydrogen intercalation of epitaxial graphene on SiC(0001), Appl Phys Lett, 105 (2014) 161602.

[103] M.G. Silly, M. D'Angelo, A. Besson, Y.J. Dappe, S. Kubsky, G. Li, F. Nicolas, D. Pierucci, M. Thomasset, Electronic and structural properties of graphene-based metal-semiconducting heterostructures engineered by silicon intercalation, Carbon, 76 (2014) 27-39.

[104] L. Kubler, K. Ait-Mansour, M. Diani, D. Dentel, J.L. Bischoff, M. Derivaz, Bidimensional intercalation of Ge between SiC(0001) and a heteroepitaxial graphite top layer, Phys Rev B, 72 (2005) 115319.

[105] M.H. Oliveira, T. Schumann, F. Fromm, R. Koch, M. Ostler, M. Ramsteiner, T. Seyller, J.M.J. Lopes, H. Riechert, Formation of high-quality quasi-free-standing bilayer graphene on SiC(0001) by oxygen intercalation upon annealing in air, Carbon, 52 (2013) 83-89.

[106] M. Ostler, F. Fromm, R.J. Koch, P. Wehrfritz, F. Speck, H. Vita, S. Bottcher, K. Horn, T. Seyller, Buffer layer free graphene on SiC(0001) via interface oxidation in water vapor, Carbon, 70 (2014) 258-265.

[107] B. Premlal, M. Cranney, F. Vonau, D. Aubel, D. Casterman, M.M. De Souza, L. Simon, Surface intercalation of gold underneath a graphene monolayer on SiC(0001) studied by scanning tunneling microscopy and spectroscopy, Appl Phys Lett, 94 (2009) 263115.

[108] C. Virojanadara, S. Watcharinyanon, A.A. Zakharov, L.I. Johansson, Epitaxial graphene on 6H-SiC and Li intercalation, Phys Rev B, 82 (2010) 205402.

[109] C. Xia, L.I. Johansson, Y.R. Niu, A.A. Zakharov, E. Janzen, C. Virojanadara, High thermal stability quasi-free-standing bilayer graphene formed on 4H-SiC(0001) via platinum intercalation, Carbon, 79 (2014) 631-635.

[110] S.L. Wong, H. Huang, Y.Z. Wang, L. Cao, D.C. Qi, I. Santoso, W. Chen, A.T.S. Wee, Quasi-Free-Standing Epitaxial Graphene on SiC (0001) by Fluorine Intercalation from a Molecular Source, Acs Nano, 5 (2011) 7662-7668.

[111] I. Palacio, A. Celis, M.N. Nair, A. Gloter, A. Zobelli, M. Sicot, D. Malterre, M.S. Nevius, W.A. de Heer, C. Berger, E.H. Conrad, A. Taleb-Ibrahimi, A. Tejeda, Atomic Structure of Epitaxial Graphene Sidewall Nanoribbons: Flat Graphene, Miniribbons, and the Confinement Gap, Nano Lett, (2014).

[112] J. Hass, J.E. Millan-Otoya, P.N. First, E.H. Conrad, Interface structure of epitaxial graphene grown on 4H-SiC(0001), Phys Rev B, 78 (2008) 205424.

[113] T. Schumann, M. Dubslaff, M.H. Oliveira, M. Hanke, J.M.J. Lopes, H. Riechert, Effect of buffer layer coupling on the lattice parameter of epitaxial graphene on SiC(0001), Phys Rev B, 90 (2014) 041403(R).

[114] S. Kim, J. Ihm, H.J. Choi, Y.W. Son, Origin of anomalous electronic structures of epitaxial graphene on silicon carbide, Physical Review Letters, 100 (2008) 176802.

[115] F. Varchon, P. Mallet, J.Y. Veuillen, L. Magaud, Ripples in epitaxial graphene on the Si-terminated SiC(0001) surface, Phys Rev B, 77 (2008) 235412.

[116] V. Borovikov, A. Zangwill, Step bunching of vicinal 6H-SiC{0001} surfaces, Phys Rev B, 79 (2009) 121406(R).

[117] F. Ming, A. Zangwill, Model and simulations of the epitaxial growth of graphene on non-planar 6H-SiC surfaces, J. Phys. D-Appl. Phys., 45 (2012) 154007.

[118] A. Mahmood, P. Mallet, J.Y. Veuillen, Quasiparticle scattering off phase boundaries in epitaxial graphene, Nanotechnology, 23 (2012) 055706.

[119] H. Huang, W. Chen, S. Chen, A.T.S. Wee, Bottom-up Growth of Epitaxial Graphene on 6H-SiC(0001), Acs Nano, 2 (2008) 2513-2518.
 [120] T. Ohta, N.C. Bartelt, S. Nie, K. Thurmer, G.L. Kellogg, Role of carbon surface diffusion on the growth of epitaxial graphene on SiC, Phys Rev B, 81 (2010) 121411R.

[121] L.D. Xian, M.Y. Chou, Diffusion of Si and C atoms on and between graphene layers, J. Phys. D-Appl. Phys., 45 (2012) 455309.
 [122] L. Muehlhoff, W.J. Choyke, M.J. Bozack, J.T. Yates, Comparative Electron Spectroscopic Studies of Surface Segregation on Sic(0001) and Sic(0001bar), J Appl Phys, 60 (1986) 2842-2853.

[123] X.S. Wu, Y.K. Hu, M. Ruan, N.K. Madiomanana, J. Hankinson, M. Sprinkle, C. Berger, W.A. de Heer, Half integer quantum Hall effect in high mobility single layer epitaxial graphene, Appl Phys Lett, 95 (2009) 223108.

[124] N. Camara, J.R. Huntzinger, G. Rius, A. Tiberj, N. Mestres, F. Perez-Murano, P. Godignon, J. Camassel, Anisotropic growth of long isolated graphene ribbons on the C face of graphite-capped 6H-SiC, Phys Rev B, 80 (2009) 125410.

[125] T.C. Cai, Z.Z. Jia, B.M. Yan, D.P. Yu, X.S. Wu, Hydrogen assisted growth of high quality epitaxial graphene on the C-face of 4H-SiC, Appl Phys Lett, 106 (2015) 013106.

[126] J.Y. Veuillen, F. Hiebel, L. Magaud, P. Mallet, F. Varchon, Interface structure of graphene on SiC: an ab initio and STM approach, J. Phys. D-Appl. Phys., 43 (2010) 374008.

[127] P.N. First, Private communication.

[128] D.L. Miller, K.D. Kubista, G.M. Rutter, M. Ruan, W.A. de Heer, M. Kindermann, P.N. First, J.A. Stroscio, Real-space mapping of magnetically quantized graphene states, Nat Phys, 6 (2010) 811-817.

[129] J. Hass, F. Varchon, J.E. Millan-Otoya, M. Sprinkle, N. Sharma, W.A. De Heer, C. Berger, P.N. First, L. Magaud, E.H. Conrad, Why multilayer graphene on 4H-SiC(000(1)over-bar) behaves like a single sheet of graphene, Physical Review Letters, 100 (2008) 125504.

[130] R. Zhang, Y.L. Dong, W.J. Kong, W.P. Han, P.H. Tan, Z.M. Liao, X.S. Wu, D.P. Yu, Growth of large domain epitaxial graphene on the C-face of SiC, J Appl Phys, 112 (2012) 104307.

[131] J. Hicks, K. Shepperd, F. Wang, E.H. Conrad, The structure of graphene grown on the SiC (000(1)over-bar) surface, J. Phys. D-Appl. Phys., 45 (2012) 154002.

[132] H. Hibino, H. Kageshima, F. Maeda, M. Nagase, Y. Kobayashi, H. Yamaguchi, Microscopic thickness determination of thin graphite films formed on SiC from quantized oscillation in reflectivity of low-energy electrons, Phys Rev B, 77 (2008) 075413.

[133] X.Z. Yu, C.G. Hwang, C.M. Jozwiak, A. Kohl, A.K. Schmid, A. Lanzara, New synthesis method for the growth of epitaxial graphene, J Electron Spectrosc, 184 (2011) 100-106.

[134] F. Wang, G. Liu, S. Rothwell, M.S. Nevius, C. Mathieu, N. Barrett, A. Sala, T.O. Mentes, A. Locatelli, P.I. Cohen, L.C. Feldman, E.H. Conrad, Pattern induced ordering of semiconducting graphene ribbons grown from nitrogen-seeded SiC, Carbon, 82 (2015) 360-367.

[135] M. Sprinkle, M. Ruan, Y. Hu, J. Hankinson, M. Rubio-Roy, B. Zhang, X. Wu, C. Berger, W.A. de Heer, Scalable templated growth of graphene nanoribbons on SiC, Nat Nanotechnol, 5 (2010) 727-731.

[136] J. Baringhaus, M. Ruan, F. Edler, A. Tejeda, M. Sicot, A. Taleb-Ibrahimi, A.P. Li, Z.G. Jiang, E.H. Conrad, C. Berger, C. Tegenkamp, W.A. de Heer, Exceptional ballistic transport in epitaxial graphene nanoribbons, Nature, 506 (2014) 349-354.

[137] Y. Hu, M. Ruan, Z.L. Guo, R. Dong, J. Palmer, J. Hankinson, C. Berger, W.A. de Heer, Structured epitaxial graphene: growth and properties, J. Phys. D-Appl. Phys., 45 (2012) 154010.

[138] H. Fukidome, Y. Kawai, F. Fromm, M. Kotsugi, H. Handa, T. Ide, T. Ohkouchi, H. Miyashita, Y. Enta, T. Kinoshita, T. Seyller, M. Suemitsu, Precise control of epitaxy of graphene by microfabricating SiC substrate, Appl Phys Lett, 101 (2012) 041605.

[139] M. Rubio-Roy, F. Zaman, Y.K. Hu, C. Berger, M.W. Moseley, J.D. Meindl, W.A. de Heer, Structured epitaxial graphene growth on SiC by selective graphitization using a patterned AlN cap, Appl Phys Lett, 96 (2010) 082112.

[140] R. Puybaret, J. Hankinson, J. Palmer, C. Bouvier, A. Ougazzaden, P.L. Voss, C. Berger, W.A. de Heer, Scalable control of graphene growth on 4H-SiC C-face using decomposing silicon nitride masks, J. Phys. D-Appl. Phys., 48 (2015) 152001.

[141] D.L. Miller, K.D. Kubista, G.M. Rutter, M. Ruan, W.A. de Heer, P.N. First, J.A. Stroscio, Structural analysis of multilayer graphene via atomic moireacute interferometry, Phys Rev B, 81 (2010) 223108.

[142] R.E. Franklin, The structure of graphitic carbons, Acta Crystallographica, 4 (1951) 253-261.

[143] L.G. Cancado, M.A. Pimenta, R. Saito, A. Jorio, L.O. Ladeira, A. Grueneis, A.G. Souza, G. Dresselhaus, M.S. Dresselhaus, Stokes and anti-Stokes double resonance Raman scattering in two-dimensional graphite, Phys Rev B, 66 (2002) 035415.

[144] M. Sprinkle, J. Hicks, A. Tejeda, A. Taleb-Ibrahimi, P. Le Fevre, F. Bertran, H. Tinkey, M.C. Clark, P. Soukiassian, D. Martinotti, J. Hass, E.H. Conrad, Multilayer epitaxial graphene grown on the SiC (000(1)over-bar) surface; structure and electronic properties, J. Phys. D-Appl. Phys., 43 (2010) 374006.

[145] J. Hicks, M. Sprinkle, K. Shepperd, F. Wang, A. Tejeda, A. Taleb-Ibrahimi, F. Bertran, P. Le Fevre, W.A. de Heer, C. Berger, E.H. Conrad, Symmetry breaking in commensurate graphene rotational stacking: Comparison of theory and experiment, Phys Rev B, 83 (2011) 205403. [146] I. Forbeaux, J.M. Themlin, J.M. Debever, High-temperature graphitization of the 6H-SiC (000(1)over-bar) face, Surf Sci, 442 (1999) 9-18. [147] L.I. Johansson, S. Watcharinyanon, A.A. Zakharov, T. Iakimov, R. Yakimova, C. Virojanadara, Stacking of adjacent graphene layers grown on C-face SiC (vol 84, 125405, 2011), Phys Rev B, 84 (2011) 129902.

[148] G. Nicotra, I. Deretzis, M. Scuderi, C. Spinella, P. Longo, R. Yakimova, F. Giannazzo, A. La Magna, Interface disorder probed at the atomic scale for graphene grown on the C face of SiC, Phys Rev B, 91 (2015) 155411.

[149] D.L. Miller, K.D. Kubista, G.M. Rutter, M. Ruan, W.A. de Heer, P.N. First, J.A. Stroscio, Observing the Quantization of Zero Mass Carriers in Graphene, Science, 324 (2009) 924-927.

[150] F. Wang, K. Shepperd, J. Hicks, M.S. Nevius, H. Tinkey, A. Tejeda, A. Taleb-Ibrahimi, F. Bertran, P. Le Fevre, D.B. Torrance, P.N. First, W.A. de Heer, A.A. Zakharov, E.H. Conrad, Silicon intercalation into the graphene-SiC interface, Phys Rev B, 85 (2012) 165449.

[151] C. Mathieu, B. Lalmi, T.O. Menteş, E. Pallecchi, A. Locatelli, S. Latil, R. Belkhou, A. Ouerghi, Effect of oxygen adsorption on the local properties of epitaxial graphene on SiC (0001), Phys Rev B, 86 (2012) 035435.

[152] G.T. de Laissardiere, D. Mayou, L. Magaud, Localization of Dirac Electrons in Rotated Graphene Bilayers, Nano Lett, 10 (2010) 804-808.
 [153] J.M.B.L. dos Santos, N.M.R. Peres, A.H. Castro, Graphene bilayer with a twist: Electronic structure, Physical Review Letters, 99 (2007) 256802

[154] M. Kindermann, P.N. First, Effective theory of rotationally faulted multilayer graphene-the local limit, J. Phys. D-Appl. Phys., 45 (2012) 154005.

[155] E.J. Mele, Band symmetries and singularities in twisted multilayer graphene, Phys Rev B, 84 (2011) 235439.

[156] R. Bistritzer, A.H. MacDonald, Moiré bands in twisted double layer graphene, PNAS, 108 (2011) 12233-12237.

[157] I. Brihuega, P. Mallet, H. Gonzalez-Herrero, G.T. de Laissardiere, M.M. Ugeda, L. Magaud, J.M. Gomez-Rodriguez, F. Yndurain, J.Y. Veuillen, Unraveling the Intrinsic and Robust Nature of van Hove Singularities in Twisted Bilayer Graphene by Scanning Tunneling Microscopy and Theoretical Analysis, Physical Review Letters, 109 (2012) 196802.

[158] V. Cherkez, G.T. de Laissardiere, P. Mallet, J.Y. Veuillen, Van Hove singularities in doped twisted graphene bilayers studied by scanning tunneling spectroscopy, Phys Rev B, 91 (2015).

[159] J. Kunc, Y. Hu, J. Palmer, C. Berger, W.A. de Heer, A method to extract pure Raman spectrum of epitaxial graphene on SiC, Appl Phys Lett, 103 (2013) 201911

[160] A.C. Ferrari, J.C. Meyer, V. Scardaci, C. Casiraghi, M. Lazzeri, F. Mauri, S. Piscanec, D. Jiang, K.S. Novoselov, S. Roth, A.K. Geim, Raman spectrum of graphene and graphene layers, Physical Review Letters, 97 (2006) 187401

[161] Y. Hu, Production and properties of epitaxial graphene on the carbon terminated face of hexagonal silicon carbide, in: School of Physics, PhD - Georgia Institute of Technology, Atlanta, 2013.

[162] S. Shivaraman, M.V.S. Chandrashekhar, J.J. Boeckl, M.G. Spencer, Thickness Estimation of Epitaxial Graphene on SiC Using Attenuation of Substrate Raman Intensity, J Electron Mater, 38 (2009) 725-730.

[163] H. Hibino, H. Kageshima, M. Nagase, Epitaxial few-layer graphene: towards single crystal growth, J. Phys. D-Appl. Phys., 43 (2010) 374005. [164] M. Sprinkle, et.al, unpublished, (2008).

[165] Z.H. Ni, W. Chen, X.F. Fan, J.L. Kuo, T. Yu, A.T.S. Wee, Z.X. Shen, Raman spectroscopy of epitaxial graphene on a SiC substrate, Phys Rev B, 77 (2008) 115416.

[166] J. Zabel, R.R. Nair, A. Ott, T. Georgiou, A.K. Geim, K.S. Novoselov, C. Casiraghi, Raman Spectroscopy of Graphene and Bilayer under Biaxial Strain: Bubbles and Balloons, Nano Lett, 12 (2012) 617-621.

[167] T.M.G. Mohiuddin, A. Lombardo, R.R. Nair, A. Bonetti, G. Savini, R. Jalil, N. Bonini, D.M. Basko, C. Galiotis, N. Marzari, K.S. Novoselov, A.K. Geim, A.C. Ferrari, Uniaxial strain in graphene by Raman spectroscopy: G peak splitting, Gruneisen parameters, and sample orientation, Phys Rev B, 79 (2009) 205433.

[168] F. Fromm, P. Wehrfritz, M. Hundhausen, T. Seyller, Looking behind the scenes: Raman spectroscopy of top-gated epitaxial graphene through the substrate, New J Phys, 15 (2013) 113006.

[169] P. Venezuela, M. Lazzeri, F. Mauri, Theory of double-resonant Raman spectra in graphene: Intensity and line shape of defect-induced and two-phonon bands, Phys Rev B, 84 (2011) 035433.

[170] J. Röhrl, M. Hundhausen, K.V. Emtsev, T. Seyller, R. Graupner, L. Ley, Raman spectra of epitaxial graphene on SiC(0001), Appl Phys Lett, 92 (2008) 201918.

[171] M. Ostler, I. Deretzis, S. Mammadov, F. Giannazzo, G. Nicotra, C. Spinella, T. Seyller, A. La Magna, Direct growth of quasi-free-standing epitaxial graphene on nonpolar SiC surfaces, Phys Rev B, 88 (2013) 085408.

[172] J. Ristein, S. Mammadov, T. Seyller, Origin of Doping in Quasi-Free-Standing Graphene on Silicon Carbide, Physical Review Letters, 108 (2012) 246104.

[173] B. Jabakhanji, N. Camara, A. Caboni, C. Consejo, B. Jouault, P. Godignon, J. Camassel, Almost free standing Graphene on SiC(000-1) and SiC(11-20), Heterosic & Wasmpe 2011, 711 (2012) 235-241.

[174] P. Hens, V. Jokubavicius, R. Liljedahl, G. Wagner, R. Yakimova, P. Wellmann, M. Syväjärvi, Sublimation growth of thick freestanding 3C-SiC using CVD-templates on silicon as seeds, Materials Letters, 67 (2012) 300-302.

[175] M. Suemitsu, S. Jiao, H. Fukidome, Y. Tateno, I. Makabe, T. Nakabayashi, Epitaxial graphene formation on 3C-SiC/Si thin films, J. Phys. D-Appl. Phys., 47 (2014).

[176] C. Coletti, K.V. Emtsev, A.A. Zakharov, T. Ouisse, D. Chaussende, U. Starke, Large area quasi-free standing monolayer graphene on 3C-SiC(111), Appl Phys Lett, 99 (2011) 081904.

[177] V.Y. Aristov, G. Urbanik, K. Kummer, D.V. Vyalikh, O.V. Molodtsova, A.B. Preobrajenski, A.A. Zakharov, C. Hess, T. Hanke, B. Buchner, I. Vobornik, J. Fujii, G. Panaccione, Y.A. Ossipyan, M. Knupfer, Graphene Synthesis on Cubic SiC/Si Wafers. Perspectives for Mass Production of Graphene-Based Electronic Devices, Nano Lett, 10 (2010) 992-995.

[178] A. Ouerghi, M. Ridene, A. Balan, R. Belkhou, A. Barbier, N. Gogneau, M. Portail, A. Michon, S. Latil, P. Jegou, A. Shukla, Sharp interface in epitaxial graphene layers on 3C-SiC(100)/Si(100) wafers, Phys Rev B, 83 (2011).

[179] G.R. Yazdi, R. Vasiliauskas, T. Iakimov, A. Zakharov, M. Syvajarvi, R. Yakimova, Growth of large area monolayer graphene on 3C-SiC and a comparison with other SiC polytypes, Carbon, 57 (2013) 477-484.

[180] S. Mammadov, J. Ristein, R.J. Koch, M. Ostler, C. Raidel, M. Wanke, R. Vasiliauskas, R. Yakimova, T. Seyller, Polarization doping of graphene on silicon carbide, 2d Mater, 1 (2014) 035003.

[181] P. Hens, A.A. Zakharov, T. Iakimov, M. Syvajarvi, R. Yakimova, Large area buffer-free graphene on non-polar (001) cubic silicon carbide, Carbon, 80 (2014) 823-829.

[182] M.Y. Han, J.C. Brant, P. Kim, Electron Transport in Disordered Graphene Nanoribbons, Physical Review Letters, 104 (2010) 056801.
 [183] L.Y. Jiao, L. Zhang, X.R. Wang, G. Diankov, H.J. Dai, Narrow graphene nanoribbons from carbon nanotubes, Nature, 458 (2009) 877-880.
 [184] J.B. Oostinga, B. Sacepe, M.F. Craciun, A.F. Morpurgo, Magnetotransport through graphene nanoribbons, Phys Rev B, 81 (2010) 193408.

[185] Z.H. Chen, Y.M. Lin, M.J. Rooks, P. Avouris, Graphene nano-ribbon electronics, Physica E, 40 (2007) 228-232.

[186] K. Todd, H.T. Chou, S. Amasha, D. Goldhaber-Gordon, Quantum Dot Behavior in Graphene Nanoconstrictions, Nano Lett, 9 (2009) 416-421. [187] F. Molitor, A. Jacobsen, C. Stampfer, J. Guttinger, T. Ihn, K. Ensslin, Transport gap in side-gated graphene constrictions, Phys Rev B, 79 (2009) 075426.

[188] J.M. Cai, P. Ruffieux, R. Jaafar, M. Bieri, T. Braun, S. Blankenburg, M. Muoth, A.P. Seitsonen, M. Saleh, X.L. Feng, K. Mullen, R. Fasel, Atomically precise bottom-up fabrication of graphene nanoribbons, Nature, 466 (2010) 470-473.

[189] P. Ruffieux, J.M. Cai, N.C. Plumb, L. Patthey, D. Prezzi, A. Ferretti, E. Molinari, X.L. Feng, K. Mullen, C.A. Pignedoli, R. Fasel, Electronic Structure of Atomically Precise Graphene Nanoribbons, Acs Nano, 6 (2012) 6930-6935.

[190] K. Wakabayashi, M. Fujita, H. Ajiki, M. Sigrist, Electronic and magnetic properties of nanographite ribbons, Phys Rev B, 59 (1999) 8271-8282.

[191] M. Ruan, Y. Hu, Z. Guo, R. Dong, J. Palmer, J. Hankinson, C. Berger, W.A. de Heer, Epitaxial graphene on silicon carbide: Introduction to structured graphene, Mrs Bull, 37 (2012) 1138-1147.

[192] W. Norimatsu, M. Kusunoki, Formation process of graphene on SiC (0001), Physica E, 42 (2010) 691-694.

[193] J. Baringhaus, J. Aprojanz, J. Wiegand, D. Laube, M. Halbauer, J. Hubner, M. Oestreich, C. Tegenkamp, Growth and characterization of sidewall graphene nanoribbons, Appl Phys Lett, 106 (2015).

[194] J. Baringhaus, F. Edler, C. Tegenkamp, Edge-states in graphene nanoribbons: a combined spectroscopy and transport study, J Phys-Condens Mat, 25 (2013) 392001.

[195] J. Robinson, X.J. Weng, K. Trumbull, R. Cavalero, M. Wetherington, E. Frantz, M. LaBella, Z. Hughes, M. Fanton, D. Snyder, Nucleation of Epitaxial Graphene on SiC(0001), Acs Nano, 4 (2010) 153-158.

[196] J. Hicks, A. Tejeda, A. Taleb-Ibrahimi, M.S. Nevius, F. Wang, K. Shepperd, J. Palmer, F. Bertran, P. Le Fevre, J. Kunc, W.A. de Heer, C. Berger, E.H. Conrad, A wide-bandgap metal-semiconductor-metal nanostructure made entirely from graphene, Nat Phys, 9 (2013) 49-54.

[197] F. Giannazzo, I. Deretzis, G. Nicotra, G. Fisichella, Q.M. Ramasse, C. Spinella, F. Roccaforte, A. La Magna, High resolution study of structural and electronic properties of epitaxial graphene grown on off-axis 4H-SiC (0001), J Cryst Growth, 393 (2014) 150-155.

[198] G. Nicotra, Q.M. Ramasse, I. Deretzis, A. La Magna, C. Spinella, F. Giannazzo, Delaminated Graphene at Silicon Carbide Facets: Atomic Scale Imaging and Spectroscopy, Acs Nano, 7 (2013) 3045-3052.

[199] M.S. Nevius, F. Wang, C. Mathieu, N. Barrett, A. Sala, T.O. Mentes, A. Locatelli, E.H. Conrad, The Bottom-up Growth of Edge Specific Graphene Nanoribbons, Nano Lett, 14 (2014) 6080-6086.

[200] A. Nakajima, H. Yokoya, Y. Furukawa, H. Yonezu, Step control of vicinal 6H-SiC(0001) surface by H-2 etching, J Appl Phys, 97 (2005) 104919.

[201] S. Frank, P. Poncharal, Z.L. Wang, W.A. de Heer, Carbon nanotube quantum resistors, Science, 280 (1998) 1744-1746.

[202] M. Orlita, C. Faugeras, R. Grill, A. Wysmolek, W. Strupinski, C. Berger, W.A. de Heer, G. Martinez, M. Potemski, Carrier Scattering from Dynamical Magnetoconductivity in Quasineutral Epitaxial Graphene, Physical Review Letters, 107 (2011) 216603.

[203] D. Sun, C. Divin, C. Berger, W.A. de Heer, P.N. First, T.B. Norris, Spectroscopic Measurement of Interlayer Screening in Multilayer Epitaxial Graphene, Physical Review Letters, 104 (2010) 136802.

[204] Y. Liu, R.F. Willis, Plasmon-phonon strongly coupled mode in epitaxial graphene, Phys Rev B, 81 (2010).

[205] H.K. Min, S. Adam, Y.J. Song, J.A. Stroscio, M.D. Stiles, A.H. MacDonald, Landau levels and band bending in few-layer epitaxial graphene, Phys Rev B, 83 (2011).

[206] A.E. Curtin, M.S. Fuhrer, J.L. Tedesco, R.L. Myers-Ward, C.R. Eddy, D.K. Gaskill, Kelvin probe microscopy and electronic transport in graphene on SiC(0001) in the minimum conductivity regime, Appl Phys Lett, 98 (2011).

[207] N.H. Shon, T. Ando, Quantum transport in two-dimensional graphite system, J Phys Soc Jpn, 67 (1998) 2421-2429.

[208] S. Das Sarma, S. Adam, E.H. Hwang, E. Rossi, Electronic transport in two-dimensional graphene, Review of modern physics, 83 (2011) 407-466.

[209] J.H. Chen, C. Jang, S. Adam, M.S. Fuhrer, E.D. Williams, M. Ishigami, Charged-impurity scattering in graphene, Nat Phys, 4 (2008) 377-381. [210] S. Kim, J. Nah, I. Jo, D. Shahrjerdi, L. Colombo, Z. Yao, E. Tutuc, S.K. Banerjee, Realization of a high mobility dual-gated graphene field-effect transistor with Al2O3 dielectric, Appl Phys Lett, 94 (2009) 062107.

[211] P. Wehrfritz, F. Fromm, S. Malzer, T. Seyller, Quasi-freestanding epitaxial graphene transistor with silicon nitride top gate, J. Phys. D-Appl. Phys., 47 (2014).

[212] J. Kedzierski, P.L. Hsu, P. Healey, P.W. Wyatt, C.L. Keast, M. Sprinkle, C. Berger, W.A. de Heer, Epitaxial graphene transistors on SIC substrates, IEEE Transactions on Electron Devices, 55 (2008) 2078-2085.

[213] V.K. Sangwan, D. Jariwala, S.A. Filippone, H.J. Karmel, J.E. Johns, J.M.P. Alaboson, T.J. Marks, L.J. Lauhon, M.C. Hersam, Quantitatively Enhanced Reliability and Uniformity of High-kappa Dielectrics on Graphene Enabled by Self-Assembled Seeding Layers, Nano Lett, 13 (2013) 1162-1167.

[214] A. Tzalenchuk, S. Lara-Avila, K. Cedergren, M. Syvajarvi, R. Yakimova, O. Kazakova, T.J.B.M. Janssen, K. Moth-Poulsen, T. Bjornholm, S. Kopylov, V. Fal'ko, S. Kubatkin, Engineering and metrology of epitaxial graphene, Solid State Communications, 151 (2011) 1094-1099.
[215] F. Krach, S. Hertel, D. Waldmann, J. Jobst, M. Krieger, S. Reshanov, A. Schoner, H.B. Weber, A switch for epitaxial graphene electronics: Utilizing the silicon carbide substrate as transistor channel, Appl Phys Lett, 100 (2012).

[216] D. Waldmann, J. Jobst, F. Speck, T. Seyller, M. Krieger, H.B. Weber, Bottom-gated epitaxial graphene, Nature Materials, 10 (2011) 357-360. [217] B. Jabakhanji, C. Consejo, N. Camara, W. Desrat, P. Godignon, B. Jouault, Quantum Hall effect of self-organized graphene monolayers on the C-face of 6H-SiC, J. Phys. D-Appl. Phys., 47 (2014) 094009.

[218] J. Jobst, D. Waldmann, F. Speck, R. Hirner, D.K. Maude, T. Seyller, H.B. Weber, Quantum oscillations and quantum Hall effect in epitaxial graphene, Phys Rev B, 81 (2010) 195434.

[219] A. Satrapinski, S. Novikov, N. Lebedeva, Precision quantum Hall resistance measurement on epitaxial graphene device in low magnetic field, Appl Phys Lett, 103 (2013) 173509.

[220] T.H. Bointon, I. Khrapach, R. Yakimova, A.V. Shytov, M.F. Craciun, S. Russo, Approaching Magnetic Ordering in Graphene Materials by FeCl3 Intercalation, Nano Lett, 14 (2014) 1751-1755. [221] Z.L. Guo, R. Dong, P.S. Chakraborty, N. Lourenco, J. Palmer, Y.K. Hu, M. Ruan, J. Hankinson, J. Kunc, J.D. Cressler, C. Berger, W.A. de Heer, Record Maximum Oscillation Frequency in C-Face Epitaxial Graphene Transistors, Nano Lett, 13 (2013) 942-947.

[222] B. Jouault, B. Jabakhanji, N. Camara, W. Desrat, A. Tiberj, J.R. Huntzinger, C. Consejo, A. Caboni, P. Godignon, Y. Kopelevich, J. Camassel, Probing the electrical anisotropy of multilayer graphene on the Si face of 6H-SiC, Phys Rev B, 82 (2010).

[223] F. Giannazzo, I. Deretzis, G. Nicotra, G. Fisichella, C. Spinella, F. Roccaforte, A. La Magna, Electronic properties of epitaxial graphene residing on SiC facets probed by conductive atomic force microscopy, Applied Surface Science, 291 (2014) 53-57.

[224] T. Schumann, K.J. Friedland, M.H. Oliveira, A. Tahraoui, J.M.J. Lopes, H. Riechert, Anisotropic quantum Hall effect in epitaxial graphene on stepped SiC surfaces (vol 85, 235402, 2012), Phys Rev B, 85 (2012).

[225] M.K. Yakes, D. Gunlycke, J.L. Tedesco, P.M. Campbell, R.L. Myers-Ward, C.R. Eddy, D.K. Gaskill, P.E. Sheehan, A.R. Laracuente, Conductance Anisotropy in Epitaxial Graphene Sheets Generated by Substrate Interactions, Nano Lett, 10 (2010) 1559-1562.

[226] S.H. Ji, J.B. Hannon, R.M. Tromp, V. Perebeinos, J. Tersoff, F.M. Ross, Atomic-scale transport in epitaxial graphene, Nature Materials, 11 (2012) 114-119.

[227] M. Nagase, H. Hibino, H. Kageshima, H. Yamaguchi, Local conductance measurements of double-layer graphene on SiC substrate, Nanotechnology, 20 (2009).

[228] S. Hertel, F. Kisslinger, J. Jobst, D. Waldmann, M. Krieger, H.B. Weber, Current annealing and electrical breakdown of epitaxial graphene, Appl Phys Lett, 98 (2011).

[229] J. Moser, A. Barreiro, A. Bachtold, Current-induced cleaning of graphene, Appl Phys Lett, 91 (2007) 163513.

[230] E. McCann, K. Kechedzhi, V.I. Fal'ko, H. Suzuura, T. Ando, B.L. Altshuler, Weak-localization magnetoresistance and valley symmetry in graphene, Physical Review Letters, 97 (2006) 146805.

[231] J. Jobst, F. Kisslinger, H.B. Weber, Detection of the Kondo effect in the resistivity of graphene: Artifacts and strategies, Phys Rev B, 88 (2013) 155412.

[232] M.O. Goerbig, Electronic properties of graphene in a strong magnetic field, Rev Mod Phys, 83 (2011) 1193-1243.

 [233] M. Orlita, M. Potemski, Dirac electronic states in graphene systems: optical spectroscopy studies, Semicond. Sci. Technol., 25 (2010).
 [234] C. Faugeras, M. Amado, P. Kossacki, M. Orlita, M. Sprinkle, C. Berger, W.A. de Heer, M. Potemski, Tuning the Electron-Phonon Coupling in Multilayer Graphene with Magnetic Fields, Physical Review Letters, 103 (2009) 186803.

[235] K.S. Novoselov, Z. Jiang, Y. Zhang, S.V. Morozov, H.L. Stormer, U. Zeitler, J.C. Maan, G.S. Boebinger, P. Kim, A.K. Geim, Room-temperature quantum hall effect in graphene, Science, 315 (2007) 1379-1379.

[236] A. Tzalenchuk, S. Lara-Avila, A. Kalaboukhov, S. Paolillo, M. Syvajarvi, R. Yakimova, O. Kazakova, T.J.B.M. Janssen, V. Fal'ko, S. Kubatkin, Towards a quantum resistance standard based on epitaxial graphene, Nat Nanotechnol, 5 (2010) 186-189.

[237] E. Pallecchi, F. Lafont, V. Cavaliere, F. Schopfer, D. Mailly, W. Poirier, A. Ouerghi, High Electron Mobility in Epitaxial Graphene on 4H-SiC(0001) via post-growth annealing under hydrogen, Sci Rep-Uk, 4 (2014).

[238] R. Ribeiro-Palau, F. Lafont, J. Brun-Picard, D. Kazazis, A. Michon, F. Cheynis, O. Couturaud, C. Consejo, B. Jouault, W. Poirier, F. Schopfer, Quantum Hall resistance standard in graphene devices under relaxed experimental conditions, Nat Nanotechnol, 10 (2015) 965-U168.

[239] C. Chua, M. Connolly, A. Lartsev, T. Yager, S. Lara-Avila, S. Kubatkin, S. Kopylov, V. Fal'ko, R. Yakimova, R. Pearce, T.J.B.M. Janssen, A. Tzalenchuk, C.G. Smith, Quantum Hall Effect and Quantum Point Contact in Bilayer-Patched Epitaxial Graphene, Nano Lett, 14 (2014) 3369-3373.
[240] F. Schwierz, Graphene Transistors: Status, Prospects, and Problems, P Ieee, 101 (2013) 1567-1584.

[241] I. Meric, M.Y. Han, A.F. Young, B. Ozyilmaz, P. Kim, K.L. Shepard, Current saturation in zero-bandgap, topgated graphene field-effect transistors, Nat Nanotechnol, 3 (2008) 654-659.

[242] F. Schwierz, Electronics Industry-Compatible Graphene Transistors, Nature, 472 (2011) 41-42.

[243] Y.Q. Wu, D.B. Farmer, A. Valdes-Garcia, W.J. Zhu, K.A. Jenkins, C. Dimitrakopoulos, P. Avouris, Y.M. Lin, Record High RF Performance for Epitaxial Graphene Transistors, 2011 leee International Electron Devices Meeting (ledm), (2011).

[244] R. Cheng, J.W. Bai, L. Liao, H.L. Zhou, Y. Chen, L.X. Liu, Y.C. Lin, S. Jiang, Y. Huang, X.F. Duan, High-frequency self-aligned graphene transistors with transferred gate stacks, Proceedings of the National Academy of Sciences of the United States of America, 109 (2012) 11588-11592.
[245] R. Krithivasan, Y. Lu, J.D. Cressler, J.S. Rieh, M.H. Khater, D. Ahlgren, G. Freeman, Half-terahertz operation of SiGeHBTs, Ieee Electron Device Letters, 27 (2006) 567-569.

[246] Y.-M. Lin, D.B. Farmer, K.A. Jenkins, Y. Wu, J.L. Tedesco, R.L.E. Myers-Ward, C. R. Jr., D.K. Gaskill, C. Dimitrakopoulos, P. Avouris, Enhanced performance in epitaxial graphene FETs with optimized channel morphology, IEEE Electron device letters 32 (2011) 1343.

[247] B. Dlubak, P. Seneor, A. Anane, C. Barraud, C. Deranlot, D. Deneuve, B. Servet, R. Mattana, F. Petroff, A. Fert, Are Al2O3 and MgO tunnel barriers suitable for spin injection in graphene?, Appl Phys Lett, 97 (2010) 092502.

 [248] J.A. Robinson, M. LaBella, K.A. Trumbull, X.J. Weng, R. Cavelero, T. Daniels, Z. Hughes, M. Hollander, M. Fanton, D. Snyder, Epitaxial Graphene Materials Integration: Effects of Dielectric Overlayers on Structural and Electronic Properties, Acs Nano, 4 (2010) 2667-2672.
 [249] Q.H. Wang, M.C. Hersam, Room-temperature molecular-resolution characterization of self-assembled organic monolayers on epitaxial

[249] Q.H. Wang, M.C. Hersam, Room-temperature molecular-resolution characterization of self-assembled organic monolayers on epitaxial graphene, Nat Chem, 1 (2009) 206-211.

[250] V.K. Sangwan, D. Jariwala, K. Everaerts, J.J. McMorrow, J.T. He, M. Grayson, L.J. Lauhon, T.J. Marks, M.C. Hersam, Wafer-scale solutionderived molecular gate dielectrics for low-voltage graphene electronics, Appl Phys Lett, 104 (2014) 083503.

[251] J.S. Moon, H.C. Seo, M. Antcliffe, D. Le, C. McGuire, A. Schmitz, L.O. Nyakiti, D.K. Gaskill, P.M. Campbell, K.M. Lee, P. Asbeck, Graphene FETs for Zero-Bias Linear Resistive FET Mixers, IEEE Electron Device Letters, 34 (2013) 465-467.

[252] J.S. Moon, D. Curtis, D. Zehnder, S. Kim, D.K. Gaskill, G.G. Jernigan, R.L. Myers-Ward, C.R. Eddy, P.M. Campbell, K.M. Lee, P. Asbeck, Low-Phase-Noise Graphene FETs in Ambipolar RF Applications, IEEE Electron Device Letters, 32 (2011) 270-272.

[253] J.S. Moon, H.C. Seo, M. Antcliffe, S. Lin, C. McGuire, D. Le, L.O. Nyakiti, D.K. Gaskill, P.M. Campbell, K.M. Lee, P. Asbeck, Graphene FET-Based Zero-Bias RF to Millimeter-Wave Detection, Ieee Electron Device Letters, 33 (2012) 1357-1359.

[254] B. Dlubak, M.B. Martin, C. Deranlot, B. Servet, S. Xavier, R. Mattana, M. Sprinkle, C. Berger, W.A. De Heer, F. Petroff, A. Anane, P. Seneor, A. Fert, Highly efficient spin transport in epitaxial graphene on SiC, Nat Phys, 8 (2012) 557-561.

[255] P. Seneor, B. Dlubak, M.B. Martin, A. Anane, H. Jaffres, A. Fert, Spintronics with graphene, Mrs Bull, 37 (2012) 1245-1254.

[256] R.G. Mani, J. Hankinson, C. Berger, W.A. de Heer, Observation of resistively detected hole spin resonance and zero-field pseudo-spin splitting in epitaxial graphene, Nat. Commun., 3 (2012) 996.

[257] T. Maassen, J.J. van den Berg, N. IJbema, F. Fromm, T. Seyller, R. Yakimova, B.J. van Wees, Long Spin Relaxation Times in Wafer Scale Epitaxial Graphene on SiC(0001), Nano Lett, 12 (2012) 1498-1502.

[258] T. Maassen, J.J. van den Berg, E.H. Huisman, H. Dijkstra, F. Fromm, T. Seyller, B.J. van Wees, Localized States Influence Spin Transport in Epitaxial Graphene, Physical Review Letters, 110 (2013) 067209.

[259] Y.M. Lin, A. Valdes-Garcia, S.J. Han, D.B. Farmer, I. Meric, Y.N. Sun, Y.Q. Wu, C. Dimitrakopoulos, A. Grill, P. Avouris, K.A. Jenkins, Wafer-Scale Graphene Integrated Circuit, Science, 332 (2011) 1294-1297.

[260] S. Hertel, D. Waldmann, J. Jobst, A. Albert, M. Albrecht, S. Reshanov, A. Schoner, M. Krieger, H.B. Weber, Tailoring the graphene/silicon carbide interface for monolithic wafer-scale electronics, Nat. Commun., 3 (2012).

[261] J. Kunc, Y. Hu, J. Palmer, Z. Guo, J. Hankinson, S.H. Gamal, C. Berger, W.A. de Heer, Planar Edge Schottky Barrier-Tunneling Transistors Using Epitaxial Graphene/SiC Junctions, Nano Lett, 14 (2014) 5170–5175.

[262] H.C. Kang, H. Karasawa, Y. Miyamoto, H. Handa, H. Fukidome, T. Suemitsu, M. Suemitsu, T. Otsuji, Epitaxial graphene top-gate FETs on silicon substrates, Solid State Electron, 54 (2010) 1071-1075.

[263] H.C. Kang, H. Karasawa, Y. Miyamoto, H. Handa, T. Suemitsu, M. Suemitsu, T. Otsuji, Epitaxial graphene field-effect transistors on silicon substrates, Solid State Electron, 54 (2010) 1010-1014.

[264] A. El Moutaouakil, H.C. Kang, H. Handa, H. Fukidome, T. Suemitsu, E. Sano, M. Suemitsu, T. Otsuji, Room Temperature Logic Inverter on Epitaxial Graphene-on-Silicon Device, Jpn J Appl Phys, 50 (2011) 070113.

[265] R. Dong, Z.L. Guo, J. Palmer, Y.K. Hu, M. Ruan, J. Hankinson, J. Kunc, S.K. Bhattacharya, C. Berger, W.A. de Heer, Wafer bonding solution to epitaxial graphene-silicon integration, J. Phys. D-Appl. Phys., 47 (2014) 094001 (094008pp).

[266] F. Wang, G. Liu, S. Rothwell, M. Nevius, A. Tejeda, A. Taleb-Ibrahimi, L.C. Feldman, P.I. Cohen, E.H. Conrad, Wide-Gap Semiconducting Graphene from Nitrogen-Seeded SiC, Nano Lett, 13 (2013) 4827-4832.

[267] M.S. Nevius, M. Conrad, F. Wang, A. Celis, M.N. Nair, A. Taleb-Ibrahimi, A. Tejeda, E.H. Conrad, Semiconducting Graphene from Highly Ordered Substrate Interactions, Physical Review Letters, 115 (2015) 136802.

[268] X.S. Wu, M. Sprinkle, X.B. Li, F. Ming, C. Berger, W.A. de Heer, Epitaxial-graphene/graphene-oxide junction: An essential step towards epitaxial graphene electronics, Physical Review Letters, 101 (2008) 026801.

[269] J.S. Moon, H.C. Seo, F. Stratan, M. Antcliffe, A. Schmitz, R.S. Ross, A.A. Kiselev, V.D. Wheeler, L.O. Nyakiti, D.K. Gaskill, K.M. Lee, P.M. Asbeck, Lateral Graphene Heterostructure Field-Effect Transistor, IEEE Electron Device Letters, 34 (2013) 1190-1192.

[270] E. Bekyarova, M.E. Itkis, P. Ramesh, C. Berger, M. Sprinkle, W.A. de Heer, R.C. Haddon, Chemical Modification of Epitaxial Graphene: Spontaneous Grafting of Aryl Groups, J Am Chem Soc, 131 (2009) 1336-1337.

[271] S. Niyogi, E. Bekyarova, M.E. Itkis, H. Zhang, K. Shepperd, J. Hicks, M. Sprinkle, C. Berger, C.N. Lau, W.A. Deheer, E.H. Conrad, R.C. Haddon, Spectroscopy of Covalently Functionalized Graphene, Nano Lett, 10 (2010) 4061-4066.

[272] J.M. Hong, S. Niyogi, E. Bekyarova, M.E. Itkis, P. Ramesh, N. Amos, D. Litvinov, C. Berger, W.A. de Heer, S. Khizroev, R.C. Haddon, Effect of Nitrophenyl Functionalization on the Magnetic Properties of Epitaxial Graphene, Small, 7 (2011) 1175-1180.

[273] M.Z. Hossain, M.B.A. Razak, S. Yoshimoto, K. Mukai, T. Koitaya, J. Yoshinobu, H. Sone, S. Hosaka, M.C. Hersam, Aqueous-Phase Oxidation of Epitaxial Graphene on the Silicon Face of SiC(0001), J Phys Chem C, 118 (2014) 1014-1020.

[274] J.M.P. Alaboson, Q.H. Wang, J.A. Kellar, J. Park, J.W. Elam, M.J. Pellin, M.C. Hersam, Conductive Atomic Force Microscope Nanopatterning of Epitaxial Graphene on SiC(0001) in Ambient Conditions, Adv Mater, 23 (2011) 2181.

[275] S. Zhou, S. Kim, E. Di Gennaro, Y. Hu, C. Gong, X. Lu, C. Berger, W. de Heer, E. Riedo, Y.J. Chabal, C. Aruta, A. Bongiorno, Film Structure of Epitaxial Graphene Oxide on SiC: Insight on the Relationship Between Interlayer Spacing, Water Content, and Intralayer Structure, Adv Mater Interfaces, 1 (2014).

[276] Z.Q. Wei, D.B. Wang, S. Kim, S.Y. Kim, Y.K. Hu, M.K. Yakes, A.R. Laracuente, Z.T. Dai, S.R. Marder, C. Berger, W.P. King, W.A. de Heer, P.E. Sheehan, E. Riedo, Nanoscale Tunable Reduction of Graphene Oxide for Graphene Electronics, Science, 328 (2010) 1373-1376.

[277] L. Brey, H.A. Fertig, Electronic states of graphene nanoribbons studied with the Dirac equation, Phys Rev B, 73 (2006) 235411.

[278] Y.W. Son, M.L. Cohen, S.G. Louie, Energy gaps in graphene nanoribbons, Physical Review Letters, 97 (2006) 216803.

[279] F. Bonaccorso, Z. Sun, T. Hasan, A.C. Ferrari, Graphene photonics and optoelectronics, Nat. Photonics, 4 (2010) 611-622.

[280] F.H.L. Koppens, T. Mueller, P. Avouris, A.C. Ferrari, M.S. Vitiello, M. Polini, Photodetectors based on graphene, other two-dimensional materials and hybrid systems, Nat Nanotechnol, 9 (2014) 780-793.

[281] P. Avouris, F.N. Xia, Graphene applications in electronics and photonics, Mrs Bull, 37 (2012) 1225-1234.

[282] Y.Y. Wang, Z.H. Ni, L. Liu, Y.H. Liu, C.X. Cong, T. Yu, X.J. Wang, D.Z. Shen, Z.X. Shen, Stacking-Dependent Optical Conductivity of Bilayer Graphene, Acs Nano, 4 (2010) 4074-4080.

[283] D. Sun, Z.K. Wu, C. Divin, X.B. Li, C. Berger, W.A. de Heer, P.N. First, T.B. Norris, Ultrafast Relaxation of Excited Dirac Fermions in Epitaxial Graphene Using Optical Differential Transmission Spectroscopy, Physical Review Letters, 101 (2008) 157402.

[284] M.T. Mihnev, J.R. Tolsma, C.J. Divin, D. Sun, R. Asgari, M. Polini, C. Berger, W.A. de Heer, A.H. MacDonald, T.B. Norris, Electronic cooling via interlayer Coulomb coupling in multilayer epitaxial graphene, Nat Commun, 6 (2015) 8105(8111pp).

[285] M.T. Mihnev, F. Wang, G. Liu, S. Rothwell, P.I. Cohen, L.C. Feldman, E.H. Conrad, T.B. Norris, Evidence for bandgap opening in buckled epitaxial graphene from ultrafast time-resolved terahertz spectroscopy, Appl Phys Lett, 107 (2015) 173107(173105pp).

[286] I. Gierz, J.C. Petersen, M. Mitrano, C. Cacho, I.C.E. Turcu, E. Springate, A. Stöhr, A. Köhler, U. Starke, A. Cavalleri, Snapshots of nonequilibrium Dirac carrier distributions in graphene, Nat Mater, 12 (2013) 1119-1124.

[287] M.T. Mihnev, F. Kadi, C.J. Divin, T. Winzer, S. Lee, C.-H. Liu, Z. Zhong, X. Wang, R.S.B. Ruoff, Claire, , W.A. de Heer, E. Malic, A. Knorr, T.B. Norris, Microscopic Origins of the Terahertz Carrier Relaxation and Cooling Dynamics in Graphene, Nat Nanotechnol, (submitted) (2015).
[288] S. Winnerl, M. Orlita, P. Plochocka, P. Kossacki, M. Potemski, T. Winzer, E. Malic, A. Knorr, M. Sprinkle, C. Berger, W.A. de Heer, H. Schneider, M. Helm, Carrier Relaxation in Epitaxial Graphene Photoexcited Near the Dirac Point, Physical Review Letters, 107 (2011) 237401.

[289] M. Breusing, S. Kuehn, T. Winzer, E. Malić, F. Milde, N. Severin, J.P. Rabe, C. Ropers, A. Knorr, T. Elsaesser, Ultrafast nonequilibrium carrier dynamics in a single graphene layer, Phys Rev B, 83 (2011) 153410.

[290] D. Brida, A. Tomadin, C. Manzoni, Y.J. Kim, A. Lombardo, S. Milana, R.R. Nair, K.S. Novoselov, A.C. Ferrari, G. Cerullo, M. Polini, Ultrafast collinear scattering and carrier multiplication in graphene, Nat Commun, 4 (2013) 1987.

[291] P. Plochocka, P. Kossacki, A. Golnik, T. Kazimierczuk, C. Berger, W.A. de Heer, M. Potemski, Slowing hot-carrier relaxation in graphene using a magnetic field, Phys Rev B, 80 (2009) 245415.

[292] M. Mittendorff, F. Wendler, E. Malic, A. Knorr, M. Orlita, M. Potemski, C. Berger, W.A. de Heer, H. Schneider, M. Helm, S. Winnerl, Carrier dynamics in Landau-quantized graphene featuring strong Auger scattering, Nat Phys, 11 (2015) 75-81.

[293] D. Sun, J. Rioux, J.E. Sipe, Y. Zou, M.T. Mihnev, C. Berger, W.A. de Heer, P.N. First, T.B. Norris, Evidence for interlayer electronic coupling in multilayer epitaxial graphene from polarization-dependent coherently controlled photocurrent generation, Phys Rev B, 85 (2012) 165427.

[294] D. Sun, C. Divin, J. Rioux, J.E. Sipe, C. Berger, W.A. de Heer, P.N. First, T.B. Norris, Coherent Control of Ballistic Photocurrents in Multilayer Epitaxial Graphene Using Quantum Interference, Nano Lett, 10 (2010) 1293-1296.

[295] J. Maysonnave, S. Huppert, F. Wang, S. Maero, C. Berger, W. de Heer, T.B. Norris, L.A. De Vaulchier, S. Dhillon, J. Tignon, R. Ferreira, J. Mangeney, Terahertz Generation by Dynamical Photon Drag Effect in Graphene Excited by Femtosecond Optical Pulses, Nano Lett, 14 (2014) 5797-5802.

[296] J. Karch, P. Olbrich, M. Schmalzbauer, C. Zoth, C. Brinsteiner, M. Fehrenbacher, U. Wurstbauer, M.M. Glazov, S.A. Tarasenko, E.L. Ivchenko, D. Weiss, J. Eroms, R. Yakimova, S. Lara-Avila, S. Kubatkin, S.D. Ganichev, Dynamic Hall Effect Driven by Circularly Polarized Light in a Graphene Layer, Physical Review Letters, 105 (2010) 227402.

[297] X. Hao, Y.F. Chen, Z.G. Wang, J.B. Liu, J.R. He, Y.R. Li, Significant photoelectrical response of epitaxial graphene grown on Si-terminated 6H-SiC, Chinese Phys B, 22 (2013).

[298] R.S. Singh, V. Nalla, W. Chen, W. Ji, A.T.S. Wee, Photoresponse in epitaxial graphene with asymmetric metal contacts, Appl Phys Lett, 100 (2012).

[299] R.J. Sun, Y. Zhang, K. Li, C. Hui, K. He, X.C. Ma, F. Liu, Tunable photoresponse of epitaxial graphene on SiC, Appl Phys Lett, 103 (2013).
[300] R.S. Singh, V. Nalla, W. Chen, A.T.S. Wee, W. Ji, Laser Patterning of Epitaxial Graphene for Schottky Junction Photodetectors, Acs Nano, 5 (2011) 5969-5975.

[301] M.E. Itkis, F.H. Wang, P. Ramesh, E. Bekyarova, S. Niyogi, X.L. Chi, C. Berger, W.A. de Heer, R.C. Haddon, Enhanced photosensitivity of electro-oxidized epitaxial graphene, Appl Phys Lett, 98 (2011) 093115

[302] E. Kusdemir, D. Ozkendir, V. Firat, C. Celebi, Epitaxial graphene contact electrode for silicon carbide based ultraviolet photodetector, J. Phys. D-Appl. Phys., 48 (2015).

[303] Z. Fei, A.S. Rodin, G.O. Andreev, W. Bao, A.S. McLeod, M. Wagner, L.M. Zhang, Z. Zhao, M. Thiemens, G. Dominguez, M.M. Fogler, A.H. Castro Neto, C.N. Lau, F. Keilmann, D.N. Basov, Gate-tuning of graphene plasmons revealed by infrared nano-imaging, Nature, 487 (2012) 82-85.

[304] J.N. Chen, M. Badioli, P. Alonso-Gonzalez, S. Thongrattanasiri, F. Huth, J. Osmond, M. Spasenovic, A. Centeno, A. Pesquera, P. Godignon, A.Z. Elorza, N. Camara, F.J.G. de Abajo, R. Hillenbrand, F.H.L. Koppens, Optical nano-imaging of gate-tunable graphene plasmons, Nature, 487 (2012) 77-81.

[305] Y. Liu, R.F. Willis, K.V. Emtsev, T. Seyller, Plasmon dispersion and damping in electrically isolated two-dimensional charge sheets, Phys Rev B, 78 (2008).

[306] C. Tegenkamp, H. Pfnur, T. Langer, J. Baringhaus, H.W. Schumacher, Plasmon electron-hole resonance in epitaxial graphene, J Phys-Condens Mat, 23 (2011).

[307] P.J. van Zwol, S. Thiele, C. Berger, W.A. de Heer, J. Chevrier, Nanoscale Radiative Heat Flow due to Surface Plasmons in Graphene and Doped Silicon, Physical Review Letters, 109 (2012) 264301.

[308] I. Crassee, M. Orlita, M. Potemski, A.L. Walter, M. Ostler, T. Seyller, I. Gaponenko, J. Chen, A.B. Kuzmenko, Intrinsic Terahertz Plasmons and Magnetoplasmons in Large Scale Monolayer Graphene, Nano Lett, 12 (2012) 2470-2474.

[309] I. Crassee, J. Levallois, A.L. Walter, M. Ostler, A. Bostwick, E. Rotenberg, T. Seyller, D. van der Marel, A.B. Kuzmenko, Giant Faraday rotation in single- and multilayer graphene, Nat Phys, 7 (2011) 48-51.

[310] J.M. Poumirol, W. Yu, X. Chen, C. Berger, W.A. de Heer, M.L. Smith, T. Ohta, W. Pan, M.O. Goerbig, D. Smirnov, Z. Jiang, Magnetoplasmons in Quasineutral Epitaxial Graphene Nanoribbons, Physical Review Letters, 110 (2013) 246803.

[311] O. Mitrofanov, W. Yuc, R.J. Thompson, Y. Jiang, Z.J. Greenberg, J. Palmer, I. Brenerb, W. Pand, C. Berger, W.A. de Heer, Z. Jiang, Terahertz near-field imaging of surface plasmon waves in graphene structures, Solid State Communications, (in print) (2015).

[312] O. Mitrofanov, W.L. Yu, R.J. Thompson, Y.X. Jiang, I. Brener, W. Pan, C. Berger, W.A. de Heer, Z.G. Jiang, Probing terahertz surface plasmon waves in graphene structures, Appl Phys Lett, 103 (2013) 111105.

[313] N. Kumada, S. Tanabe, H. Hibino, H. Kamata, M. Hashisaka, K. Muraki, T. Fujisawa, Plasmon transport in graphene investigated by timeresolved electrical measurements, Nat. Commun., 4 (2013).

[314] M. Koshino, Stacking-dependent optical absorption in multilayer graphene, New J Phys, 15 (2013) 015010.

[315] E. Stolyarova, K.T. Rim, S.M. Ryu, J. Maultzsch, P. Kim, L.E. Brus, T.F. Heinz, M.S. Hybertsen, G.W. Flynn, High-resolution scanning tunneling microscopy imaging of mesoscopic graphene sheets on an insulating surface, Proceedings of the National Academy of Sciences of the United States of America, 104 (2007) 9209-9212.

[316] K.R. Knox, S.C. Wang, A. Morgante, D. Cvetko, A. Locatelli, T.O. Mentes, M.A. Nino, P. Kim, R.M. Osgood, Spectromicroscopy of single and multilayer graphene supported by a weakly interacting substrate. ArXiv:, 0806.0355 (2008).

[317] K.R. Knox, A. Locatelli, M.B. Yilmaz, D. Cvetko, T.O. Mentes, M.A. Nino, P. Kim, A. Morgante, R.M. Osgood, Making angle-resolved photoemission measurements on corrugated monolayer crystals: Suspended exfoliated single-crystal graphene, Phys Rev B, 84 (2011) 115401.
 [318] B. Friedhelm, B. Abderrezak, Structure, energetics, and electronic states of III–V compound polytypes, Journal of Physics: Condensed Matter, 25 (2013) 273201.

[319] C. Mathieu, N. Barrett, J. Rault, Y.Y. Mi, B. Zhang, W.A. de Heer, C. Berger, E.H. Conrad, O. Renault, Microscopic correlation between chemical and electronic states in epitaxial graphene on SiC(000(1)over-bar), Phys Rev B, 83 (2011) 235436.

[320] X.B. Li, X.S. Wu, M. Sprinkle, F. Ming, M. Ruan, Y.K. Hu, C. Berger, W.A. de Heer, Top- and side-gated epitaxial graphene field effect transistors, Phys Status Solidi A, 207 (2010) 286-290.

[321] P.N. First, W.A. de Heer, T. Seyller, C. Berger, J.A. Stroscio, J.-S. Moon, Epitaxial Graphenes on Silicon Carbide, Mrs Bull, 35 (2010) 296-305.
 [322] W.A. de Heer, C. Berger, X. Wu, P.N. First, E.H. Conrad, X. Li, T. Li, M. Sprinkle, J. Hass, M. L. Sadowski, M. Potemski, G. Martinez, Epitaxial graphene, Solid State Communications, 143 (2007) 92-100.













 γ_0

Y

B₁

• B2

B3

ł

 γ_1

A10

(d) ABC

A1 q

B1

Y₁

FIGURE 1

2n





FIGURE 3





FIGURE 5



FIGURE 6























FIGURE 17

















FIGURE 23



FIGURE 24



FIGURE 25



FIGURE 26


FIGURE 27



FIGURE 28



FIGURE 29



FIGURE 30



FIGURE 31



FIGURE 32





FIGURE 34



FIGURE 35